UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-Q

(Mark One)

[X] QUARTERLY REPORT PURSUANT TO SECTION 13 or 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended June 30, 2006

[] TRANSITION REPORT PURSUANT TO SECTION 13 or 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from

Commission File No. 0-15279

GENERAL COMMUNICATION, INC.

(Exact name of registrant as specified in its charter)

STATE OF ALASKA (State or other jurisdiction of incorporation or organization)

92-0072737 (I.R.S. Employer Identification No.)

2550 Denali Street Suite 1000 Anchorage, Alaska

(Address of principal executive offices)

99503 (Zip Code)

Registrant's telephone number, including area code: (907) 868-5600

Former name, former address and former fiscal year, if changed since last report

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes X No

Indicate by check mark whether the registrant is an accelerated filer (as defined in Rule 12b-2 of the Exchange Act). Yes X No .

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, or a non-accelerated filer. See definition of "accelerated filer and larger accelerated filer" in Rule 12b-2 of the Exchange Act. (Check one).

Large accelerated filer [] Accelerated filer [X] Non-accelerated filer []

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes $\mbox{No } \mbox{X}$.

The number of shares outstanding of the registrant's classes of common stock as of August 2, 2006 was:

50,226,931 shares of Class A common stock; and 3,380,257 shares of Class B common stock.

1

<TABLE> <CAPTION>

GENERAL COMMUNICATION, INC. FORM 10-Q FOR THE QUARTER ENDED JUNE 30, 2006

TABLE OF CONTENTS

Page No.

PART I. FINANCIAL INFORMATION

Ttem 1. Financial Statements Consolidated Balance Sheets as of June 30, 2006

Consolidated Statements of Operations for the three and six months ended June 30, 2006 (unaudited) and 2005 (unaudited)
Consolidated Statements of Stockholders' Equity for the three and six months ended June 30, 2006 (unaudited) and 2005 (unaudited)
Consolidated Statements of Cash Flows for the six months ended June 30, 2006 (unaudited) and 2005 (unaudited)9
Notes to Interim Condensed Consolidated Financial Statements (unaudited)
Management's Discussion and Analysis of Financial Condition and Results of Operations
Quantitative and Qualitative Disclosures About Market Risk
Controls and Procedures56
FORMATION
Legal Proceedings5
Unregistered Sales of Equity Securities and Use of Proceeds5
Submission of Matters to a Vote of Security Holders58
Exhibits59
ms are omitted, as they are not applicable.
60

2

Cautionary Statement Regarding Forward-Looking Statements

You should carefully review the information contained in this Quarterly Report, but should particularly consider any risk factors that we set forth in this Quarterly Report and in other reports or documents that we file from time to time with the Securities and Exchange Commission ("SEC"). In this Quarterly Report, in addition to historical information, we state our future strategies, plans, objectives or goals and our beliefs of future events and of our future operating results, financial position and cash flows. In some cases, you can identify those so-called "forward-looking statements" by words such as "may," "will," "should," "expects," "plans," "anticipates," "believes," "estimates," "predicts," "potential," "project," or "continue" or the negative of those words and other comparable words. All forward-looking statements involve known and unknown risks, uncertainties and other important factors that may cause our actual results, performance, achievements, plans and objectives to differ materially from any future results, performance, achievements, plans and objectives expressed or implied by these forward-looking statements. In evaluating those statements, you should specifically consider various factors, including those outlined below. Those factors may cause our actual results to differ materially from any of our forward-looking statements. For these statements, we claim the protection of the safe harbor for forward-looking statements provided by the Private Securities Litigation Reform Act of 1995. Such risks, uncertainties and other factors include but are not limited to those identified under "Risk Factors" in our December 31, 2005 annual report on Form 10-K, including in conjunction with the forward-looking statements included in this Quarterly Report.

You should not place undue reliance on any such forward-looking statements. Further, any forward-looking statement, and such risks, uncertainties and other factors speak only as of the date on which they were originally made and we expressly disclaim any obligation or undertaking to disseminate any updates or revisions to any forward-looking statement to reflect any change in our expectations with regard to those statements or any other change in events, conditions or circumstances on which any such statement is based, except as required by law. New factors emerge from time to time, and it is not possible for us to predict what factors will arise or when. In addition, we cannot assess the impact of each factor on our business or the extent to which any factor, or combination of factors, may cause actual results to differ materially from those contained in any forward-looking statements.

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS

CAPTION> amounts in thousands)		(Unaudited) June 30,
ecember 31, ASSETS		2006
105 		
	<c></c>	
Cash and cash equivalents	\$	45,686
·		
Receivables		80,731
3,279 Less allowance for doubtful receivables 317		5,550
Net receivables		75,181
Deferred income taxes, net		20,801
Prepaid expenses 347		6,286
Inventories 556		2,881
Notes receivable from related parties 2		2,685
Property held for sale 312		2,315
Other current assets 572		5 , 938
Total current assets		161,773
operty and equipment in service, net of depreciation		434,847
3,008 nstruction in progress 337		24,306
Net property and equipment 51,345		459,153
ble certificates 1,565		191,565
,181		42,181
her intangible assets, net of amortization 201		7,813
ferred loan and senior notes costs, net of amortization of \$1,953 and \$1,451 at June 30, 2006 and December 31, 2005, respectively		7,509
011 tes receivable from related parties 544		84
her assets 299		8,143
Total other assets 9,801		257 , 295
	 \$	878 , 221

</TABLE>

See accompanying notes to interim condensed consolidated financial statements.

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS (Continued)

<pre><caption> (Amounts in thousands)</caption></pre>	(Unaudited) June 30,
December 31, LIABILITIES AND STOCKHOLDERS' EQUITY 2005	2006
SS> SC>	<c></c>
Current liabilities:	ć 1 00 <i>4</i>
Current maturities of obligations under long-term debt and capital leases 1,769	\$ 1,894
Accounts payable 23,217	26,640
Deferred revenue	15,139
Accrued payroll and payroll related obligations	13,176
17,925 Accrued interest	8,703
9,588 Accrued liabilities	6 , 759
5,814	·
Subscriber deposits 361	408
Total current liabilities 76,113	72,719
Long-term debt	473,360
474,115 Obligation under capital lease, excluding current maturity	1,192
 Obligation under capital lease due to related party, excluding current maturity	597
628 Deferred income taxes, net of deferred income tax benefit	77,955
69,753 Other liabilities	12,146
9,546	
Total liabilities	637 , 969
630,155	
	
Stockholders' equity:	
Common stock (no par):	
Class A. Authorized 100,000 shares; issued 51,568 and 51,200 shares at	177 100
	177,108
Class A. Authorized 100,000 shares; issued 51,568 and 51,200 shares at June 30, 2006 and December 31, 2005, respectively	177,108
Class A. Authorized 100,000 shares; issued 51,568 and 51,200 shares at June 30, 2006 and December 31, 2005, respectively 178,351 Class B. Authorized 10,000 shares; issued 3,380 and 3,843 shares at June 30, 2006 and December 31, 2005, respectively; convertible on a	
Class A. Authorized 100,000 shares; issued 51,568 and 51,200 shares at June 30, 2006 and December 31, 2005, respectively 178,351 Class B. Authorized 10,000 shares; issued 3,380 and 3,843 shares at	177,108 2,855
Class A. Authorized 100,000 shares; issued 51,568 and 51,200 shares at June 30, 2006 and December 31, 2005, respectively 178,351 Class B. Authorized 10,000 shares; issued 3,380 and 3,843 shares at June 30, 2006 and December 31, 2005, respectively; convertible on a share-per-share basis into Class A common stock 3,247 Less cost of 290 and 291 Class A and Class B common shares held in treasury	
Class A. Authorized 100,000 shares; issued 51,568 and 51,200 shares at June 30, 2006 and December 31, 2005, respectively 178,351 Class B. Authorized 10,000 shares; issued 3,380 and 3,843 shares at June 30, 2006 and December 31, 2005, respectively; convertible on a share-per-share basis into Class A common stock 3,247	
Class A. Authorized 100,000 shares; issued 51,568 and 51,200 shares at June 30, 2006 and December 31, 2005, respectively 178,351 Class B. Authorized 10,000 shares; issued 3,380 and 3,843 shares at June 30, 2006 and December 31, 2005, respectively; convertible on a share-per-share basis into Class A common stock 3,247 Less cost of 290 and 291 Class A and Class B common shares held in treasury at June 30, 2006 and December 31, 2005, respectively (1,730)	2,855 (1,723)
Class A. Authorized 100,000 shares; issued 51,568 and 51,200 shares at June 30, 2006 and December 31, 2005, respectively 178,351 Class B. Authorized 10,000 shares; issued 3,380 and 3,843 shares at June 30, 2006 and December 31, 2005, respectively; convertible on a share-per-share basis into Class A common stock 3,247 Less cost of 290 and 291 Class A and Class B common shares held in treasury at June 30, 2006 and December 31, 2005, respectively (1,730) Paid-in capital 16,425	2,855 (1,723) 17,856
Class A. Authorized 100,000 shares; issued 51,568 and 51,200 shares at June 30, 2006 and December 31, 2005, respectively 178,351 Class B. Authorized 10,000 shares; issued 3,380 and 3,843 shares at June 30, 2006 and December 31, 2005, respectively; convertible on a share-per-share basis into Class A common stock 3,247 Less cost of 290 and 291 Class A and Class B common shares held in treasury at June 30, 2006 and December 31, 2005, respectively (1,730) Paid-in capital	2,855 (1,723) 17,856 (1,279)
Class A. Authorized 100,000 shares; issued 51,568 and 51,200 shares at June 30, 2006 and December 31, 2005, respectively 178,351 Class B. Authorized 10,000 shares; issued 3,380 and 3,843 shares at June 30, 2006 and December 31, 2005, respectively; convertible on a share-per-share basis into Class A common stock 3,247 Less cost of 290 and 291 Class A and Class B common shares held in treasury at June 30, 2006 and December 31, 2005, respectively (1,730) Paid-in capital 16,425 Notes receivable with related parties issued upon stock option exercise	2,855 (1,723) 17,856 (1,279) 45,435
Class A. Authorized 100,000 shares; issued 51,568 and 51,200 shares at June 30, 2006 and December 31, 2005, respectively 178,351 Class B. Authorized 10,000 shares; issued 3,380 and 3,843 shares at June 30, 2006 and December 31, 2005, respectively; convertible on a share-per-share basis into Class A common stock 3,247 Less cost of 290 and 291 Class A and Class B common shares held in treasury at June 30, 2006 and December 31, 2005, respectively (1,730) Paid-in capital 16,425 Notes receivable with related parties issued upon stock option exercise (1,722) Retained earnings 49,049	2,855 (1,723) 17,856 (1,279) 45,435
Class A. Authorized 100,000 shares; issued 51,568 and 51,200 shares at June 30, 2006 and December 31, 2005, respectively 178,351 Class B. Authorized 10,000 shares; issued 3,380 and 3,843 shares at June 30, 2006 and December 31, 2005, respectively; convertible on a share-per-share basis into Class A common stock 3,247 Less cost of 290 and 291 Class A and Class B common shares held in treasury at June 30, 2006 and December 31, 2005, respectively (1,730) Paid-in capital 16,425 Notes receivable with related parties issued upon stock option exercise (1,722) Retained earnings 49,049	2,855 (1,723) 17,856 (1,279) 45,435
Class A. Authorized 100,000 shares; issued 51,568 and 51,200 shares at June 30, 2006 and December 31, 2005, respectively 178,351 Class B. Authorized 10,000 shares; issued 3,380 and 3,843 shares at June 30, 2006 and December 31, 2005, respectively; convertible on a share-per-share basis into Class A common stock 3,247 Less cost of 290 and 291 Class A and Class B common shares held in treasury at June 30, 2006 and December 31, 2005, respectively (1,730) Paid-in capital 16,425 Notes receivable with related parties issued upon stock option exercise (1,722) Retained earnings 49,049 Total stockholders' equity	2,855 (1,723) 17,856 (1,279) 45,435
Class A. Authorized 100,000 shares; issued 51,568 and 51,200 shares at June 30, 2006 and December 31, 2005, respectively 178,351 Class B. Authorized 10,000 shares; issued 3,380 and 3,843 shares at June 30, 2006 and December 31, 2005, respectively; convertible on a share-per-share basis into Class A common stock 3,247 Less cost of 290 and 291 Class A and Class B common shares held in treasury at June 30, 2006 and December 31, 2005, respectively (1,730) Paid-in capital 16,425 Notes receivable with related parties issued upon stock option exercise (1,722) Retained earnings 49,049 Total stockholders' equity 243,620	2,855 (1,723) 17,856 (1,279) 45,435

See accompanying notes to interim condensed consolidated financial statements.

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF OPERATIONS (Unaudited)

<caption></caption>	diced)		
Months Ended	Three Mon	ths Ended	Six
Tune 20	June	30,	
June 30, (Amounts in thousands, except per share amounts) 2005		2005	
		<c></c>	<c></c>
<c> Revenues 217,175</c>	\$ 118,220	110,665	231,042
Cost of goods sold (exclusive of depreciation and amortization shown separately below)	38,598	36,045	74 , 782
71,245 Selling, general and administrative expenses	40,667	38,019	80,281
75,199 Bad debt expense (recovery)	1,338	194	1,839
(159) Depreciation and amortization expense 36,052		18,348	
Operating income		18,059	
34,838			
Other income (expense): Interest expense	(8,696)	(8,403)	(17,250)
(16,735) Amortization of loan and senior notes fees	(251)	(448)	(502)
(931) Interest income	482	112	639
291 Other	282		169
			
Other expense, net	(8,183)	(8,739)	(16,944)
(17, 375)			
Net income before income taxes and cumulative effect of a change in accounting principle 17,463	9,262	9,320	16,863
Income tax expense	3 , 856	4,036	7,535
7,516			
Net income before cumulative effect of a change in accounting principle 9,947	5,406	5,284	9,328
Cumulative effect of a change in accounting principle, net of income tax benefit of \$425			(608)
 Net income 9,947	5,406	5,284	8 , 720
Preferred stock dividends 148			
 Net income available to common shareholders 9,799	\$ 5,406		8 , 720
========	========	=======	========
Basic net income per common share: Net income before cumulative effect of a change in accounting principle	\$ 0.10	0.10	0.17
0.18 Cumulative effect of a change in accounting principle			(0.01)
3 1 1			

Net income	\$	0.10	0.10	0.16
0.18	Y	0.10	0.10	0.10
========	===	======	========	========
Diluted net income per common share:				
Net income before cumulative effect of a change in accounting principle	\$	0.09	0.09	0.16
0.18 Cumulative effect of a change in accounting principle				(0.01)
Net income	\$	0.09	0.09	0.15
0.18	===		=======	========

</TABLE>

See accompanying notes to interim condensed consolidated financial statements.

6

<TABLE>

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES
CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY
SIX MONTHS ENDED JUNE 30, 2006 AND 2005
(Unaudited)

<CAPTION>

<caption></caption>							
(Amounts in thousands) Total	Class A Common Stock	Class B Common Stock		Paid-in	Notes Receivable Issued to Related Parties	Retained Earnings	
<\$>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>
Balances at December 31, 2004	\$186,883	3,248	(1,702)	14,957	(3,016)	33,900	
234,270	•	ŕ		·		,	
Net income						9,947	
9,947							
Tax effect of excess stock compensation expense for tax purposes over amounts recognized for financial reporting purposes				104			
104				101			
Common stock repurchases						(8,252)	
(8,252)							
Common stock retirements	(8,538)					8,538	
Shares issued under stock option plan 469	469						
Redemption of Series B redeemable preferred stock (2,358)						(2,358)	
Stock based compensation expense				96			
96							
Purchase of treasury stock			(30)				
Payment received on note receivable issued to related party upon stock option exercise					38		
38							
Preferred stock dividends (148)						(148)	
Balances at June 30, 2005 234,136	\$178,814	3,248	(1,732)	15 , 157	(2,978)	41,627	

</TABLE>

See accompanying notes to interim condensed consolidated financial statements.

(Continued)

<TABLE>

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES
CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY
SIX MONTHS ENDED JUNE 30, 2006 AND 2005
(Unaudited)
(Continued)

Class A Common Stock			Paid-in	Issued to Related	Retained Earnings	
<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>
•	•	, , ,	•		,	
					8,720	
			(191)			
		(1)			(21.870)	
		(-/			(22,010)	
(9,167)	(369)				9,536	
7 , 683						
23	(23)					
218						
				443		
			1,622			
		8				
\$177 , 108	2,855	(1,723)	17,856	(1,279)	45,435	
	Common Stock	Common Stock Stock C>	Class A Class B Shares Common Stock Stock Treasury CC> C> C> C> C> (C> (1,730) (1) (9,167) (369) 7,683 (23) 218 218 8	Class A Common Stock Common Stock Treasury Capital CC> C> C> C> C> C> C> C> C> S178,351 3,247 (1,730) 16,425 (191) (1) (9,167) (369) 23 (23) 218 1,622 8	Class A Class B Shares Issued to Common Stock Stock Treasury Capital Parties CC> CC> CC> CC> CC> CC> S178,351 3,247 (1,730) 16,425 (1,722) (1) (191) (1) (23 (23) (243) 443 8 443 8	Class A Class B Shares Issued to Stock Treasury Paid-in Related Retained Earnings CC> CC> CC> CC> CC> CC> CC> CC> S178,351 3,247 (1,730) 16,425 (1,722) 49,049 (1) (21,870) (9,167) (369) 9,536 7,683 9,536 7,683 443 218 1,622 1,622 8

Notes

</TABLE>

See accompanying notes to interim condensed consolidated financial statements.

8

<TABLE>

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS SIX MONTHS ENDED JUNE 30, 2006 AND 2005 (Unaudited)

<caption></caption>			
(Amounts in thousands)	2	1006	2005
<\$>	<c></c>		<c></c>
Cash flows from operating activities:			
Net income	\$	8,720	9,947
Adjustments to reconcile net income to net cash provided by operating activities, net of acquisition:			
Depreciation and amortization expense	4	0,333	36,052
Deferred income tax expense		7,409	7,516
Share-based compensation expense		2,260	96
Cumulative effect of a change in accounting principle, net		608	
Amortization of loan and senior notes fees		502	931
Deferred compensation		239	449
Bad debt expense		233	172
Loss on disposal of property and equipment		19	459
Other noncash income and expense items		468	598
Change in operating assets and liabilities	(7,068)	(2,867)
Net cash provided by operating activities	5	3,723	53,353
Cash flows from investing activities:			
Purchases of property and equipment	(3	5,219)	(47,690)
Purchases of other assets and intangible assets	(3,133)	(2,021)
Notes receivable payments from related parties		649	
Notes receivable issued to related parties		(55)	(13)
Net cash acquired in business combination		52	
Additions to property held for sale		(3)	(191)
Proceeds from sales of assets			445

Net cash used in investing activities	(37,709)	(49,470)
Cash flows from financing activities:		
Purchase of common stock to be retired	(21,870)	(8,048)
Proceeds from common stock issuance	7,683	469
Repayment of Senior Credit Facility	(925)	
Payments received on notes receivable from related parties issued upon stock		
option exercise	443	
Repayments of capital lease obligations	(20)	(3,171)
Purchase of treasury stock	(1)	
(30)		
Redemption of Series B redeemable preferred stock		(6,607)
Payment of preferred stock dividends		(237)
Payment of debt issuance costs		
(86)		
Net cash used in financing activities	(14,690)	(17,710)
Net increase (decrease) in cash and cash equivalents	1,324	(13,827)
Cash and cash equivalents at beginning of period	44,362	31,452
Cash and cash equivalents at end of period	\$ 45,686	17,625

========

</TABLE>

See accompanying notes to interim condensed consolidated financial statements.

9

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES Notes to Interim Condensed Consolidated Financial Statements (Unaudited)

The accompanying unaudited interim condensed consolidated financial statements include the accounts of General Communication, Inc. ("GCI") and its subsidiaries and have been prepared in accordance with generally accepted accounting principles for interim financial information. Accordingly, they do not include all of the information and footnotes required by accounting principles generally accepted in the United States of America for complete financial statements. They should be read in conjunction with our audited consolidated financial statements for the year ended December 31, 2005, filed as part of our annual report on Form 10-K. In the opinion of management, all adjustments (consisting of normal recurring accruals) considered necessary for a fair presentation have been included. The results of operations for interim periods are not necessarily indicative of the results that may be expected for an entire year or any other period.

(1) Business and Summary of Significant Accounting Principles

In the following discussion, GCI and its direct and indirect subsidiaries are referred to as "we," "us" and "our."

(a) Business

 $\ensuremath{\mathsf{GCI}},$ an Alaska corporation, was incorporated in 1979. We offer the following services:

- o Long-distance telephone service between Alaska and the remaining United States and foreign countries,
- Cable television services throughout Alaska,
- o Facilities-based competitive local access services in Anchorage, Fairbanks, and Juneau, Alaska,
- o Internet access services.
- O Origination and termination of traffic in Alaska for certain common carriers,
- o Private line and private network services,
- Managed services to certain commercial customers,
- o Broadband services, including our SchoolAccess(R) offering to rural school districts and a similar offering to rural hospitals and health clinics.
- Sales and service of dedicated communications systems and related equipment,
- o Lease and sales of capacity on our fiber optic cable systems used in the transmission of interstate and intrastate private line, switched message long-distance and Internet services within Alaska and between Alaska and the remaining United States and foreign countries,
- Distribution of white and yellow pages directories to residential and business customers in certain markets we serve and on-line directory products, and

- o Resale of wireless telephone services and sale of wireless telephone handsets and accessories.
- (b) Principles of Consolidation The consolidated financial statements include the consolidated accounts of GCI and its wholly owned subsidiaries with all significant intercompany transactions eliminated.

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES

Notes to Interim Condensed Consolidated Financial Statements

(Unaudited)

(c) Earnings per Common Share Earnings per common share ("EPS") and common shares used to calculate basic and diluted EPS consist of the following (amounts in thousands, except per share amounts):

<table> <caption></caption></table>				Three	: Months	Ende	d June 30,		
			2006					2005	
		Income (Num-	Shares (Denom-	Per	-share		Income (Num-	Shares	Per-
share								: - \	
Amounts		erator)	inator)	AII	ounts		erator)	inator)	
	<pre><s> Net income Less Series B preferred stock dividends</s></pre>	<c> \$ 5,406</c>	<c></c>	<c></c>			5,284 55	<c></c>	<c></c>
0.10	Basic EPS: Net income	5,406	55,688	\$	0.10		5 , 229	54,637	\$
0.10	Effect of Dilutive Securities: Unexercised stock options		1,572					975	
	Diluted EPS:								
0.09	Net income	\$ 5,406	•		0.09		5 , 229	55 , 612	
<pre> </pre>									

			2006	51X	Months i	 	June 30,	2005			
		Income (Num-	Shares (Denom-	Per	-share		Income (Num-	Shares (Denom-	Per-		
share		erator)	inator)	Amounts		Amounts			erator)	inator)	
Amounts											
	<pre><s> Net income before cumulative effect of a change in</s></pre>	<c></c>	<c></c>	<c></c>		<c:< td=""><td>></td><td><c></c></td><td><c></c></td></c:<>	>	<c></c>	<c></c>		
	accounting principle	\$ 9,328				\$	9,947				
	Less Series B preferred stock dividends					_	148				
	Basic EPS:										
0 10	Net income before cumulative effect of a change in accounting principle	9,328	55 , 526	\$	0.17		9,799	54,815	\$		
0.18	effect of a change in	9,328	55,526 1,415	\$	0.17		9,799	54,815 1,104	\$		
	effect of a change in accounting principle Effect of Dilutive Securities:		·	\$					\$		
0.18	effect of a change in accounting principle Effect of Dilutive Securities:		·	\$ 							

______ _____

</TABLE>

11 (Continued)

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES
Notes to Interim Condensed Consolidated Financial Statements
(Unaudited)

EPS are presented in accordance with Statement of Financial Accounting Standard ("SFAS") No. 128, "Earnings Per Share." The dilutive effect of share-based compensation arrangements are computed using the treasury stock method. In applying the treasury stock method, assumed proceeds are computed as the sum of (a) the amount, if any, the employee must pay upon exercise, (b) the amount of compensation cost attributed to future services and not yet recognized, and (c) the amount of excess tax benefits, if any, that would be credited to additional paid-in capital assuming exercise of the options. Stock option agreements that may be settled in common stock or in cash are presumed to be settled in common stock and the resulting potential common shares are included in the denominator of the diluted EPS calculation since the effect is more dilutive. The numerator of the diluted EPS calculation has been adjusted for changes in net income that would result if the agreements had been reported as equity instruments for financial reporting purposes during 2006.

Stock options for the three and six months ended June 30, 2006 and 2005, which have been excluded in the computations of diluted EPS because the effect of including these stock options would have been anti-dilutive consist of the following (shares, in thousands):

<TABLE>

D d d		Three Month	ns Ended	Six Months
Ended		June 3	•	June 30,
2005		2006	2005	2006
<c></c>	<\$>	<c></c>	<c></c>	<c></c>
	Weighted average shares associated with outstanding stock options	1,336	582	1,430
401		==========	=======	=========

</TABLE>

Series B redeemable preferred stock common equivalent shares outstanding of \$470,000 and \$623,000 for the three and six months ended June 30, 2005, respectively, are not included in the diluted EPS calculations because they are anti-dilutive for purposes of calculating EPS.

We have not issued securities other than common stock that contractually entitle the holder to participate in dividends and earnings when, and if, we declare dividends on our common stock and, therefore, we do not apply the two-class method of calculating earnings per share.

12 (Continued)

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES

Notes to Interim Condensed Consolidated Financial Statements

(Unaudited)

(d) Common Stock

Following are the changes in common stock for the six months ended June 30, 2006 and 2005 (shares, in thousands):

<TABLE>

	Class A	Class B
<s> Balances at December 31, 2004 Class B shares converted to Class A Shares issued under stock option plan Shares retired</s>	<c> 51,825 12 83 (892)</c>	<c> 3,862 (12) </c>
Balances at June 30, 2005	51,028 =======	3,850 ======
Balances at December 31, 2005 Class B shares converted to Class A Shares issued under stock option plan	51,200 27 1,166	3,843 (27)

Shares issues under the Director Compensation Plan Shares retired

17 (842) (436)

Balances at June 30, 2006

51,568

3,380

</TABLE>

Our Board of Directors has authorized a common stock buyback program for the repurchase of our Class A and Class B common stock in order to reduce our outstanding shares of Class A and Class B common stock. Our Board of Directors authorized us and we obtained permission from our lenders for up to \$50.0 million of repurchases through June 30, 2006. We are authorized to continue our stock repurchases of up to \$5.0 million per quarter indefinitely and to use stock option exercise proceeds to repurchase additional shares. During the six months ended June 30, 2006 and 2005 we repurchased 1,857,000 and 931,000 shares of our Class A and B common stock at a cost of approximately \$21.9 million and \$8.7 million, respectively. The cost of the repurchased common stock is included in Retained Earnings on our Consolidated Balance Sheets.

If stock repurchases are less than the total approved quarterly amount the difference may be carried forward and applied against future stock repurchases. We expect to continue the repurchases for an indefinite period subject to the availability of free cash flow, availability under our credit facilities, and the price of our Class A and Class B common stock. The repurchases have and will continue to comply with the restrictions of SEC Rule 10b-18.

13 (Continued)

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES
Notes to Interim Condensed Consolidated Financial Statements
(Unaudited)

(e) Asset Retirement Obligations

(f) Share-based Payment Arrangements

compensation.

Following is a reconciliation of the beginning and ending aggregate carrying amount of our asset retirement obligations at June 30, 2006 and 2005 (amounts in thousands):

<TABLE>

<\$>	<c></c>
Balance at December 31, 2004	\$ 2 , 971
Accretion expense for the six months ended June 30, 2005	99
Balance at June 30, 2005	\$ 3,070 ======
Balance at December 31, 2005 Accretion expense for the six months ended	\$ 3,210
June 30, 2006 Liability settled	86
Balance at June 30, 2006	\$ 3,292

</TABLE>

Our asset retirement obligations are included in Other Liabilities.

Effective January 1, 2006, we adopted the provisions of SFAS No.

123(R), "Share-Based Payment," and related interpretations, to account for share-based compensation using the modified prospective transition method and therefore will not restate our prior period results. SFAS 123(R) supersedes Accounting Principles Board ("APB") Opinion No. 25, "Accounting for Stock Issued to Employees" and revises guidance in SFAS 123, "Accounting for Stock-Based Compensation." Among other things, SFAS 123(R) requires that compensation expense be recognized in the financial statements for share-based awards based on the grant date fair value of those awards. The modified prospective transition method applies to (a) unvested stock options under our 1986 Stock Option Plan ("Option Plan") and unvested stock options not issued pursuant to a plan that were outstanding as of December 31, 2005 based on the grant date fair value estimated in accordance with the pro forma provisions of SFAS 123, and (b) any new share-based awards granted subsequent to December 31, 2005, based on the grant-date fair value estimated in accordance with the provisions of SFAS 123(R). Additionally, share-based compensation expense includes an estimate for pre-vesting forfeitures and is recognized over the requisite service periods of the awards on a straight-line basis, which is

generally commensurate with the vesting term. We have recorded \$2,260,000 of share-based compensation expense, net of estimated forfeitures, during the six months ended June 30, 2006 as a result of our adoption of SFAS 123(R). See note 4 for information on the assumptions we used to calculate the fair value of share-based

Prior to January 1, 2006, we accounted for all of our stock option agreements in accordance with APB No. 25 and related interpretations. Accordingly, compensation expense for a stock option grant was recognized only if the exercise price was less than the market value of our common stock on the grant date. Prior to our adoption of SFAS 123(R), as required under the disclosure provisions of SFAS 123, as amended, we provided pro forma net income and income per common share for each period as if we had applied the fair value method to measure share-based compensation expense.

14 (Continued)

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES

Notes to Interim Condensed Consolidated Financial Statements

(Unaudited)

SFAS 123(R) requires the benefits associated with tax deductions in excess of recognized compensation cost to be reported as a financing cash flow rather than as an operating cash flow as previously required.

The table below shows the effect of adopting SFAS No. 123(R) on selected items and what those items would have been under previous guidance under APB No. 25 and SFAS No. 123 for the six months ended June 30, 2006 (amounts in thousands, except per share amounts):

<TABLE> <CAPTION>

SFAS 123		P	as Reported	Under APB No. 25	Under
<c></c>	<\$>		<c></c>	<c></c>	
	Operating income	\$	33,807	35,966	
33,807 16,863	Net income before income taxes and cumulative effect of a change in accounting principle		16,863	19,022	
10,003	Cumulative effect of a change in accounting principle, net of income tax benefit of \$425		(608)		
0.000	Net income available to common stockholders		8,720	10,599	
9,328	Cash flow from operating activities (1)		54,073	54,073	
54,073 (15,133)	Cash flow from financing activities (1)		(15,133)	(15,133)	
. 15	Basic net income per common share: Net income before cumulative effect of a change in accounting principle	\$	0.17	0.19	
0.17	Cumulative effect of a change in accounting principle		(0.01)		
0.17	Net income	\$ ==	0.16	0.19	
0.16	Diluted net income per common share: Net income before cumulative effect of a change in accounting principle Cumulative effect of a change in accounting principle	\$	0.16	0.19	
0.16	Net income	\$	0.15	0.19	=

(Continued)

</TABLE>

15

⁽¹⁾ For the six months ended June 30, 2006, we did not record any excess tax benefit generated from stock option exercises since we are in a net operating loss position and the income tax deduction will not yet reduce income taxes payable.

(Unaudited)

The table below summarizes the impact on our results of operations for the six months ended June 30, 2006 of outstanding stock options recognized under the provisions of SFAS 123(R) (amounts in thousands, except per share amounts):

<TABLE> <CAPTION>

<pre></pre>				2006
Income tax benefit 929 Net decrease in income Decrease in EPS: Basic and diluted 5 0.02		<\$>	<c></c>	
Net decrease in income \$ 1,331 =================================		Share-based employee compensation expense	\$	2,260
Decrease in EPS: Basic and diluted \$ 0.02				

The following illustrates the effect on net income and EPS for the six months ended June 30, 2005 as if we had applied the fair value method to measure share-based compensation, as required under the disclosure provisions of SFAS No. 123 (amounts in thousands, except | Income tax benefit | | 929 || Basic and diluted \$ 0.02 ================================== | | Net decrease in income | \$ | 1,331 |
``` The following illustrates the effect on net income and EPS for the six months ended June 30, 2005 as if we had applied the fair value method to measure share-based compensation, as required under the disclosure provisions of SFAS No. 123 (amounts in thousands, except ```		Decrease in EPS:		
The following illustrates the effect on net income and EPS for the six months ended June 30, 2005 as if we had applied the fair value method to measure share-based compensation, as required under the disclosure provisions of SFAS No. 123 (amounts in thousands, except		Basic and diluted	\$	0.02
six months ended June 30, 2005 as if we had applied the fair value method to measure share-based compensation, as required under the disclosure provisions of SFAS No. 123 (amounts in thousands, except				
per share amounts):		six months ended June 30, 2005 as if we had applied the fair value method to measure share-based compensation, as required under the		
		-		
<CAPTION>

CAPTION			hree Months ded June 30,	Six Months
Ended			2005	June 30,
2005				
	<\$>	<c></c>		<c></c>
	Net income available to common shareholders, as reported Total share-based employee compensation expense included	\$	5 <b>,</b> 229	9,799
	in reported net income, net of related tax effects  Less share-based employee compensation expense determined  under the SFAS 123 fair value method, net of related		48	96
	tax effects		(401)	(911)
	Pro forma net income	\$	4,876	8,984
========	===			
	EPS:			
	Basic - as reported	\$	0.10	0.18
	<del></del> -			
	Diluted - as reported	\$	0.09	0.18
			0.00	0.16
	Basic and diluted - pro forma	\$	0.09	0.16

## </TABLE>

(q) Exchanges of Nonmonetary Assets

The cost of a nonmonetary asset or service acquired in exchange for another nonmonetary asset or service is based upon the fair value of the asset surrendered to obtain it unless the fair value is not determinable, the exchange of a product or property held for sale in the ordinary course of business for a product or property to be sold in the same line of business to facilitate sales to customers other than the parties to the exchange, or the exchange lacks commercial substance. If the exceptions apply we value the transaction using the recorded amount (after reduction, if appropriate, for an indicated impairment of value) of the nonmonetary asset or service relinquished. A gain or loss may be recognized on the exchange.

(Continued)

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES Notes to Interim Condensed Consolidated Financial Statements (Unaudited)

- (h) Rental Costs Incurred During a Construction Period We recognize rental costs associated with ground or building operating leases that are incurred during a construction period as rental expense in income from continuing operations.
- (i) Reporting of Accounting Changes and Error Corrections On January 1, 2006, we adopted SFAS 154, "Accounting Changes and

Error Corrections—a replacement of APB Opinion No. 20 and Financial Accounting Standards Board ("FASB") Statement No. 3." SFAS 154 replaces APB Opinion No. 20, "Accounting Changes," and FASB Statement No. 3, "Reporting Accounting Changes in Interim Financial Statements," and changes the requirements for the accounting for and reporting of a change in accounting principle. SFAS 154 applies to all voluntary changes in accounting principle. It also applies to changes required by an accounting pronouncement in the unusual instance that the pronouncement does not include specific transition provisions. When a pronouncement includes specific transition provisions, those provisions should be followed.

# (j) New Accounting Pronouncements

In June 2006, the FASB issued Emerging Issues Task Force ("EITF") Issue No. 06-03, "How Taxes Collected from Customers and Remitted to Governmental Authorities Should Be Presented in the Income Statement (That Is, Gross versus Net Presentation)" which includes any tax assessed by a governmental authority that is directly imposed on a revenue-producing transaction between a seller and a customer and may include, but is not limited to, sales, use, value added, and some excise taxes. EITF No. 06-03 states that the presentation of taxes within its scope on either a gross (included in revenues and costs) or a net (excluded from revenues) basis is an accounting policy decision that should be disclosed in a company's financial statements. In addition, for any such taxes that are reported on a gross basis, a company should disclose the amounts of those taxes in interim and annual financial statements for each period for which an income statement is presented if those amounts are significant. The disclosure of those taxes can be done on an aggregate basis. We will begin application of EITF No. 06-03 on January 1, 2007 and do not expect it to have a material effect on our results of operations, financial position and cash flows.

In July 2006, the FASB issued FASB Interpretation ("FIN") 48, "Accounting for Uncertainty in Income Taxes" which clarifies the accounting for uncertainty in income taxes recognized in an enterprise's financial statements in accordance with FASB Statement No. 109, "Accounting for Income Taxes." FIN 48 prescribes a recognition threshold and measurement attribute for the financial statement recognition and measurement of a tax position taken or expected to be taken in a tax return. FIN 48 also provides guidance on derecognition, classification, interest and penalties, accounting in interim periods, disclosure, and transition. We will begin application of FIN 48 on January 1, 2007 and do not expect it to have a material effect on our results of operations, financial position, and cash flows.

#### (k) Reclassifications

Reclassifications have been made to the 2005 financial statements to make them comparable with the 2006 presentation.

(Continued)

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES
Notes to Interim Condensed Consolidated Financial Statements
(Unaudited)

(2) Consolidated Statements of Cash Flows Supplemental Disclosures Changes in operating assets and liabilities, net of acquisition, consist of (amounts in thousands):

<TABLE> <CAPTION>

Six month periods ended June 30,	2006	2005
<\$>	<c></c>	<c></c>
Increase in accounts receivable	\$ (2,591)	(3,130)
Decrease in prepaid expenses	521	696
Increase in inventories	(1,325)	(284)
(Increase) decrease in other current assets	(1,255)	137
Increase (decrease) in accounts payable	3 <b>,</b> 309	(437)
Decrease in deferred revenues	(1,300)	(1,106)
Increase (decrease) in accrued payroll and payroll		
related obligations	(4,749)	546
Increase in accrued liabilities	883	338
Decrease in accrued interest	(885)	(23)
Increase (decrease) in subscriber deposits	47	(48)
Increase in components of other long-term liabilities	277	444
	\$ (7,068)	(2,867)

#### </TABLE>

We paid interest totaling approximately \$18.0 million and \$16.7 million during the six months ended June 30, 2006 and 2005, respectively.

ended June 30, 2006 and 2005, respectively. We paid income taxes of \$76,000 and \$100,000 during the six months ended June 30, 2006 and 2005, respectively.

We recorded a net cumulative effect adjustment (expense) of \$672,000 during the six months ended June 30, 2006 to recognize the effect of initially measuring share-based compensation liability instruments at fair value. We recorded a net cumulative effect adjustment (benefit) of \$64,000 during the six months ended June 30, 2006 for share-based compensation instruments outstanding at December 31, 2005 for which the requisite service is not expected to be rendered. We recorded \$104,000 during the six months ended June 30, 2005 in paid-in capital in recognition of the income tax effect of excess stock compensation expense for tax purposes over amounts recognized for financial reporting purposes.

During the six months ended June 30, 2006 we financed a \$1.2 million acquisition of a building through a capital lease obligation.

#### (3) Intangible Assets

There have been no events or circumstances that indicate the recoverability of the carrying amounts of indefinite-lived and definite-lived intangible assets has changed as of June 30, 2006. The remaining useful lives of our cable certificates and goodwill were evaluated as of June 30, 2006 and events and circumstances continue to support an indefinite useful life. We reviewed the useful lives assigned to our definite-lived intangible assets and believe the lives continue to be appropriate as of June 30, 2006.

18 (Continued)

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES
Notes to Interim Condensed Consolidated Financial Statements
(Unaudited)

Cable certificates and goodwill are not allocated to a reportable segment but are included in the All Other category of our segment assets.

Amortization expense for amortizable intangible assets was as follows (amounts in thousands):

<TABLE>

<CAPTION>

Amortization expense	\$ 399	313
<\$>	<c></c>	<c></c>
	2006	2005
	Three Months	Ended June 30,

#### </TABLE>

Amortization expense for amortizable intangible assets for each of the five succeeding fiscal years is estimated to be (amounts in thousands):

Years Ending December 31,	
,	
2006	\$ 1,626
2007	1,538
2008	1,287
2009	963
2010	112

### (4) Share-Based Compensation

Our 1986 Stock Option Plan, as amended, provides for the grant of options for a maximum of 13.2 million shares of GCI Class A common stock, subject to adjustment upon the occurrence of stock dividends, stock splits, mergers, consolidations or certain other changes in corporate structure or capitalization. If an option expires or terminates, the shares subject to the option will be available for further grants of options under the Option Plan. Substantially all stock options granted vest in equal installments over a period of five years, and expire ten years from the date of grant. New shares are issued when stock option agreements are exercised, unless the stock option agreements are settled in cash. Our share repurchase program as described in note 11 in the "Notes to Consolidated Financial Statements" included in Part II of our December 31, 2005 annual report on Form 10-K may include the purchase of shares issued pursuant to stock option agreement exercise transactions. We are unable to estimate the number of such shares we may purchase during the annual period beginning July 1, 2006.

We use a Black-Scholes-Merton option pricing model to estimate the fair value of share-based awards under SFAS 123(R), which is the same valuation technique we previously used for pro forma disclosures under SFAS 123. The Black-Scholes-Merton option pricing model incorporates various and highly subjective assumptions, including expected term and expected volatility. We have reviewed our historical pattern of option exercises and have determined that meaningful differences in option

exercise activity existed among employee job categories. Therefore, for all stock options granted after January 1, 2006, we have categorized these awards into two groups of employees for valuation purposes, which is the same technique we previously used for pro forma disclosures under SFAS 123.

19 (Continued)

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES
Notes to Interim Condensed Consolidated Financial Statements
(Unaudited)

We estimated the expected term of options granted by evaluating the vesting period of stock option awards, employee's past exercise and post-vesting employment departure behavior, and expected volatility of the price of the underlying shares.

We estimated the expected volatility of our common stock at the grant date using the historical volatility of our common stock over the most recent period equal to the expected stock option term and evaluated the extent to which available information indicated that future volatility may differ from historical volatility.

The risk-free interest rate assumption was determined using the Federal Reserve nominal rates for U.S. Treasury zero-coupon bonds with maturities similar to those of the expected term of the award being valued. We have never paid any cash dividends on our common stock and we do not anticipate paying any cash dividends in the foreseeable future. Therefore, we assumed an expected dividend yield of zero.

The following table shows our assumptions used to compute the share-based compensation expense and pro forma information for stock options granted during the six months ended June 30, 2006 and 2005:

	2006	2005
Expected term (years)	5.0 - 5.9	5.0 - 5.2
Volatility	54.7% - 57.6%	44.8% - 45.5%
Risk-free interest rate	4.8% - 5.1%	3.7% - 4.2%

SFAS 123(R) requires us to measure share-based compensation liability instruments at fair value as of January 1, 2006. Previously, we measured those liability instruments at their intrinsic value determined as of their grant date. The transition impact of adopting SFAS No. 123(R) attributed to measuring such liability instruments at fair value totaled approximately \$1.1 million, net of income tax benefit of \$469,000 and is reported as a component of the cumulative effect of a change in accounting principle in the accompanying Consolidated Statement of Operations for the six months ended June 30, 2006.

Additionally, SFAS 123(R) requires us to estimate pre-vesting option forfeitures at the time of grant and periodically revise those estimates in subsequent periods if actual forfeitures differ from those estimates. We record share-based compensation expense only for those awards expected to vest using an estimated forfeiture rate based on our historical pre-vesting forfeiture data. Previously, we accounted for forfeitures as they occurred under the pro forma disclosure provisions of SFAS 123 for periods prior to 2006. The transition impact of adopting SFAS No. 123(R), attributed to accruing for expected forfeitures on outstanding share-based awards, totaled \$108,000, net of income tax expense of \$44,000 and is reported as a component of the cumulative effect of a change in accounting principle in the accompanying Consolidated Statement of Operations for the six months ended June 30, 2006.

The weighted average grant date fair value of options granted during the six months ended June 30, 2006 and 2005 was \$6.46 per share and \$4.55 per share, respectively. The total fair value of options vesting during

20 (Continued)

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES

Notes to Interim Condensed Consolidated Financial Statements

(Unaudited)

the six months ended June 30, 2006 and 2005 was approximately \$2.2 million and \$2.9 million, respectively.

Unrecognized share-based compensation expense was approximately \$12.6 million as of June 30, 2006, relating to a total of 3.2 million unvested stock options. We expect to recognize this share-based compensation expense over a weighted average period of approximately 3.3 years.

The following is a summary of our 1986 Stock Option Plan activity for the six months ended June 30, 2006:

	Shares	Weighted Average Exercise Price
<\$>	<c></c>	<c></c>
Outstanding at December 31, 2005	6,550,777	\$7.27
Granted	931,950	\$12.07
Exercised	(1,165,692)	\$6.61
Forfeited	(51,900)	\$8.96
Outstanding at June 30, 2006	6,265,135	\$8.09
Available for grant at June 30, 2006	440,115	
	===========	

</TABLE>

The following is a summary of activity for stock options granted not pursuant to the 1986 Stock Option Plan for the six months ended June 30, 2006:

<TABLE> <CAPTION>

	Shares	Weighted Average Exercise Price
<\$>	<c></c>	<c></c>
Outstanding at December 31, 2005 and June 30, 2006	250,000	\$6.50
Available for grant at June 30, 2006		
	============	

</TABLE>

21 (Continued)

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES

Notes to Interim Condensed Consolidated Financial Statements
(Unaudited)

The following is a summary of all outstanding stock options at June 30, 2006:  $\tt <TABLE> < CAPTION>$ 

# Options Outstanding

Range of Exercise Prices	Shares	Weighted Average Remaining Contractual Life (years)	Weighted Average Exercise Price	Aggregate Intrinsic Value (thousands)
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>
\$3.11-\$6.00	1,013,651	4.97	\$5.49	\$6 <b>,</b> 756
\$6.01-\$6.35	52,000	4.53	\$6.13	\$307
\$6.36-\$6.50	1,287,889	4.03	\$6.50	\$7 <b>,</b> 277
\$6.51-\$7.00	201,234	3.69	\$7.00	\$1,037
\$7.01-\$7.25	1,125,000	5.69	\$7.25	\$5 <b>,</b> 513
\$7.26-\$8.40	691 <b>,</b> 993	5.98	\$8.01	\$2 <b>,</b> 863
\$8.41-\$9.75	793 <b>,</b> 520	8.36	\$9.40	\$2,183
\$9.76-\$11.50	976,348	9.35	\$10.88	\$1 <b>,</b> 239
\$11.51-\$11.95	1,000	9.75	\$11.95	\$
\$11.96-\$13.11	372,500	9.92	\$13.11	\$
\$3.11-\$13.11	6,515,135	6.33	\$8.03	\$27,175

</TABLE>
<TABLE>
<CAPTION>

Options Vested

Range of Exercise Prices	Shares	Weighted Average Remaining Contractual Life (years)	Weighted Average Exercise Price	Aggregate Intrinsic Value (thousands)
<\$>	<c></c>	<c></c>	<c></c>	<c></c>
\$3.11-\$6.00	701,501	4.33	\$5.31	\$4 <b>,</b> 799
\$6.01-\$6.35	51,200	4.50	\$6.13	\$305
\$6.36-\$6.50	1,185,089	3.80	\$6.50	\$6 <b>,</b> 696
\$6.51-\$7.00	200,834	3.69	\$7.00	\$1,035
\$7.01-\$7.25	549 <b>,</b> 995	5.64	\$7.25	\$2 <b>,</b> 695
\$7.26-\$8.40	449,293	4.94	\$7.81	\$1,948
\$8.41-\$9.75	129,333	6.49	\$9.02	\$404
\$9.76-\$11.50	53,198	6.19	\$10.79	\$72
\$11.51-\$11.95			\$	\$
\$11.96-13.11			\$	\$
\$3.11-\$13.11	3,320,443	4.52	\$6.74	\$17 <b>,</b> 954

types of customers.

The total intrinsic value, determined as of the date of exercise, of options exercised in the six months ended June 30, 2006 and 2005 were \$5.8 million and \$284,000, respectively. We received \$7.7 million and \$469,000 in cash from stock option exercises in the six months ended June 30, 2006 and 2005, respectively. We used cash of \$4.9 million and \$0 to settle stock option agreements in the six months ended June 30, 2006 and 2005, respectively.

22 (Continued)

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES
Notes to Interim Condensed Consolidated Financial Statements
(Unaudited)

(5) Industry Segments Data Our reportable segments are business units that offer different products. The reportable segments are each managed separately and serve distinct

In the fourth quarter of 2005 we committed to a reorganization plan to more efficiently meet the demands of technological and product convergence by realigning along customer lines rather than product lines. Beginning January 1, 2006 our four reportable segments became Consumer, Network Access, Commercial and Managed Broadband, replacing the Long-distance, Cable, Local Access and Internet services segments.

Reportable segment data and All Other Category data for the six months ended June 30, 2005 have been reclassified for comparability purposes, as follows:

- o Revenue and cost of goods sold (exclusive of depreciation and amortization expense) ("Cost of Goods Sold") for the six months ended June 30, 2005 were reclassified to conform to the new segment organizational structure. A combination of specific identification and general allocations were employed to reclassify amounts for the six months ended June 30, 2005. Allocated amounts were generally determined using segment revenue or customer counts derived from first quarter of 2006 segment data. We believe the first quarter of 2006 division of revenue and customers by segment is representative of the customer composition for the six months ended June 30, 2005 for purposes of reclassifying revenue and Cost of Goods Sold amounts for the six months ended June 30, 2005.
- o Selling, general and administrative ("SG&A") expenses for the six months ended June 30, 2005 were reclassified to conform to the new segment organizational structure. A combination of specific and general allocations were employed to reclassify amounts for the six months ended June 30, 2005, as follows:
  - Certain SG&A expenses for the six months ended June 30, 2006 were directly charged to each new reportable segment. The amount of comparable SG&A directly charged to each segment during the six months ended June 30, 2005 based upon our new organizational structure is not practicable to calculate. We believe the 2006 amounts are representative of the amounts allocable during the same period of 2005, and therefore allocated such amounts to each reportable segment in the six months ended June 30, 2005.
  - o The remaining SG&A expenses, consisting of corporate related expenses further described below, were allocated to each segment using the percentage of each segment's margin for the year ended December 31, 2004 to total margin for the same period.
- Bad debt recovery for the six months ended June 30, 2005 was reclassified to conform to the new segment organizational structure. A combination of specific identification and general allocations based upon segment revenue for the year ended December 31, 2005 were employed to reclassify bad debt recovery for the six months ended June 30, 2005.

23 (Continued)

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES
Notes to Interim Condensed Consolidated Financial Statements
(Unaudited)

Depreciation and amortization expense for the six months ended June 30, 2005 is no longer allocated to reportable segments as our Chief Operating Decision Maker now evaluates each segments' performance based upon its earnings from operations before depreciation, amortization, net interest expense and income tax expense.

Consumer. We offer a full range of voice, video, data and wireless services to residential customers.

Network Access. We offer a full range of voice and data services to common carrier customers.

Commercial. We offer a full range of voice, video, data and wireless services to business and governmental customers.

Managed Broadband. We offer data services to rural school districts, rural hospitals and health clinics through our SchoolAccess(R) and Rural Health initiatives.

Corporate related expenses including engineering, operations and maintenance of our core network, information technology, accounting, legal and regulatory, human resources, and other general and administrative expenses for the six months ended June 30, 2006 are allocated to our segments using segment margin for the year ended December 31, 2005. Bad debt expense for the six months ended June 30, 2006 is allocated to our segments using a combination of specific identification and allocations based upon segment revenue for the six months ended June 30, 2006.

We evaluate performance and allocate resources based on earnings from operations before depreciation and amortization expense, net interest expense and income tax expense. The accounting policies of the reportable segments are the same as those described in the summary of significant accounting policies in note 1 in the "Notes to Consolidated Financial Statements" included in Part II of our December 31, 2005 annual report on Form 10-K. Intersegment sales are recorded at cost plus an agreed upon intercompany profit.

We earn all revenues through sales of services and products within the United States. All of our long-lived assets are located within the United States of America, except approximately 82% of our undersea fiber optic cable systems which transit international waters and all of our satellite transponders.

24 (Continued)

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES

Notes to Interim Condensed Consolidated Financial Statements

(Unaudited)

Summarized financial information for our reportable segments for the three and six months ended June 30, 2006 and 2005 follows (amounts in thousands):

<TABLE> <CAPTION>

VO111 1 1 V				Network	Commer-	Managed	Total
Report	able Three months ended June 30,		Consumer	Access	cial	Broadband	Segments
	<\$> 2006	<c></c>		<c></c>	<c></c>	<c></c>	<c></c>
	Revenues: Intersegment External	\$	44,223	41,377	1,386 26,013	 6 <b>,</b> 607	1,386 118,220
	Total revenues	\$	44,223	41,377	27,399	6 <b>,</b> 607	119,606
	Earnings from operations before depreciation, amortization, net interest expense and income taxes	\$	7,878	22,812	5,156	2,053	37 <b>,</b> 899
	2005						
	Revenues: Intersegment External	\$	15 40,349	1,607 36,907	26,407	39 7 <b>,</b> 002	110,665
	Total revenues	\$	40,364	38,514	32,573	7,041	118,492

interest expense and income taxes \$ 7,228 20,569 7,061 1,549 36,407

</TABLE>

25 (Continued)

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES
Notes to Interim Condensed Consolidated Financial Statements
(Unaudited)

<TABLE> <CAPTION>

10111 11				Network	Commer-	Managed	Total
Report	able Six months ended June 30,		Consumer	Access	cial	Broadband	Segments
	<\$> 2006	<c></c>		<c></c>	<c></c>	<c></c>	<c></c>
	Revenues: Intersegment External	\$	 86,886	79 <b>,</b> 200	2,724 52,141	12,815	2,724 231,042
	Total revenues	\$			54,865	12,815	233,766
=====	Earnings from operations before depreciation, amortization, net interest expense and income taxes	\$	15 <b>,</b> 955	42,681	11,812	3 <b>,</b> 861	74,309
	2005						
	Revenues: Intersegment External	\$			11,989 51,766		15,496 217,175
	Total revenues	\$	80,570	74,490	63,755	13,856	232,671
=====	Earnings from operations before depreciation, amortization, net interest expense and income taxes				12,725	4,135	70,890
====== <td> E&gt;</td> <td>:=====</td> <td></td> <td>======</td> <td></td> <td></td> <td></td>	 E>	:=====		======			
<table:< td=""><td>A reconciliation of reportable segment follows (amounts in thousands):</td><td>rever</td><td>nues to con</td><td>solidated reve</td><td>nues</td><td></td><td></td></table:<>	A reconciliation of reportable segment follows (amounts in thousands):	rever	nues to con	solidated reve	nues		
<capti< td=""><td>ON&gt;</td><td></td><td></td><td>June 2006</td><td>ths Ended 30, 2005</td><td>Jun 2006</td><td>e 30, 2005</td></capti<>	ON>			June 2006	ths Ended 30, 2005	Jun 2006	e 30, 2005
	<s></s>		<c:< td=""><td></td><td><c></c></td><td><c></c></td><td><c></c></td></c:<>		<c></c>	<c></c>	<c></c>
	Reportable segment revenues Less intersegment revenues eliminate consolidation	ed in	\$	119,606	118,492 7,827	233,766	232,671 15,496
	Consolidated revenues		\$ =	118,220	110,665	231,042	217,175
			_				

</TABLE>

26 (Continued)

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES

Notes to Interim Condensed Consolidated Financial Statements
(Unaudited)

A reconciliation of reportable segment earnings from operations before depreciation and amortization expense, net interest expense and income taxes to consolidated net income before income taxes and cumulative effect of a change in accounting principle follows (amounts in thousands):

	Three Months Ended June 30,		Six Months Ended June 30,	
	2006	2005	2006	2005
<\$>	<c></c>	<c></c>	<c></c>	<c></c>
Reportable segment earnings from operations before depreciation and amortization expense, net interest expense and income				
taxes	\$ 37 <b>,</b> 899	36,407	74,309	70 <b>,</b> 890
Less depreciation and amortization expense	20,172	18,348	40,333	36,052
Less other income	282		169	
Consolidated operating income	17,445	18,059	33,807	34,838
Less other expense, net	8,183	8 <b>,</b> 739	16,944	17,375
Consolidated net income before income taxes and cumulative effect of a				
change in accounting principle	\$ 9,262 =======	9,320 = =======	16,863 ======	17,463

#### </TABLE>

#### (6) Commitments and Contingencies

Litigation, Disputes, and Regulatory Matters
We are involved in various other lawsuits, billing disputes, legal
proceedings, and regulatory matters that have arisen from time to time in
the normal course of business. While the ultimate results of these items
cannot be predicted with certainty we do not expect at this time the
resolution of them to have a material adverse effect on our financial
position, results of operations or liquidity.

Capital Lease Obligation

On March 31, 2006, through our subsidiary GCI Communication Corp. ("GCC") we entered into an agreement to lease transponder capacity on Intelsat, Ltd.'s ("Intelsat" acquired PanAmSat Corporation on June 30, 2006) Galaxy 18 spacecraft that is expected to be launched during 2007. We will also lease capacity on the Horizons 1 satellite, which is owned jointly by Intelsat and JSAT International, Inc. The leased capacity is expected to replace our existing transponder capacity on Intelsat's Galaxy 10R satellite when it reaches its end of life.

We will lease, subject to a termination option, C-band and Ku-Band transponders over an expected term of approximately 14 years once the satellite is placed into commercial operation in its assigned orbital location, and the transponders meet specific performance specifications and are made available for our use. The present value of the lease payments, excluding telemetry, tracking and command services and back-up protection, is expected to total \$77.0 million to \$82.0 million. We will

27 (Continued)

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES

Notes to Interim Condensed Consolidated Financial Statements

(Unaudited)

record the capital lease obligation and the addition to our Property and Equipment when the satellite is made available for our use which is expected to occur approximately one month after the expected July 2007 launch.

A summary of estimated future minimum lease payments for this lease follows (amounts in thousands):

Years ending December 31:	
2006	\$ 
2007	4,584
2008	9,168
2009	9,168
2010	9,168
2011 and thereafter	96,264
Total minimum lease payments	\$ 128,352

Upon payment of a monthly fee we have the option to terminate the lease of the C-band transponders through June 1, 2007. We may forfeit our termination option at which time we would no longer be obligated to continue paying the monthly fee. If we elect to terminate our C-band transponder lease we must return the transponders and pay a termination

Telecommunication Services Agreement and Capacity Leases We leased a portion of our 800-mile fiber optic system capacity that extends from Prudhoe Bay to Valdez via Fairbanks, and provided management and maintenance services for this capacity to a significant customer. The lessee signed a contract with a competitor in March 2005, started the transition of its circuits from our fiber optic cable system to our competitor's microwave system in June 2006, and plans to have the transition complete by November 2006. We expect to sign an agreement with our competitor to lease capacity on our fiber optic cable system and provide certain other services in association with their contract. We expect this transition to result in a \$9.5 million annual decrease in the data component of Commercial Services segment revenue when it is completed with a decrease of approximately one-half the annual decrease in the year ended December 31, 2006 depending upon the pace of the transition.

The following summary of minimum future service revenues reflects the termination of our contract and includes the expected revenue from the new agreement we expect to sign with our competitor in addition to our other operating lease agreements included in the summary in note 16 included in Part II of our December 31, 2005 annual report on Form 10-K (amounts in thousands):

2011 and thereafter	52,700	
2010	5,574	
2009	8,154	
	10,734	
2008	10,734	
2007	11,694	
2006	\$ 19,301	
Years ending December 31,		

28 (Continued)

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES

Notes to Interim Condensed Consolidated Financial Statements

(Unaudited)

Anchorage Unbundled Network Elements Arbitration
On June 25, 2004, the Regulatory Commission of Alaska ("RCA") issued a comprehensive decision setting forth new rates for unbundled network elements ("UNE"), resale, and terms and conditions for interconnection in the Anchorage arbitration. Significantly, the RCA raised the loop rate in Anchorage to \$19.15 but subsequently reduced the loop rate on reconsideration to \$18.64. The RCA also issued other various arbitration rulings adverse to us, including adopting Alaska Communications Systems Group, Inc.'s ("ACS") non-recurring and collocation cost models. On December 7, 2004, the Commission issued a final order approving an interconnection agreement. We have appealed various Commission arbitration rulings.

On September 30, 2005, the ACS subsidiary serving Anchorage filed a petition with the Federal Communications Commission ("FCC"), seeking forbearance from the requirement that it provide access to UNEs, and that to the extent it voluntarily did so, that the pricing provisions of the Act would not apply. We filed our opposition on January 9, 2006 and our reply on February 23, 2006. The FCC is required under statute to issue a decision by September 30, 2006, or on its own motion, in an additional 90 days. If a decision is not issued within the required timeframe, the petition is deemed granted. The ability to obtain UNEs is an important element of our local exchange and exchange access services business, and the outcome of this proceeding could result in a change in our ability to access segments of the Anchorage market via the facilities of the ILEC and the cost of doing so. We cannot predict at this time the outcome of this proceeding or its impact on us; however, our net cost of providing local telephone services in Anchorage could be materially adversely affected by an adverse decision.

Alaska DigiTel, LLC ("Alaska DigiTel") Investment We agreed to invest approximately \$29.5 million in exchange for a majority equity interest in Alaska DigiTel, a small Alaska Personal Communication Services ("PCS") provider. The existing owners will retain a minority ownership interest and voting control of Alaska DigiTel. The exact percentage and dollar amounts for our interest in Alaska DigiTel will vary in proportion to the amount the existing owners elect to retain, but we expect to own between 75% and 85% after completion of the transaction. The transaction is based on a post closing enterprise valuation of \$37.0 million for Alaska DigiTel. We will fund the transaction from cash on hand, by drawing down additional debt, or a combination of the two. In June 2006, we entered into a Reorganization Agreement with the members of Alaska DigiTel and certain other parties setting forth the formal terms and conditions the principal terms of which are the same as those in the superseded memorandum of understanding. GCI, Denali PCS, LLC and Alaska DigiTel ("Alaska DigiTel Applicants") jointly filed applications with the FCC seeking the requisite regulatory consent to the parties' transaction. Matanuska

Telephone Association filed a Petition to Deny the applications with the FCC in February 2006. ACS Wireless, Inc. filed an objection to the applications with the FCC in July 2006. The Alaska DigiTel Applicants are opposing these adverse filings. It is uncertain when the FCC will act upon the contested applications. We are uncertain when this transaction will close.

29 (Continued)

GENERAL COMMUNICATION, INC. AND SUBSIDIARIES

Notes to Interim Condensed Consolidated Financial Statements

(Unaudited)

We have provided a \$3.0 million bank depository account as collateral for an Alaska DigiTel term loan. The amount is classified as Cash and Cash Equivalents on our June 30, 2006 Consolidated Balance Sheet.

Indefeasible Right of Use ("IRU") Commitment
On July 31, 2006, through our subsidiary GCC we entered into an
agreement to purchase an IRU in the Kodiak-Kenai Cable Company, LLC's
marine-based fiber optic cable system linking Anchorage to Kenai, Homer,
Kodiak and Seward, Alaska. The new system is under construction and we
anticipate it will be placed into service in the first quarter of 2007.
We have committed to purchase a minimum of \$8.0 million to \$8.5 million
in IRU capacity in three installments through 2011.

30

PART I. ITEM 2.

# MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

In the following discussion, General Communication, Inc. and its direct and indirect subsidiaries are referred to as "we," "us" and "our."  $\,$ 

Management's Discussion and Analysis of Financial Condition and Results of Operations discusses our consolidated financial statements, which have been prepared in accordance with accounting principles generally accepted in the United States of America. The preparation of these financial statements requires us to make estimates and assumptions that affect the reported amounts of assets and liabilities and the disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. On an on-going basis, we evaluate our estimates and judgments, including those related to unbilled revenues, Cost of Goods Sold accruals, allowance for doubtful accounts, share-based compensation expense, depreciation, amortization and accretion periods, intangible assets, income taxes, and contingencies and litigation. We base our estimates and judgments on historical experience and on various other factors that are believed to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. Actual results may differ from these estimates under different assumptions or conditions. See also our "Cautionary Statement Regarding Forward-Looking Statements."

# General Overview

Through our focus on long-term results, acquisitions, and strategic capital investments, we strive to consistently grow our revenues and expand our margins. We have historically met our cash needs for operations, regular capital expenditures and maintenance capital expenditures through our cash flows from operating activities. Historically, cash requirements for significant acquisitions and major capital expenditures have been provided largely through our financing activities.

As of January 1, 2006 we were reorganized under Consumer, Network Access, Commercial and Managed Broadband reportable segments, replacing the Long-distance, Cable, Local Access and Internet services reportable segments. The realignment along customer lines rather than product lines allows us to more efficiently meet the demands of technological and product convergence.

Segment and All Other category data for the six and three months ended June 30, 2005 have been reclassified to reflect the organizational changes for comparability purposes. A combination of specific identification and general allocations were employed to reclassify 2005 balances. Allocated amounts were generally determined using segment revenue or customer counts derived from first quarter of 2006 segment data. We believe the first quarter of 2006 division of revenue and customers by segment is representative of the three and six months ended June 30, 2005 customer composition for purposes of reclassifying 2005 revenue and Cost of Goods Sold balances.

31

The Network Access segment provides services to other common carrier customers and the Managed Broadband segment provides services to rural school districts and rural hospitals and health clinics. Following are our segments and the services and products each offers to its customers:

#### Reportable Segments

	Services and Products	Consumer	Network Access	Commercial	Managed Broadband
<s> Voice</s>		<c></c>	<c> X</c>	<c></c>	<c></c>
Video		X		Х	
Data		X	Х	Х	X
Wireles	58 	X		Х	

#### </TABLE>

An overview of our services and products follows.

Voice Services and Products Long-distance Services

We generate long-distance services revenues from monthly plan fees and usage charges.

Factors that have the greatest impact on year-to-year changes in long-distance services revenues include the rate per minute charged to customers and usage volumes expressed as minutes of use.

Common carrier traffic routed to us for termination in Alaska is largely dependent on traffic routed to our common carrier customers by their customers. Pricing pressures, new program offerings, and market and business consolidations continue to evolve in the markets served by our other common carrier customers. If, as a result, their traffic is reduced, or if their competitors' costs to terminate or originate traffic in Alaska are reduced, our traffic will also likely be reduced, and our pricing may be reduced to respond to competitive pressures, consistent with federal law. Additionally, disruption in the economy resulting from terrorist attacks and other attacks or acts of war could affect our carrier customers. We are unable to predict the effect on us of such changes. However, given the materiality of other common carrier revenues to us, a significant reduction in traffic or pricing could have a material adverse effect on our financial position, results of operations and liquidity.

Due in large part to the favorable synergistic effects of our bundling strategy focused on consumer and commercial customers, long-distance services continues to be a significant contributor to our overall performance, although the migration of traffic from our voice products to our data and wireless products continues.

Our long-distance service faces significant competition from AT&T Alascom, long-distance resellers, and local telephone companies that have entered the long-distance market. We believe our approach to developing, pricing, and providing long-distance services and bundling different business segment services will continue to allow us to be competitive in providing those services.

## Local Access Services

We generate local access services revenues from three primary sources: (1) basic dial tone services; (2) private line and special access services; and (3) features and other charges, including voice mail, caller ID, distinctive ring, inside wiring and subscriber line charges.

The primary factors that contribute to year-to-year changes in local access services revenues include the average number of subscribers to our services

32

during a given reporting period, the average monthly rates charged for non-traffic sensitive services, the number and type of additional premium features selected, the traffic sensitive access rates charged to carriers and the Universal Service Program.

We estimate that our June 30, 2006 and 2005 total lines in service represent a statewide market share of approximately 26% and 24%, respectively. At June 30, 2006 and 2005 approximately 86% and 85%, respectively, of our lines are provided on our own facilities and leased local loops. At June 30, 2006 and 2005 approximately 7% and 6%, respectively, of our lines are provided using the UNE platform delivery method.

Our local access service faces significant competition in Anchorage, Fairbanks, and Juneau from ACS, which is the largest incumbent local exchange carrier ("ILEC") in Alaska, and from AT&T Alascom, Inc. ("Alascom") in Anchorage for Consumer services. Alascom has received certification from the Regulatory Commission of Alaska to provide local access services in Fairbanks and Juneau. We believe our approach to developing, pricing, and providing local access services and bundling different services will allow us to be competitive in providing those services.

On September 23, 2005, February 2, 2006 and July 18, 2006, the RCA issued orders granting us certification to serve the service areas of Ketchikan Public Utilities, Cordova Telephone Cooperative, Copper Valley Telephone Cooperative, Matanuska Telephone Cooperative, the Glacier State area served by ACS of the Northland, Alaska Telephone Company, Interior Telephone Company, United-KUC and Mukluk Telephone Company. The affected rural local exchange carriers have appealed various aspects of the certification rulings.

We plan to offer service in these new areas using a combination of methods. To a large extent, we plan to use our existing cable network to deliver local services. Where we do not have cable plant, we may use wireless technologies and resale of other carrier's services. We may lease portions of an existing carrier's network or seek wholesale discounts, but our application is not dependent upon access to either UNEs of the Incumbent Local Exchange Carrier's ("ILEC") network or wholesale discount rates for resale of ILEC services.

On May 2, 2005 we tendered an interconnection request to the City of Ketchikan d/b/a Ketchikan Public Utilities ("KPU"), which had been authorized by the RCA to provide video programming services through its KPU CommVision division on April 26, 2005. Under the terms of Section 251(f)(1)(C) of the Telecommunications Act of 1996 KPU's current rural exemption from negotiation will be forfeited if, and when, KPU commences offering video programming. On June 3, 2005, we entered into a stipulation with KPU recognizing that KPU will forfeit its rural exemption and that negotiations for interconnection would commence when KPU commences offering video programming. Negotiations began in December 2005 and culminated in a contract in June 2006. The Ketchikan City Council approved the contract in June 2006 and the contract will be submitted to the RCA for final approval in August 2006.

We plan to have deployed more than 35,000 additional digital local phone service ("DLPS") lines which utilize our Anchorage coaxial cable facilities by December 31, 2006. This service delivery method allows us to utilize our own cable facilities to provide local access service to our customers and avoid paying local loop charges to the ILEC.

#### Directory Advertising

We sell advertising in our yellow pages directories to commercial customers and distribute white and yellow pages directories to customers in certain markets we serve. We also sell on-line directory products.

33

## Video Services and Products

We generate cable services revenues from three primary sources: (1) digital and analog programming services, including monthly basic and premium subscriptions, pay-per-view movies and one-time events, such as sporting events; (2) equipment rentals and installation; and (3) advertising sales.

Our cable systems serve 40 communities and areas in Alaska, including the state's four largest population centers, Anchorage, Fairbanks, the Matanuska-Susitna Valley and the Kenai Peninsula.

The primary factors that contribute to period-to-period changes in cable services revenues include average monthly subscription rates and pay-per-view buys, the mix among basic, premium and digital tier services, the average number of cable television subscribers during a given reporting period, set-top box utilization and related rates, revenues generated from new product offerings, and sales of cable advertising services.

We increased rates charged for certain cable services in eleven communities, including three of the state's four largest population centers, Anchorage, the Matanuska-Susitna Valley, and Fairbanks. The rate increases were primarily effective in January 2006 and increased approximately 5% for those customers who experienced an adjustment.

### Data Services and Products

Internet Services

We generate Internet services revenues from two primary sources: (1) access product services, including cable modem and dial-up access; and (2) network management services.

The primary factors that contribute to year-to-year changes in Internet services revenues include the average number of subscribers to our services during a given reporting period, the average monthly subscription rates, the amount of bandwidth purchased by large commercial customers, and the number and type of additional premium features selected.

Marketing campaigns continue to be deployed featuring bundled products. Our Internet offerings are bundled with various combinations of our long-distance, cable, and local access services and provide free or discounted basic or premium Internet services. Value-added premium Internet features are available for additional charges.

We compete with a number of Internet service providers in our markets. We believe our approach to developing, pricing, and providing Internet services

allows us to be competitive in providing those services.

Private Line and Private Network Services

We generate private line and private network services revenue from two primary sources: (1) leasing capacity on our facilities that utilize voice and data transmission circuits, dedicated to particular subscribers, which link a device in one location to another in a different location and (2) through the sale of Internet Protocol based data services on a secured shared network to businesses linking multiple enterprise locations. The factor that has the greatest impact on year-to-year changes in private line and private network services revenues is the number of private lines and private networks in use. We compete against Alascom, ACS and other local telecommunication service providers.

34

#### Managed Services

We design, sell, install, service and operate, on behalf of certain customers, communications and computer networking equipment and provide field/depot, third party, technical support, communications consulting and outsourcing services. We also supply integrated voice and data communications systems incorporating interstate and intrastate digital private lines, point-to-point and multipoint private network and small earth station services. There are a number of competing companies in Alaska that actively sell and maintain data and voice communications systems.

Our ability to integrate communications networks and data communications equipment has allowed us to maintain our market position based on "value added" support services rather than price competition. These services are blended with other transport products into unique customer solutions, including managed services and outsourcing.

#### Broadband Services

We generate broadband services revenue through our SchoolAccess(R) and Rural Health initiatives. Our customers may purchase end-to-end broadband services solutions blended with other transport and software products such as video conferencing and unique web content services. There are several competing companies in Alaska that actively sell broadband services.

Our ability to provide end-to-end broadband services solutions has allowed us to maintain our market position based on "value added" products and services rather than solely based on price competition. These services are blended with other transport and software products into unique customer solutions, including SchoolAccess(R) and Rural Health applications such as video conferencing and unique web content services.

## Wireless Services and Products

We generate wireless services and equipment revenues from four primary sources: (1) monthly plan fees; (2) usage and roaming charges; (3) wireless Internet access; and (4) handset and accessory sales.

We offer wireless services by reselling Dobson Communications Corporation's ("Dobson") services. We provide limited wireless local access and Internet services using our own facilities. We compete against Dobson, ACS, Alaska DigiTel, and resellers of those services in Anchorage and other markets.

We have 22,900 and 12,200 combined Consumer and Commercial wireless lines in service at June 30, 2006 and 2005, respectively. A wireless line in service is defined as a revenue generating wireless device. Our average wireless revenue per combined Consumer and Commercial subscriber is \$53.90 and \$52.62 during the three and six months ended June 30, 2006, respectively, calculated by dividing our combined Consumer and Commercial usage revenues by our combined Consumer and Commercial subscriber count.

We have reached a definitive agreement to invest \$29.5 million in Alaska DigiTel. In exchange for the investment, we will receive a majority equity interest in Alaska DigiTel but will not own voting control of the venture. We view our investment as an incremental way to participate in future growth of the wireless industry in Alaska. Our existing distribution agreement with Dobson remains in full effect and our existing wireless products will continue to compete with Alaska DigiTel in the Alaska market. The transaction is subject to customary closing conditions, including documentation and regulatory approvals. The Alaska DigiTel Applicants jointly filed applications with the FCC seeking the requisite regulatory consent to the parties' transaction. Matanuska Telephone Association filed a Petition to Deny the applications with the FCC in February 2006. ACS Wireless, Inc. filed an objection to the applications with

35

the FCC in July 2006. The Alaska DigiTel Applicants are opposing these adverse filings. It is uncertain when the FCC will act upon the contested applications. We are uncertain when this transaction will close.

31

Results of Operations

The following table sets forth selected Statements of Operations data as a percentage of total revenues for the periods indicated (underlying data rounded

to the nearest thousands): <TABLE> <CAPTION>

Percent-Percent-

aσe

age

Change (1)

Change (1)			Change (1)			
Change (1)	Three Mor	nths Ended	2006	Six Mont	ths Ended	
2006	Tily	ne 30,	VS.	June	3.0	
VS.	Jui	ie 30,	vs.	June	= 30,	
(Unaudited)	2006	2005	2005	2006	2005	
2005						
<pre><s> Statements of Operations Data:</s></pre>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>
Revenues:						
Consumer services segment 7.9%	37.4%	36.5%	9.6%	37.6%	37.1%	
Network Access services segment	35.0%	33.3%	12.1%	34.3%	32.7%	
11.5% Commercial services segment	22.0%	23.9%	(1.5%)	22.6%	23.8%	
0.7%	22.05	23.96	(1.36)	22.05	23.05	
Managed broadband services segment	5.6%	6.3%	(5.7%)	5.5%	6.4%	
(7.3%)						
Total revenues 6.4%	100.0%	100.0%	6.8%	100.0%	100.0%	
Selling, general and administrative						
expenses 6.8%	34.4%	34.4%	7.0%	34.7%	34.6%	
Bad debt expense (recovery)	1.1%	0.2%	589.7%	0.8%	(0.1%)	
1,256.6%	45.40	4.5.50	4.0.00	45.50	4.5.50	
Depreciation and amortization expense 11.9%	17.1%	16.6%	10.0%	17.5%	16.6%	
Operating income	14.4%	16.3%	(6.1%)	14.4%	16.0%	
(4.3%)  Net income before income taxes and						
cumulative effect of a change in						
accounting principle $(6.2\%)$	7.4%	8.4%	(5.8%)	7.1%	8.0%	
Net income before cumulative effect						
of a change in accounting principle	4.4%	4.8%	(2.7%)	3.9%	4.6%	
(8.9%) Net income	4.4%	4.8%	(2.7%)	3.7%	4.6%	
(15.0%)			(= /			
<fn></fn>						

_____

(1) Percentage change in underlying data.

_____

</FN> </TABLE>

Three Months Ended June 30, 2006 ("second quarter of 2006") Compared To Three Months Ended June 30, 2005 ("second quarter of 2005")

Overview of Revenues and Cost of Goods Sold Total revenues increased 6.8% from \$110.7 million in the second quarter of 2005 to \$118.2 million in the second quarter of 2006. Revenue increases in our Consumer and Network Access segments were partially off-set by decreased revenue in our Commercial and Managed Broadband segments. See the discussion below for more information by segment.

Total Cost of Goods Sold increased 7.1% from \$36.0 million in the second quarter of 2005 to \$38.6 million in the second quarter of 2006. Increases in the

Consumer, Network Access and Commercial segments' Cost of Goods Sold were partially off-set by decreased Cost of Goods Sold in our Managed Broadband segment. See the discussion below for more information by segment.

Consumer Services Segment Overview

Consumer services segment revenue in the second quarter of 2006 represented 37.4% of consolidated revenues. The components of Consumer services segment revenue are as follows (amounts in thousands):

<TABLE>

<CAPTION>

Second Quarter 2006 2005

-----

Percentage Change

<\$>	<b>/</b> C>	<c></c>	<0>
	<c></c>		<c></c>
Voice	\$ 11 <b>,</b> 451	11 <b>,</b> 593	(1.2%)
Video	22,329	21,142	5.6%
Data	7 <b>,</b> 258	6 <b>,</b> 321	14.8%
Wireless	3,185	1,293	146.3%
Total Consumer services segment revenue	\$ 44,223	40,349	9.6%
	=========		

-----

 $\mbox{\ensuremath{^{\prime}}{TABLE}}\mbox{\ensuremath{^{\prime}}{Selected}}$  key performance indicators for our Consumer services segment follow:

<CAPTION>

	June 30,		Percentage	
	2006	2005	Change	
<\$>	<c></c>	<c></c>	<c></c>	
Voice:				
Total local access lines in service(1)	67,700	68,200	(0.7%)	
DLPS local access lines in service(1)	25,300	12,400	104.0%	
Video:				
Basic subscribers(2)	121,900	121,200	0.6%	
Digital programming tier subscribers (3)	55,100	48,700	13.1%	
HD/DVR converter boxes(4)	18,800	7,400	154.1%	
Homes passed	217,100	211,000	2.9%	
Data:				
Cable modem subscribers (5)	75,000	64,300	16.6%	
<fn></fn>				

_____

- (1) A local access line in service is defined as a revenue generating circuit or channel connecting a customer to the public switched telephone network.
- (2) A basic cable subscriber is defined as one basic tier of service delivered to an address or separate subunits thereof regardless of the number of outlets purchased.
- (3) A digital programming tier subscriber is defined as one digital programming tier of service delivered to an address or separate subunits thereof regardless of the number of outlets or digital programming tiers purchased. Digital programming tier subscribers are a sub-set of basic subscribers.
- (4) An HD/DVR converter box is defined as one box rented by a digital programming or basic tier subscriber. A digital programming or basic tier subscriber is not required to rent an HD/DVR converter box to receive service.
- (5) A cable modem subscriber is defined by the purchase of cable modem service regardless of the level of service purchased. If one entity purchases multiple cable modem service access points, each access point is counted as a subscriber. Cable modem subscribers may also be basic subscribers though basic cable service is not required to receive cable modem service.

_____

</FN></TABLE>

38

<TABLE>

	Second Quarter		Percentage	
	2006	2005	Change	
<\$>	<c></c>	<c></c>	<c></c>	
Voice:				
Long-distance minutes carried (in millions)	35.9	40.5	(11.4%)	
Video:				
Average monthly gross revenue per subscriber(1)	\$60.92	\$59.22	2.9%	
<fn></fn>				

(1) Quarter-to-date average monthly consumer video revenues divided by the average of consumer video basic subscribers at the beginning and ending of the period.

-----

______

</FN></TABLE>

We have 92,800 Consumer long-distance customers at June 30, 2006. A long-distance customer is defined as a customer account that is invoiced a monthly long-distance plan fee or has made a long-distance call during the month. Due to our implementation of a new unified billing system on September 1,

month. Due to our implementation of a new unified billing 2005 this statistic is not available for June 30, 2005.

Consumer Services Segment Revenues

The 1.2% decrease in voice revenue is primarily due to the decrease in long-distance minutes carried for these customers. The decrease is partially off-set by a \$157,000 increase in support from the Universal Service Program.

The 5.6% increase in video revenue is primarily due to an increase in digital programming tier subscribers in the second quarter of 2006, the rate increases previously described, and a 23.2% increase in equipment rental revenue to \$3.2 million in the second quarter of 2006. The increase in equipment rental revenue is primarily caused by the increased use of digital distribution technology.

The 14.8% increase in data revenue is primarily due to a 16.7% increase in cable modem revenue to \$6.1 million in the second quarter of 2006 as compared to the second quarter of 2005. The increase in cable modem revenue is primarily due to the increase in subscribers.

The 146.3% increase in wireless revenue is primarily due to the increase in wireless subscribers.

Consumer Services Segment Cost of Goods Sold Consumer services segment Cost of Goods Sold increased 9.0% to \$17.1 million from the second quarter of 2005 to the second quarter of 2006 primarily due to

from the second quarter of 2005 to the second quarter of 2006 primarily due to increased video and wireless Cost of Goods Sold resulting from increased revenue. The increase in Cost of Goods Sold is partially off-set by decreased voice Cost of Goods Sold primarily due to cost savings resulting from the increased deployment of DLPS lines in the last six months of 2005 and the first six months of 2006.

Network Access Services Segment Overview
Network access services segment revenue

Network access services segment revenue in the second quarter of 2006 represented 35.0% of consolidated revenues. The components of Network Access services segment revenue are as follows (amounts in thousands):

<CAPTION>

	Second Quarter		Percentage	
		2006	2005	Change
<\$>	<c></c>		<c></c>	<c></c>
Voice	\$	27,844	23,940	16.3%
Data		13,533	12,967	4.4%
Total Network Access services segment revenue	\$	41,377	36,907	12.1%
	===		========	

</TABLE>

<CAPTION>

39

Selected key performance indicators for our Network Access services segment follow: <TABLE>

	Second Quarter		Percentage	
	2006	2005	Change	
<\$>	<c></c>	<c></c>	<c></c>	
Voice:				
Total local access lines in service (1)	3,300	3,600	(8.3)%	

<FN>

(1) A local access line in service is defined as a revenue generating circuit or channel connecting a customer to the public switched telephone network.

- -----

</FN>
</TABLE>

Notronh Accord

Network Access Services Segment Revenues

The 16.3% increase in voice revenue is primarily due to the increase in minutes carried for our other common carrier customers partially off-set by a 4.2% decrease in our rate per minute on minutes carried for other common carriers. The average rate per minute decrease is primarily due to the annual 3.0% rate decrease mandated by the Consolidated Appropriations Act for Fiscal Year 2005 effective January 2005 which will result in rate decreases of 3.0% per year through 2010 and a change in the composition of traffic carried.

Network Access Services Segment Cost of Goods Sold Network Access services segment Cost of Goods Sold increased 12.9% to \$8.8 million from the second quarter of 2005 to the second quarter of 2006 primarily due to increased voice minutes carried.

Commercial Services Segment Overview

Commercial services segment revenue in the second quarter of 2006 represented

22.0% of consolidated revenues. The components of Commercial services segment revenue are as follow (amounts in thousands):
<TABLE>

<CAPTION>

	Second Quarter		Percentage	
		2006	2005	Change
<\$>	<c></c>		<c></c>	<c></c>
Voice	\$	8,097	8,796	(7.9%)
Video		1,933	1,889	2.3%
Data		15,400	15,468	(0.4%)
Wireless		583	254	129.5%
Total Commercial services segment revenue	\$ ==	26,013 ======	26,407	(1.5%)

_____

</TABLE>

40

Selected key performance indicators for our Commercial services segment follow: <TABLE> <CAPTION>

	June 30,		Percentage	
	2006	2005	Change	
<\$>	<c></c>	<c></c>	<c></c>	
Voice:				
Total local access lines in service(1)	40,400	40,100	0.7%	
DLPS local access lines in service(1)	1,100	400	175.0%	
Data:				
Cable modem subscribers(2)	7,100	5,900	20.3%	
<fn></fn>				

- _____
- (1) A local access line in service is defined as a revenue generating circuit or channel connecting a customer to the public switched telephone network.
- (2) A cable modem subscriber is defined by the purchase of cable modem service regardless of the level of service purchased. If one entity purchases multiple cable modem service access points, each access point is counted as a subscriber.

-----

</FN></TABLE>

<TABLE>

	Second Quarter		Percentage	
	2006	2005	Change	
<\$>	<c></c>	<c></c>	<c></c>	
Voice:				
Long-distance minutes carried (in millions)	34.4	35.9	(4.2%)	

We have 11,700 Commercial long-distance customers at June 30, 2006. A long-distance customer is defined as a customer account that is invoiced a monthly long-distance plan fee or has made a long-distance call during the month. Due to our implementation of a new unified billing system on September 1, 2005 this statistic is not available for June 30, 2005.

We leased a portion of our 800-mile fiber optic system capacity that extends from Prudhoe Bay to Valdez via Fairbanks, and provided management and maintenance services for this capacity to a significant customer. The lessee signed a contract with a competitor in March 2005, started the transition of their circuits from our fiber optic cable system to our competitor's microwave system in June 2006, and plans to have the transition complete by November 2006. We expect to sign an agreement with our competitor to lease capacity on our fiber optic cable system in association with their contract. We expect this transition to result in a \$9.5 million annual decrease in the data component of Commercial Services segment revenue when it is completed with a decrease of approximately one-half the annual decrease in the year ended December 31, 2006 depending upon the pace of the transition.

Commercial Services Segment Revenues

The 7.9% decrease in voice revenue is primarily due to the decrease in minutes carried for our Commercial customers.

The 0.4% decrease in data revenue is primarily due to the following:

- o A \$475,000 or 84.8% decrease resulting from product sales in second quarter 2005 that did not recur in second quarter 2006, and
- o A \$65,000 or 0.9% decrease in managed services revenue primarily due to

a \$428,000 or 12.0% decrease in revenue earned from the lease and provision of management and maintenance services on a portion of our 800-mile fiber optic system capacity that extends from Prudhoe Bay to Valdez via Fairbanks as described above partially off-set by a \$363,000 or 10.3% increase in special project revenues.

The decrease is partially off-set by a \$494,000 or 15.6% increase in private line and private network services due to an increase in the number of circuits sold.

4 -

Commercial Services Segment Cost of Goods Sold
Commercial services segment Cost of Goods Sold increased 4.1% to \$11.6 million
from the second quarter of 2005 to the second quarter of 2006 primarily due to a
\$449,000 or 198.7% increase in wireless Cost of Goods Sold resulting from
increased revenue.

Managed Broadband Services Segment Overview
Managed broadband services segment revenue in the second quarter of 2006
represented 5.6% of consolidated revenues. Managed broadband services segment
revenue, which includes data products only, decreased \$397,000 to \$6.6 million
in the second quarter of 2006 as compared to the second quarter of 2005.

Selected key performance indicators for our Commercial services segment follow: <TABLE> <CAPTION>

	June 30,		Percentage	
	2006	2005	Change	
<\$>	<c></c>	<c></c>	<c></c>	
Managed broadband services:				
SchoolAccess(R) customers	45	43	4.7%	
Rural health customers	21	21	0.0%	

#### </TABLE>

Managed Broadband Services Segment Revenues

The decrease in Managed Broadband services segment revenue is primarily due to a decrease in our multi-site SchoolAccess(R) customers in the second quarter of 2006 as compared to 2005 and a decrease in the rate charged for certain circuits purchased by our rural health customers. An increase in single-site SchoolAccess(R) customers from which we generate less revenue was partially off-set by the decrease in multi-site SchoolAccess(R) customers.

Managed Broadband Services Segment Cost of Goods Sold Managed broadband services segment Cost of Goods Sold decreased \$316,000 to \$1.1 million from the second quarter of 2005 to the second quarter of 2006. The decrease is primarily due to a second quarter of 2005 analysis of circuit costs directly contributing to the Managed Broadband services segment revenues. The analysis resulted in a \$535,000 increase in Broadband services Cost of Goods Sold in the second quarter of 2005 that included first and second quarter 2005 circuit costs.

Selling, General and Administrative Expenses
Selling, general and administrative expenses increased 7.0% to \$40.7 million in
the second quarter of 2006 primarily due to a \$1.5 million increase in our
share-based compensation expense mostly resulting from our adoption of SFAS No.
123(R) on January 1, 2006. Following our adoption of SFAS No. 123(R) we
recognized \$1.6 million in share-based compensation expense which was allocated
to our reportable segments as follows (amounts in thousands):

<TABLE>
<CAPTION>

Reportable	Seaments
Keborcanie	Degments

			Network		Managed	
		Consumer	Access	Commercial	Broadband	Total
<\$>	 <c></c>		<c></c>	<c></c>	<c></c>	<c></c>
Share-based compensation expense	\$	508	605	348	121	1,582
	==					

#### </TABLE>

As a percentage of total revenues, selling, general and administrative expenses were 34.4% in the second quarter of 2006 and 2005.

42

#### Bad Debt Expense

Bad debt expense increased \$1.1 million to a net expense of \$1.3 million in the second quarter of 2006 as compared to the second quarter of 2005. The increase is primarily due to the following:

- o A \$996,000 decrease in the realization of a recovery from MCI, Inc. (merged with Verizon Communications Inc. ("Verizon") in January 2006) through a reduction to bad debt expense in the second quarter of 2006 as compared to the second quarter of 2005, and
- o A \$479,000 increase in the allowance for our long-distance, local service and Internet invoices generated by the unified order management

and fulfillment, billing, customer service, cash application, and credit and collection system to which we converted on September 1, 2005. The allowance increase is due to an increase in the age of certain invoices as we resolve billing issues.

Depreciation and Amortization Expense

Depreciation and amortization expense increased 10.0% to \$20.2 million in the second quarter of 2006. The increase is primarily due to our \$95.3 million investment in equipment and facilities placed into service during 2005 for which a full year of depreciation will be recorded in 2006 and the \$21.4 million investment in equipment and facilities placed into service during the second quarter of 2006 for which a partial year of depreciation will be recorded in 2006.

Income Tax Expense

Income tax expense totaled \$3.9 million in the second quarter of 2006 and \$4.0 million in the second quarter of 2005. Our effective income tax rate decreased from 43.3% in the second quarter of 2005 to 41.6% in the second quarter of 2006.

At June 30, 2006, we have (1) tax net operating loss carryforwards of approximately \$153.8 million that will begin expiring in 2010 if not utilized, and (2) alternative minimum tax credit carryforwards of approximately \$2.2 million available to offset regular income taxes payable in future years. We estimate that we will utilize net operating loss carryforwards of \$22.0 million to \$27.0 million during the year ended December 31, 2006. Our utilization of certain net operating loss carryforwards is subject to limitations pursuant to Internal Revenue Code section 382.

Tax benefits associated with recorded deferred tax assets are considered to be more likely than not realizable through future reversals of existing taxable temporary differences and future taxable income exclusive of reversing temporary differences and carryforwards. The amount of deferred tax asset considered realizable, however, could be reduced in the near term if estimates of future taxable income during the carryforward period are reduced which would result in additional income tax expense. We estimate that our effective annual income tax rate for financial statement purposes will be 43% to 45% in the year ended December 31, 2006.

Six Months Ended June 30, 2006 ("2006") Compared To Six Months Ended June 30, 2005 ("2005")

Overview of Revenues and Cost of Goods Sold Total revenues increased 6.4% from \$217.2 million in 2005 to \$231.0 million in 2006. Revenue increases in our Consumer, Network Access and Commercial segments were partially off-set by decreased revenue in our Managed Broadband segment. See the discussion below for more information by segment.

Total Cost of Goods Sold increased 5.1% from \$71.2 million in 2005 to \$74.8

43

million in 2006. Increases in the Consumer and Network Access segments' Cost of Goods Sold were partially off-set by decreased Cost of Goods Sold in our Commercial and Managed Broadband segments. See the discussion below for more information by segment.

Consumer Services Segment Overview

Consumer services segment revenue in 2006 represented 37.6% of consolidated revenues. The components of Consumer services segment revenue are as follow (amounts in thousands):

<TABLE> <CAPTION>

		2006	2005	Percentage Change
<\$>	<c></c>		<c></c>	<c></c>
Voice	\$	22,763	23,589	(3.5%)
Video		44,331	42,136	5.2%
Data		14,219	12,566	13.2%
Wireless		5 <b>,</b> 573	2 <b>,</b> 250	147.7%
Total Consumer services segment revenue	\$ ==	86,886 ======	80,541 =======	7.9%

</TABLE>

Selected key performance indicators for our Consumer services segment follow: <TABLE>

2006	2005	Percentage Change
<c></c>	<c></c>	<c></c>

Long-distance minutes carried (in millions) 72.8 81.3 (10.5%)

Video:

Average monthly gross revenue per subscriber(1) \$60.55 \$59.24 2.2%

_____

(1) Year-to-date average monthly consumer video revenues divided by the average of consumer video basic subscribers at the beginning and ending of the period.

-----

 $</\,\mathrm{FN}>$ 

<FN>

</TABLE>

Please refer to our three-month results of operations discussion for our June 30, 2006 and 2005 selected key performance indicators.

Consumer Services Segment Revenues

The 3.5% decrease in voice revenue is primarily due to the decrease in long-distance minutes carried for these customers and a \$502,000 decrease in support from the Universal Service Program in 2006 as compared to 2005. The decrease is partially off-set by an approximately \$300,000 increase in local service revenue in 2006 as compared to 2005 due to the implementation of the monthly network access fee in April 2005.

The 5.2% increase in video revenue is primarily due to an increase in digital programming tier subscribers in 2006, the rate increases previously described, and a 22.1% increase in equipment rental revenue to \$6.4 million in 2006. The increase in equipment rental revenue is primarily caused by the increased use of digital distribution technology.

The 13.2% increase in data revenue is primarily due to a 13.4% increase in cable modem revenue to \$12.0 million in 2006 as compared to 2005. The increase in cable modem revenue is primarily due to the increase in subscribers.

44

The 147.7% increase in wireless revenue is primarily due to the increase in wireless subscribers.

Consumer Services Segment Cost of Goods Sold

Consumer services segment Cost of Goods Sold increased 7.8% to \$33.0 million from 2005 to 2006 primarily due to increased video and wireless Cost of Goods Sold resulting from increased revenue. The increase in Cost of Goods Sold is partially off-set by decreased voice Cost of Goods Sold primarily due to cost savings resulting from the increased deployment of DLPS lines in the last six months of 2005 and first six months of 2006.

Network Access Services Segment Overview

Network access services segment revenue in 2006 represented 34.3% of consolidated revenues. The components of Network Access services segment revenue are as follows (amounts in thousands): <TABLE>

<CAPTION>

Percentage 2006 2005 Change <S> <C> <C> <C> \$ 52,328 26,872 44,909 Voice 16.5% 26,142 Data 2.8% 71,051 \$ 79,200 11.5% Total Network Access services segment revenue

</TABLE>

Selected key performance indicators for our Network Access services segment

follow: <TABLE>

<CAPTION>

	2006	2005	Percentage Change
<\$>	<c></c>	<c></c>	<c></c>
Voice:			
Long-distance minutes carried (in millions)	619.5	497.9	24.4%

  |  |  |Network Access Services Segment Revenues

The 16.5% increase in voice revenue is primarily due to the increase in minutes carried for our other common carrier customers partially off-set by a 4.3% decrease in our rate per minute on minutes carried for other common carriers. The average rate per minute decrease is primarily due to the annual 3.0% rate decrease mandated by the Consolidated Appropriations Act for Fiscal Year 2005 effective January 2005 which will result in rate decreases of 3.0% per year through 2010 and a change in the composition of traffic carried for other common

carriers.

Network Access Services Segment Cost of Goods Sold Network Access services segment Cost of Goods Sold increased 12.8% to \$17.6 million from 2005 to 2006 primarily due to increased voice minutes carried.

45

Commercial Services Segment Overview

Commercial services segment revenue in 2006 represented 22.6% of consolidated revenues. The components of Commercial services segment revenue are as follows (amounts in thousands):

<TABLE>

<CAPTION>

		2006	2005	Percentage Change
<\$>	<c></c>		<c></c>	<c></c>
Voice	\$	16,120	17,577	(8.3%)
Video		3,659	3 <b>,</b> 533	3.6%
Data		31,310	30,214	3.6%
Wireless		1,052	442	138.0%
Total Commercial services segment revenue	\$	52,141	51,766	0.7%

-----

</TABLE>

Selected key performance indicators for our Commercial services segment follow:

<TABLE>

<CAPTION>

	2006	2005	Percentage Change
<\$>	<c></c>	<c></c>	<c></c>
Voice:			
Long-distance minutes carried (in millions)	69.5	71.0	(2.1%)

  |  |  |Please refer to our three-month results of operations discussion for our June 30, 2006 and 2005 selected key performance indicators.

Commercial Services Segment Revenues

The 8.3% decrease in voice revenue is primarily due to the decrease in minutes carried for our Commercial customers.

The 3.6% increase in data revenue is primarily due to the following:

- O A \$682,000 increase in managed services revenue primarily due to a \$988,000 or 14.8% increase in special project revenues partially off-set by a \$306,000 or 4.3% decrease in revenue earned from the lease and provision of management and maintenance services on a portion of our 800-mile fiber optic system capacity that extends from Prudhoe Bay to Valdez via Fairbanks as described above, and
- o A 7.8% increase to \$7.1 million in private line and private network services due to an increase in the number of circuits sold.

The 138.0% increase in wireless revenue is primarily due to the increase in wireless subscribers.

Commercial Services Segment Cost of Goods Sold Commercial services segment Cost of Goods Sold decreased 2.3% to \$22.0 million from 2005 to 2006 primarily due to decreased voice minutes carried and cost savings resulting from the increased deployment of DLPS lines in the last six months of 2005 and first six months of 2006. The overall Cost of Goods Sold decrease is partially off-set by increased managed services Cost of Goods Sold primarily due to increased managed services revenue in 2006 as compared to 2005.

46

Managed Broadband Services Segment Overview
Managed broadband services segment revenue in 2006 represented 5.5% of
consolidated revenues. Managed broadband services segment revenue, which
includes data products only, decreased 7.3% to \$12.8 million in 2006 as compared
to 2005.

Please refer to our three-month results of operations discussion for our June 30, 2006 and 2005 selected key performance indicators.

Managed Broadband Services Segment Revenues

The decrease in Managed Broadband services segment revenue is primarily due to a decrease in our multi-site SchoolAccess(R) customers in 2006 as compared to 2005 and a decrease in the rate charged for certain circuits purchased by our rural health customers. An increase in single-site SchoolAccess(R) customers from which we generate less revenue was partially off-set by the decrease in multi-site SchoolAccess(R) customers.

Managed Broadband Services Segment Cost of Goods Sold
Managed broadband services segment Cost of Goods Sold decreased \$332,000 to \$2.1
million from 2005 to 2006.

Selling, General and Administrative Expenses Selling, general and administrative expenses increased 6.8% to \$80.3 million in 2006 primarily due to the following:

A \$2.2 million increase in our share-based compensation expense primarily due to our adoption of SFAS No. 123(R) on January 1, 2006. Following our adoption of SFAS No. 123(R) we recognized \$2.3 million in share-based compensation expense which was allocated to our reportable segments as follows (amounts in thousands):

<TABLE> <CAPTION>

#### Reportable Segments

	Network			Managed		
		Consumer	Access	Commercial	Broadband	Total
<s></s>	<c:< td=""><td>&gt;</td><td><c></c></td><td><c></c></td><td><c></c></td><td><c></c></td></c:<>	>	<c></c>	<c></c>	<c></c>	<c></c>
Share-based compensation expense	\$	770	811	509	170	2,260
	==		=========	==========	=========	========

# </TABLE>

A \$1.8 million increase in health insurance costs primarily resulting from a decrease in our reserve for incurred but not reported health insurance claims in 2005 in order to reflect historical experience that was not repeated in 2006 and an increase in our medical claims in 2006.

The selling, general and administrative expenses increase is partially off-set by a \$677,000 decrease in our company-wide success sharing bonus accrual in 2006.

As a percentage of total revenues, selling, general and administrative expenses increased to 34.7% in 2006 from 34.6% in 2005, primarily due to an increase in selling, general and administrative expenses without a proportional increase in revenues.

47

Bad Debt Expense (Recovery)
Bad debt expense (recovery) increased \$2.0 million to a net expense of \$1.8 million in 2006 primarily due to the following:

- o A \$1.5 million decrease in the realization of a recovery from Verizon through a reduction to bad debt expense in 2006 as compared to 2005,
- A \$859,000 increase in the allowance for our long-distance, local service and Internet invoices generated by the unified order management and fulfillment, billing, customer service, cash application, and credit and collection system to which we converted on September 1, 2005. The allowance increase is due to an increase in the age of certain invoices as we resolve billing issues.

# Depreciation and Amortization Expense

Depreciation and amortization expense increased 11.9% to \$40.3 million in 2006. The increase is primarily due to our \$95.3 million investment in equipment and facilities placed into service during 2005 for which a full year of depreciation will be recorded in 2006 and the \$21.4 million investment in equipment and facilities placed into service during 2006 for which a partial year of depreciation will be recorded in 2006.

#### Income Tax Expense

Income tax expense totaled \$7.5 million in 2006 and \$7.5 million in 2005. Our effective income tax rate increased from 43.0% in 2005 to 44.7% in 2006 due primarily to adjustments to deferred tax assets and liabilities balances in 2006.

Cumulative Effect of a Change in Accounting Principle On January 1, 2006 we adopted SFAS No. 123(R), "Share-Based Payment." SFAS 123(R) requires us to measure share-based compensation liability instruments at fair value as of January 1, 2006. Previously, we measured those liability instruments at their intrinsic value determined as of their grant date. The transition impact (expense) of adopting SFAS No. 123(R) attributed to measuring such liability instruments at fair value totaled approximately \$1.1 million, net of income tax benefit of \$469,000 and is reported as a component of the cumulative effect of change in accounting principle in the accompanying June 30, 2006 Consolidated Statement of Operations.

Additionally, SFAS 123(R) requires us to estimate pre-vesting option forfeitures at the time of grant and periodically revise those estimates in subsequent periods if actual forfeitures differ from those estimates. We record share-based compensation expense only for those awards expected to vest using an estimated forfeiture rate based on our historical pre-vesting forfeiture data. Previously, we accounted for forfeitures as they occurred under the pro forma disclosure

provisions of SFAS 123 for periods prior to 2006. The transition impact (benefit) of adopting SFAS No. 123(R) attributed to accruing for expected forfeitures on outstanding share-based awards totaled \$108,000, net of income tax expense of \$44,000 and is reported as a component of the cumulative effect of change in accounting principle in the accompanying June 30, 2006 Consolidated Statement of Operations.

48

Multiple System Operator ("MSO") Operating Statistics Our operating statistics include capital expenditures and customer information from our Consumer and Commercial services segments which offer services utilizing our cable services' facilities.

Our capital expenditures by standard reporting category for the six months ended June 30, 2006 and 2005 follows (amounts in thousands):

		2006	2005
Customer premise equipment Line extensions Upgrade/rebuild Support capital Scalable infrastructure Commercial	\$	7,799 4,677 2,561 610 386 17	7,138 1,363 7,441 508 1,818
Sub-total Remaining reportable segments capital expenditures		16,050 19,169	18,437 29,253
	\$ ==	35 <b>,</b> 219	47,690 ========

The standardized definition of a customer relationship is the number of customers that receive at least one level of service utilizing our cable services segment's facilities, encompassing voice, video, and data services, without regard to which services customers purchase. At June 30, 2006 and 2005 we had 123,800 and 125,400 customer relationships, respectively.

The standardized definition of a revenue generating unit is the sum of all primary analog video, digital video, high-speed data, and telephony customers, not counting additional outlets. At June 30, 2006 and 2005 we had 247,800 and 220,500 revenue generating units, respectively.

Liquidity and Capital Resources

Cash flows from operating activities totaled \$53.7 million for the six months ended June 30, 2006 as compared to \$53.4 million for the six months ended June 30, 2005.

Other sources of cash during the six months ended June 30, 2006 included \$7.7 million from the issuance of our Class A common stock. Other uses of cash during the six months ended June 30, 2006 included expenditures of \$35.2 million for property and equipment, including construction in progress, and the purchase of \$21.9 million of common stock to be retired.

Working capital totaled \$89.1 million at June 30, 2006, a \$12.5 million increase as compared to \$76.5 million at December 31, 2005. The increase is primarily due to a \$2.6 million reclassification of a long-term deposit to current assets, a \$2.5 million reclassification of the current portion of Notes Receivable from Related Parties from non-current assets at June 30, 2006 as compared to December 31, 2005 and a \$1.3 million increase in Cash and Cash Equivalents resulting from a net positive cash flow during the six months ended June 30, 2006.

Net receivables increased \$2.2 million from December 31, 2005 to June 30, 2006 primarily due to the timing of payments on trade receivables from several large customers.

4 9

Senior Notes

We were in compliance with all Senior Notes loan covenants at June 30, 2006.

Senior Credit Facility

We were in compliance with all Senior Credit Facility loan covenants at June 30, 2006.

Capital Lease Obligation

On March 31, 2006, through our subsidiary GCC we entered into an agreement to lease transponder capacity on Intelsat's Galaxy 18 spacecraft that is expected to be launched during 2007. We will also lease capacity on the Horizons 1 satellite, which is owned jointly by Intelsat and JSAT International, Inc. The leased capacity is expected to replace our existing transponder capacity on Intelsat's Galaxy 10R satellite when it reaches its end of life.

We will lease, subject to a termination option, C-band and Ku-Band transponders over an expected term of approximately 14 years once the satellite is placed

into commercial operation in its assigned orbital location, and the transponders meet specific performance specifications and are made available for our use. The present value of the lease payments, excluding telemetry, tracking and command services and back-up protection, is expected to total \$77.0 million to \$82.0 million. We will record the capital lease obligation and the addition to our Property and Equipment when the satellite is made available for our use which is expected to occur approximately one month after the expected July 2007 launch.

A summary of estimated future minimum lease payments for this lease follows (amounts in thousands):

Years ending December 31:		
2006	\$	
2007		4,584
2008		9,168
2009		9,168
2010		9,168
2011 and thereafter	9	6,264
Total minimum lease payments	\$ 12	8,352
	====	

Upon payment of a monthly fee, we have the option to terminate the lease of the C-band transponders through June 1, 2007. We may forfeit our termination option at which time we would no longer be obligated to continue paying the monthly fee. If we elect to terminate our C-band transponder lease we must return the transponders and pay a termination fee.

### Capital Expenditures

Our expenditures for property and equipment, including construction in progress, totaled \$35.2 million and \$47.7 million during the six months ended June 30, 2006 and 2005, respectively. Our capital expenditures requirements in excess of approximately \$25.0 million per year are largely success driven and are a result of the progress we are making in the marketplace. We expect our 2006 expenditures for property and equipment for our core operations, including construction in progress, to total \$80.0 million to \$90.0 million, depending on available opportunities and the amount of cash flow we generate during 2006.

### Share Repurchases

GCI's Board of Directors has authorized a common stock buyback program for the repurchase of our Class A and Class B common stock in order to reduce our

50

outstanding shares of Class A and Class B common stock. Our Board of Directors authorized us and we obtained permission from our lenders for up to \$50.0 million of repurchases through June 30, 2006. We are authorized to continue our stock repurchases of up to \$5.0 million per quarter indefinitely and to use stock option exercise proceeds to repurchase additional shares. If stock repurchases are less than the total approved quarterly amount the difference may be carried forward and applied against future stock repurchases. During the six months ended June 30, 2006 we repurchased 1,857,442 shares of our common stock at a cost of approximately \$21.9 million. We expect to continue the repurchases for an indefinite period subject to the availability of free cash flow, availability under our credit facilities, and the price of our Class A and Class B common stock. The repurchases have and will continue to comply with the restrictions of SEC Rule 10b-18.

## Other Expenditures

We agreed to invest approximately \$29.5 million in exchange for a majority equity interest in Alaska DigiTel, a small Alaska PCS provider. The existing owners will retain a minority ownership interest and voting control of Alaska DigiTel. The exact percentage and dollar amounts for our interest in Alaska DigiTel will vary in proportion to the amount the existing owners elect to retain, but we expect to own between 75% and 85% after completion of the transaction. The transaction is based on a post closing enterprise valuation of \$37.0 million for Alaska DigiTel. We will fund the transaction from cash on hand, by drawing down additional debt, or a combination of the two. In June 2006 we entered into a Reorganization Agreement with the members of Alaska DigiTel and certain other parties setting forth the formal terms and conditions the principal terms of which are the same as those in the superseded memorandum of understanding. The Alaska DigiTel Applicants jointly filed applications with the FCC seeking the requisite regulatory consent to the parties' transaction. Matanuska Telephone Association filed a Petition to Deny the applications with the FCC in February 2006. ACS Wireless, Inc. filed an objection to the applications with the FCC in July 2006. The Applicants are opposing these adverse filings. It is uncertain when the FCC will act upon the contested applications. We are uncertain when this transaction will close.

We have provided a \$3.0 million bank depository account as collateral for an Alaska DigiTel term loan. The amount is classified as Cash and Cash Equivalents on our June 30, 2006 Consolidated Balance Sheet.

On July 31, 2006, through our subsidiary GCC we entered into an agreement to purchase an IRU in the Kodiak-Kenai Cable Company, LLC's marine-based fiber optic cable system linking Anchorage to Kenai, Homer, Kodiak and Seward, Alaska.

The new system is under construction and we anticipate it will be placed into service in the first quarter of 2007. We have committed to purchase a minimum of \$8.0 million to \$8.5 million in IRU capacity in three installments through 2011.

The long-distance, local access, cable, Internet and wireless services industries continue to experience substantial competition, regulatory uncertainty, and continuing technological changes. Our future results of operations will be affected by our ability to react to changes in the competitive and regulatory environment and by our ability to fund and implement new or enhanced technologies. We are unable to determine how competition, economic conditions, and regulatory and technological changes will affect our ability to obtain financing under acceptable terms and conditions.

We believe that we will be able to meet our liquidity and capital requirements, and fixed charges during the upcoming year through our cash flows from operating activities, existing cash, cash equivalents, short-term investments, credit facilities, and other external financing and equity sources. Should cash flows

51

be insufficient to support additional borrowings and principal payments scheduled under our existing credit facilities, capital expenditures will likely be reduced.

Schedule of Certain Known Contractual Obligations

The following table details future projected payments associated with certain known contractual obligations as of December 31, 2005, the date of our most recent fiscal year-end balance sheet. Our schedule of certain known contractual obligations has been updated to reflect our transponder capacity capital lease obligation discussed above.

<TABLE> <CAPTION>

Payments Due by Period

		Total	Less than 1 Year	1 to 3 Years	4 to 5 Years	More Than 5 Years
			(Amour	nts in thous	ands)	
<\$>	<c:< td=""><td>&gt;</td><td><c></c></td><td><c></c></td><td><c></c></td><td><c></c></td></c:<>	>	<c></c>	<c></c>	<c></c>	<c></c>
Long-term debt	\$	479,550	1,725	3,425	3,200	471,200
Interest on long-term debt		197,200	23,200	46,400	46,400	81,200
Capital lease obligations, including						
interest		130,594	252	14,268	18,852	97 <b>,</b> 222
Operating lease commitments		79,532	19,359	24,254	15,875	20,044
Purchase obligations		26,322	15,753	10,569		
Other		29,500	29,500			
Total contractual obligations	\$	942,698	89,789	98,916	84,327	669,666

### </TABLE>

For long-term debt included in the above table, we have included principal payments on our Senior Credit Facility and on our Senior Notes. Interest on amounts outstanding under our Senior Credit Facility is based on variable rates and therefore the amount is not determinable. Our Senior Notes require semi-annual interest payments of \$11.6 million through February 2014. For a discussion of our Senior Notes and Senior Credit Facility see note 7 in the "Notes to Consolidated Financial Statements" included in Part II of our December 31, 2005 annual report on Form 10-K.

For a discussion of our capital and operating leases, see note 16 in the "Notes to Consolidated Financial Statements" included in Part II of our December 31, 2005 annual report on Form 10-K and note 6 in the accompanying "Notes to Interim Condensed Consolidated Financial Statements."

Purchase obligations include a remaining commitment to purchase a certain number of outdoor, network powered multi-media adapters, indoor multi-media adapters, cable modems, and cable modem termination systems of \$6.9 million and a remaining \$10.8 million commitment for our Alaska Airlines agreement as further described in note 16 in the "Notes to Consolidated Financial Statements" included in Part II of our December 31, 2005 annual report on Form 10-K. The contracts associated with these commitments are non-cancelable. Purchase obligations also include open purchase orders for goods and services for capital projects and normal operations totaling \$8.6 million which are not included in our Consolidated Balance Sheets at December 31, 2005, because the goods had not been received or the services had not been performed at December 31, 2005. The open purchase orders are cancelable.

Other consists of our commitment to acquire a substantial equity interest in Alaska DigiTel for approximately \$29.5 million as further described above.

52

New Accounting Standards

For a discussion of our new accounting standards see note 1(j) in the accompanying "Notes to Interim Condensed Consolidated Financial Statements"

### Critical Accounting Policies

Our accounting and reporting policies comply with accounting principles generally accepted in the United States of America. The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions. The financial position and results of operations can be affected by these estimates and assumptions, which are integral to understanding reported results. Critical accounting policies are those policies that management believes are the most important to the portrayal of our financial condition and results, and require management to make estimates that are difficult, subjective or complex. Most accounting policies are not considered by management to be critical accounting policies. Several factors are considered in determining whether or not a policy is critical in the preparation of financial statements. These factors include, among other things, whether the estimates are significant to the financial statements, the nature of the estimates, the ability to readily validate the estimates with other information including third parties or available prices, and sensitivity of the estimates to changes in economic conditions and whether alternative accounting methods may be utilized under accounting principles generally accepted in the United States of America. For all of these policies, management cautions that future events rarely develop exactly as forecast, and the best estimates routinely require adjustment. Management has discussed the development and the selection of critical accounting policies with our Audit Committee.

Those policies considered to be critical accounting policies for the six months ended June 30, 2006 are described below.

- We maintain allowances for doubtful accounts for estimated losses resulting from the inability of our customers to make required payments. We also maintain an allowance for doubtful accounts based on our assessment of the likelihood that our customers will satisfactorily comply with rules necessary to obtain supplemental funding from the Universal Service Administrative Company ("USAC") for services provided by us under our packaged communications offerings to rural hospitals, health clinics and school districts. We base our estimates on the aging of our accounts receivable balances, financial health of specific customers, regional economic data, changes in our collections process, regulatory requirements, and our customers' compliance with USAC rules. If the financial condition of our customers were to deteriorate or if they are unable to emerge from reorganization proceedings, resulting in an impairment of their ability to make payments, additional allowances may be required. If their financial condition improves or they emerge successfully from reorganization proceedings, allowances may be reduced. Such allowance changes could have a material effect on our consolidated financial condition and results of operations.
- O We record all assets and liabilities acquired in purchase acquisitions, including goodwill and other intangibles, at fair value as required by SFAS No. 141, "Business Combinations." Goodwill and indefinite-lived assets such as our cable certificates are not amortized but are subject, at a minimum, to annual tests for impairment and quarterly evaluations of whether events and circumstances continue to support an indefinite useful life as required by SFAS No. 142, "Goodwill and Other Intangible Assets." Other intangible assets are amortized over their

53

estimated useful lives using the straight-line method, and are subject to impairment if events or circumstances indicate a possible inability to realize the carrying amount as required by SFAS No. 142. The initial goodwill and other intangibles recorded and subsequent impairment analysis requires management to make subjective judgments concerning estimates of the applicability of quoted market prices in active markets and, if quoted market prices are not available and/or are not applicable, how the acquired asset will perform in the future using a discounted cash flow analysis. Estimated cash flows may extend beyond ten years and, by their nature, are difficult to determine over an extended timeframe. Events and factors that may significantly affect the estimates include, among others, competitive forces, customer behaviors and attrition, changes in revenue growth trends, cost structures and technology, and changes in discount rates, performance compared to peers, material and ongoing negative economic trends, and specific industry or market sector conditions. In determining the reasonableness of cash flow estimates, we review historical performance of the underlying asset or similar assets in an effort to improve assumptions utilized in our estimates. In assessing the fair value of goodwill and other intangibles, we may consider other information to validate the reasonableness of our valuations including third-party assessments. These evaluations could result in a change in useful lives in future periods and could result in write-down of the value of intangible assets. Our cable certificate and goodwill assets are our only indefinite-lived intangible assets and because of the significance of our cable certificate and goodwill assets to our consolidated balance sheet, our annual and quarterly impairment

analyses and quarterly evaluations of remaining useful lives are critical. Any changes in key assumptions about the business and its prospects, changes in market conditions or other externalities, or recognition of previously unrecognized intangible assets for impairment testing purposes could result in an impairment charge and such a charge could have a material adverse effect on our consolidated results of operations.

- We estimate unbilled long-distance services Cost of Goods Sold based upon minutes of use carried through our network and established rates. We estimate unbilled costs for new circuits and services, and network changes that result in traffic routing changes or a change in carriers. Carriers that provide service to us regularly make network changes that can lead to new, revised or corrected billings. Such estimates are revised or removed when subsequent billings are received, payments are made, billing matters are researched and resolved, tariffed billing periods lapse, or when disputed charges are resolved. Revisions to previous estimates could either increase or decrease costs in the year in which the estimate is revised which could have a material effect on our consolidated financial condition and results of operations.
- Our income tax policy provides for deferred income taxes to show the effect of temporary differences between the recognition of revenue and expenses for financial and income tax reporting purposes and between the tax basis of assets and liabilities and their reported amounts in the financial statements in accordance with SFAS No. 109, "Accounting for Income Taxes." We have recorded deferred tax assets of approximately \$62.9 million associated with income tax net operating losses that were generated from 1992 to 2003, and that expire from 2010to 2025. Pre-acquisition income tax net operating losses associated with acquired companies are subject to additional deductibility limits. We have recorded deferred tax assets of approximately \$2.2 million associated with alternative minimum tax credits that do not expire. Significant management judgment is required in developing our provision for income taxes, including the determination of deferred tax assets and liabilities and any valuation allowances that may be required against the deferred tax assets. In conjunction with certain 1996 acquisitions, we determined that approximately \$20.0 million of the

54

acquired net operating losses would not be utilized for income tax purposes, and elected with our December 31, 1996 income tax returns to forego utilization of such acquired losses. Deferred tax assets were not recorded associated with the foregone losses and, accordingly, no valuation allowance was provided. We have not recorded a valuation allowance on the deferred tax assets as of June 30, 2006 based on management's belief that future reversals of existing taxable temporary differences and estimated future taxable income exclusive of reversing temporary differences and carryforwards, will, more likely than not, be sufficient to realize the benefit of these assets over time. In the event that actual results differ from these estimates or if our historical trends change, we may be required to record a valuation allowance on deferred tax assets, which could have a material adverse effect on our consolidated financial position or results of operations.

Other significant accounting policies, not involving the same level of measurement uncertainties as those discussed above, are nevertheless important to an understanding of the financial statements. Policies related to revenue recognition, share-based payments, and financial instruments require difficult judgments on complex matters that are often subject to multiple sources of authoritative guidance. Certain of these matters are among topics currently under reexamination by accounting standards setters and regulators. No specific conclusions reached by these standard setters appear likely to cause a material change in our accounting policies, although outcomes cannot be predicted with confidence. A complete discussion of our significant accounting policies can be found in note 1 in "Notes to Consolidated Financial Statements" included in Part II of our December 31, 2005 annual report on Form 10-K.

Geographic Concentration and the Alaska Economy

We have one major customer, Verizon. Our remaining customers are located primarily throughout Alaska. Because of this geographic concentration, our growth and operations depend upon economic conditions in Alaska. The economy of Alaska is dependent upon the natural resources industries, and in particular oil production, as well as tourism, government, and United States military spending. Any deterioration in these markets could have an adverse impact on us. All of the federal funding and the majority of investment revenues are dedicated for specific purposes, leaving oil revenues as the primary source of general operating revenues. Tourism, air cargo, and service sectors have helped offset the prevailing pattern of oil industry downsizing that has occurred during much of the last several years.

No assurance can be given that the driving forces in the Alaska economy, and in particular, oil production, will continue at appropriate levels to provide an environment for expanded economic activity.

No assurance can be given that oil companies doing business in Alaska will be successful in discovering new fields or further developing existing fields which are economic to develop and produce oil with access to the pipeline or other means of transport to market, even with a reduced level of royalties. We are not able to predict the effect of changes in the price and production volumes of North Slope oil on Alaska's economy or on us.

5

PART I. ITEM 3.

Quantitative and Qualitative Disclosures About Market Risk

We are exposed to various types of market risk in the normal course of business, including the impact of interest rate changes. We do not hold derivatives for trading purposes.

Our Senior Credit Facility carries interest rate risk. Amounts borrowed under this Agreement bear interest at LIBOR plus 1.50% or less depending upon our Total Leverage Ratio (as defined). Should the LIBOR rate change, our interest expense will increase or decrease accordingly. As of June 30, 2006, we have borrowed \$158.4 million subject to interest rate risk. On this amount, each 1% increase in the LIBOR interest rate would result in \$1,584,000 of additional gross interest cost on an annualized basis.

PART I. ITEM 4.

Controls and Procedures

### (a) Evaluation of Disclosure Controls and Procedures

As of the end of the period covered by this Quarterly Report on Form 10-Q, we carried out an evaluation of the effectiveness of the design and operation of our "disclosure controls and procedures" (as defined in the Securities Exchange Act of 1934 ("Exchange Act") Rules 13a-15(e)) under the supervision and with the participation of our management, including our Chief Executive Officer and our Chief Financial Officer. Based upon that evaluation, our Chief Executive Officer and our Chief Financial Officer concluded that our disclosure controls and procedures are effective.

Disclosure controls and procedures are controls and other procedures that are designed to ensure that information required to be disclosed in our reports filed or submitted under the Exchange Act is recorded, processed, summarized and reported, within the time periods specified in the SEC's rules and forms. Disclosure controls and procedures include, without limitation, controls and procedures designed to ensure that information required to be disclosed in our reports filed under the Exchange Act is accumulated and communicated to management to allow timely decisions regarding required disclosure.

### (b) Changes in Internal Controls

During the first quarter of 2006 we corrected certain configuration settings in our new customer billing system so that credits and other adjustments are properly recorded. We will continue to vigorously monitor the effectiveness of these processes, procedures and controls, and will make any further changes as management determines appropriate.

There were no changes in our internal control over financial reporting identified in connection with the evaluation of our controls performed during the quarter ended June 30, 2006 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

We may enhance, modify, and supplement internal controls and disclosure controls and procedures based on experience.

56

Internal controls are a system designed to provide reasonable assurance to the Company's management and board of directors regarding the preparation and fair presentation of its financial statements. All internal control systems, no matter how well designed, have inherent limitations. Therefore, even those systems determined to be effective can provide only reasonable assurance with respect to financial statement preparation and presentation. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

PART II. ITEM 1.

LEGAL PROCEEDINGS

Information regarding material pending legal proceedings to which we are a party is included in note 6 to the accompanying "Notes to Interim Condensed

Consolidated Financial Statements" and is incorporated herein by reference.

PART II.

ITEM 2.

UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

- (a) Not applicable.
- (b) Not applicable.

(c) The following table provides information about repurchases of shares of our Class A and Class B common stock during the quarter ended June 30, 2006:

<TABLE>

Issuer Purchases of Equity Securities

Period	(a) Total Number of Shares Purchased	Price Paid	Announced Plans	•
<pre><s> April 1, 2006 to</s></pre>	<c></c>	<c></c>	<c></c>	<c></c>
April 30, 2006	576,193 (3)	\$11.45	2,880,435	\$20,720,000
May 1, 2006 to May 31, 2006	172,300 (3)	\$11.63	3,052,735	\$18,715,000
June 1, 2006 to June 30, 2006	792,000 (3)	\$12.27	3,844,735	\$8,999,000
Total	1,540,493 ========			

<FN>

(1) The repurchase plan was publicly announced on November 3, 2004. Our plan does not have an expiration date, however transactions pursuant to the plan are subject to periodic approval by our Board of Directors. We expect to continue the repurchases throughout 2006 subject to the availability of

57

free cash flow, availability under our credit facilities, and the price of our Class A and Class B common stock. We do not intend to terminate this plan in 2006.

- (2) The total amount approved for repurchase was \$50.0 million through June 30, 2006 consisting of \$30.0 million through December 31, 2005 and \$20.0 million through June 30, 2006. If stock repurchases are less than the total approved quarterly amount the difference may be carried forward and applied against future stock repurchases, subject to Board of Directors approval.
- (3) Open-market purchases and private party transactions made under our publicly announced repurchase plan.

</FN>
</TABLE>

PART II. ITEM 4.

SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

- (a) Date of the meeting: June 26, 2006 Purpose of meeting: Annual shareholders meeting
- (b) Name of each director elected at the meeting and the name of each other director whose term of office as a director continued after the meeting:

<TABLE> <CAPTION>

Name	Votes for	Votes withheld
<s></s>	<c></c>	<c></c>
Stephen M. Brett	77,181,772	526,470
Ronald A. Duncan	73,387,700	4,320,542
Stephen R. Mooney	59,016,743	18,691,499
Scott M. Fisher	77,289,488	418,754

</TABLE>

Directors, in addition to those listed above, whose term of office as director continued after the meeting:

Jerry A. Edgerton William P. Glasgow James M. Schneider

(c) Not applicable

PART II. ITEM 6.

EXHIBITS

58

<TABLE> <CAPTION>

Exhibit No.	Description
 <s></s>	<c></c>
10.135	Tenth Amendment to Contract for Alaska Access Services between General Communication, Inc. and its wholly owned subsidiary GCI Communication Corp., and MCI Communications Services, Inc. d/b/a Verizon Business Services (successor-in-interest to MCI Network Services, Inc., which was formerly known as MCI WorldCom Network Services) *
10.136	Reorganization Agreement among General Communication, Inc., Alaska DigiTel, LLC, The Members of Alaska DigiTel, LLC, AKD Holdings, LLC and The Members of Denali PCS, LLC dated as of June 16, 2006 (Nonmaterial schedules and exhibits to the Reorginati Agreement have been omitted pursuant to Item 601b.2 of Regulation S-K. We agree to furnish supplementally to the Commission upon request a copy of any omitted schedul or exhibit.) *
31.1	Certification Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 by our President and Director
31.2	Certification Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 by our Senior Vice President, Chief Financial Officer, Secretary and Treasurer
32.1	Certification Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 by our President and Director
32.2	Certification Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 by our Senior Vice President, Chief Financial Officer, Secretary and Treasurer

-----

59

## SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Securities and Exchange Commission. Each omitted Confidential Portion is marked by four asterisks.

## GENERAL COMMUNICATION, INC.

<TABLE> <CAPTION>

Signature	Title	Date
<\$>	<c></c>	<c></c>
/s/ Ronald A. Duncan	President and Director (Principal Executive Officer)	August 8, 2006
Ronald A. Duncan		
/s/ John M. Lowber	Senior Vice President, Chief Financial Officer, Secretary and Treasurer	August 8, 2006
John M. Lowber	(Principal Financial Officer)	
/s/ Alfred J. Walker	Vice President, Chief Accounting Officer	August 8, 2006
Alfred J. Walker 		

 (Principal Accounting Officer) |  |**** CONFIDENTIAL PORTION has been omitted pursuant to a request for confidential treatment by the Company to, and the material has been separately filed with, the SEC. Each omitted Confidential Portion is marked by four asterisks.

### TENTH AMENDMENT TO CONTRACT FOR ALASKA ACCESS SERVICES

This TENTH AMENDMENT TO THE CONTRACT FOR ALASKA ACCESS SERVICES ("Tenth Amendment") is effective as of the 1st day of May, 2006 ("Effective Date"), by and between GENERAL COMMUNICATION, INC. and its wholly owned subsidiary, GCI COMMUNICATION CORP., both Alaska corporations (together, "GCI") with offices located at 2550 Denali Street, Suite 1000, Anchorage, Alaska 99503-2781 and MCI COMMUNICATIONS SERVICES, INC. d/b/a Verizon Business Services (successor-in-interest to MCI Network Services, Inc., which was formerly known as MCI WORLDCOM Network Services ("Verizon"), with offices located at 1133 19th Street, N.W., Washington, D.C. 20036 (GCI with Verizon, collectively, the "Parties," and individually, a "Party").

#### RECITALS

WHEREAS, GCI and Verizon entered into that certain Contract for Alaska Access Services dated January 1, 1993 ("Original Agreement"), as amended by (i) the First Amendment to Contract for Alaska Access Services dated as of March 1, 1996, (ii) the Second Amendment to the Contract for Alaska Access Services dated as of January 1, 1998, (iii) the Third Amendment to Contract for Alaska Access Services dated as of March 1, 1998, (iv) the Fourth Amendment to Contract for Alaska Access Services dated as of January 1, 1999, (v) the Fifth Amendment to Contract for Alaska Access Services dated as of August 7, 2000, (vi) the Sixth Amendment to Contract for Alaska Access Services dated as of February 14, 2001, (vii) the Seventh Amendment to Contract for Alaska Access Services dated as of March 8, 2001, (vii) the Eighth Amendment to the Contract for Alaska Access Services dated as of July 1, 2003 and (viii) the Ninth Amendment to the Contract for Alaska Access Services dated as of January 23, 2005 (collectively, "Agreement") which set forth the general terms and conditions under which GCI provides certain telecommunications services to Verizon; and,

WHEREAS, the Parties have agreed to change the amount of certain credits under the Agreement, as described below,

### AGREEMENT

NOW, THEREFORE, in consideration of the premises and the mutual agreements herein contained, and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties agree as follows:

1. Definitions of Terms. All capitalized terms used in the Tenth Amendment

## 1 **** CONFIDENTIAL TREATMENT

but not defined herein shall have the meanings given to such terms in the Agreement. Additionally, any reference to Verizon in this Tenth Amendment shall be a reference to MCI in the Original Agreement, as amended, and visa versa.

2. Rates and Charges. Section 2.B.9 of the Agreement shall be deleted in its entirety and the following inserted in its place:

2.B.(9) Verizon **** and **** Services. Verizon **** and **** Services shall be **** GCI FCC Tariff #1 and **** shall reflect the requested terrestrial or satellite ****. Each month GCI shall calculate the total **** Verizon **** and **** Service **** for all **** requirements of **** and below and **** will be calculated and applied as follows:

**** Verizon Northbound Traffic **** and shall be identified on such **** as ****; and

**** Verizon Southbound Traffic **** and shall be identified on such **** as ****.

Further, during ****, GCI shall calculate the total **** Verizon **** and **** Service **** for all **** requirements of **** or **** services and **** will be calculated and applied as follows:

**** Verizon Northbound Traffic **** and shall be identified on such **** as ****; and

**** Verizon Southbound Traffic **** and shall be identified on such **** as ****.

During **** and each ****, the **** shall be subject to **** based on

**** northbound and southbound **** Verizon transports on GCI's network **** (the "Total Annual Verizon ****"), as follows:

Should the Total Annual Verizon **** equal or exceed ****, then GCI shall provide Verizon with a ****; and

Should the Total Annual Verizon **** not exceed ****, then GCI shall provide Verizon ****.

In the event that any of the above **** cannot be fully used ****, the remaining amount of such **** as directed by Verizon or **** Verizon upon request.

# 2 **** CONFIDENTIAL TREATMENT

- 3. ****. Verizon acknowledges and agrees that the **** and the **** are both being **** by **** under this Tenth Amendment, as described above. In consideration for GCI's agreement hereunder to increase these ****, which is good and valuable consideration, the sufficiency of which Verizon acknowledges, Verizon agrees **** to replace **** during the term of the Agreement, as defined and as may be extended under Paragraph 3 thereof.
- 4. Effect of Amendment. All other terms and conditions of the Agreement not expressly modified by this Tenth Amendment shall remain in full force and effect. The Parties hereby affirm and agree such terms remain binding.
- 5. Further Assurances. The Parties shall cooperate in good faith, and enter into such other instruments and take such other actions, as may be necessary or desirable, to fully implement the intent of this Tenth Amendment.
- 6. Counterparts; Signatures. This Tenth Amendment may be executed in counterparts, each of which shall be deemed an original and both of which together shall constitute one and the same instrument. When signed by each Party's authorized representative, a copy or facsimile of this Tenth Amendment shall have the same force and effect as one bearing an original signature.

This Tenth Amendment, together with the Agreement, is the complete agreement of the Parties and supersedes all other prior contracts and representations concerning its subject matter. Any further amendments must be in writing and signed by both Parties.

IN WITNESS WHEREOF, the Parties hereto each acting with proper authority have executed this Tenth Amendment as of the Effective Date.

MCI COMMUNICATIONS SERVICES, INC.

By: /s/Peter H. Reynolds Printed Name: Peter H. Reynolds Title: Director

GCI COMMUNICATION CORP.

By: /s/Richard Westlund Printed Name: Richard Westlund

Title: Senior Vice President & General Manager, Carrier Services

**** CONFIDENTIAL PORTION has been omitted pursuant to a request for confidential treatment by the Company to, and the material has been separately filed with, the SEC. Each omitted Confidential Portion is marked by four asterisks.

### REORGANIZATION AGREEMENT

among

GENERAL COMMUNICATION, INC.,

ALASKA DIGITEL, LLC,

THE MEMBERS OF ALASKA DIGITEL, LLC,

AKD HOLDINGS, LLC

and

THE MEMBERS OF DENALI PCS, LLC

Dated as of June 16, 2006

<TABLE> <CAPTION>

TABLE OF CONTENTS

	TABLE OF CONTENTS	
<s></s>		<c></c>
Page		
 anamion 1	DEFINITIONS	1
SECTION I	DEFINITIONS.	1
SECTION 2	TERMS OF THE REORGANIZATION	
2.1.	Contribution to Parent; Conversion to Units	
2.3.	Contributions by GCI	
2.4.	Purchase and Redemption of Units	
2.5.	Management Agreement; Profits Interest	10
2.6.	Operating Agreement	
2.7.	Closing	10
anamion 2	DEDDECENMENTANG AND MADDANISTED OF AND DADDING AND MUC AND MUDDED	1.1
3.1.	REPRESENTATIONS AND WARRANTIES OF AKD, PARENT AND THE AKD MEMBERS	
3.2.	The Transaction Agreements	
3.3.	AKD Business Assets.	
3.4.	Financial Matters.	
3.5.	Operational Matters	
3.6.	Employee Matters	
3.7.	Environmental Matters	
3.8.	Communications Regulatory Matters	
3.9.	Transactional Matters	23
SECTION 4	REPRESENTATIONS AND WARRANTIES OF THE DENALI MEMBERS	23
4.1.	Organization and Standing; Capitalization	
4.2.	The Transaction Agreements	
4.3.	No Assets	25
4.4.	Financial Matters	
4.5.	Operational Matters	
4.6.	Employee Matters	
4.7. 4.8.	Environmental Matters  Communications Regulatory Matters	
4.9.	Transactional Matters	
1.9.	Transactional nattorio	
SECTION 5	REPRESENTATIONS AND WARRANTIES OF GCI	31
5.1.	Organization and Standing	
5.2.	Execution and Validity of Agreements	
5.3.	No Violation or Approval	
5.4. 5.5.	Brokers, Finders, etc	
5.5.	NO OTAL WALLANCIES	
SECTION 6	COVENANTS	
6.1.	Exclusivity; Acquisition Proposals	32
6.2.	Notices and Consents	32
6.3.	i Preparation for Closing	าา
6.4.	Notification of Certain Matters	
6.5.	Other Limitations on Conduct of AKD's and Properties' Businesses	
6.6.	Other Limitations on Conduct of Denali's Business	
6.7.	Access to Information	
6.8.	Cooperation	35
6.9.	Announcements	
6.10.	Use of Proceeds of Interim Capital Loans	36

	7 CLOSING DELIVERIES
7.1.	Of AKD, Parent and the AKD Members36
7.2.	Of the Denali Members
7.3.	Of GCI38
CECETON	8 CONDITIONS PRECEDENT
8.1.	Conditions to Obligations of AKD, Parent, the AKD Members and the Denali Members
8.2.	Conditions to Obligations of AKD, Parent, the AKD Members and the Denail Members
8.3.	Conditions to the Obligations of Each Party
0.3.	Conditions to the Obligations of Each Party40
CECTION	9 POST-CLOSING COVENANTS
9.1.	Further Assurances
9.2.	Announcements 41
۶.۷.	Announcements
SECTION	10 INDEMNIFICATION41
10.1.	Indemnification by AKD, Parent and the AKD Members41
10.2.	Indemnification by the Denali Members41
10.2.	Indemnification by the behalf members
10.3.	Survival; Time Limits for Indemnification
10.4.	Basket and Cap
10.5.	Defense of Claims
10.6.	Detense of Claims43
CECETON	11 TERMINATION
11.1.	Right to Terminate
11.2.	Effect of Termination
11.3.	Specific Performance
	12 MISCELLANEOUS
12.1.	Arbitration
12.2.	Governing Law
12.3.	Notices
12.4.	Entire Agreement, Assignability, Etc47
12.5.	Counterparts48
12.6.	Representations as to Knowledge48
12.7.	Headings, Terms48
12.8.	Amendments
12.9.	Waivers
	ii
12.10	
12.11	
12.12	Expenses
12.13	. Construction
12.14	. AKD Members' Agent and Denali Members' Agent
12.15	. Incorporation of Exhibits
12.16	No Tax Advice

	EXHIBITS	
A	- Management Agreement	
В	- Operating Agreement	
C	- Opinion Items for Opinion of The Bogatin Law Firm, PLC	
D	- Opinion Items for Opinion of Lukas, Nace, Gutierrez & Sachs,	
-	Chartered	
E	- Non-Competition Agreement	
_	Non competition rigidement	
	SCHEDULES	
	SCHEPOLLO	
1.1	- AKD Interest-Bearing Obligations	
2.3.2	- Examples of Closing Adjustments	
2.3.2		
	Enamples of foot offoling najabolicites	
4	- AKD Disclosure Schedule	
5	- Denali Disclosure Schedule	
7.1(n)	- Persons to Execute Non-Competition Agreements	
	iii THIS REORGANIZATION AGREEMENT (this "Agreement") is made as of	
June 16, 2006, among General Communication, Inc., an Alaska corporation ("GCI"), Alaska DigiTel, LLC, an Alaska limited liability company ("AKD"), PacifiCom Holdings, L.L.C., a Delaware limited liability company ("Pacificom"), Red River Wireless, LLC, a Delaware limited liability company ("Red River" and, together with Pacificom, individually and collectively the "Denali Members"), Graystone Holdings, LLC, an Alaska limited liability company ("Graystone" and, together with Pacificom and Red River, individually and collectively, the "AKD Members"), and AKD Holdings, LLC, a Delaware limited liability company ("Parent").

## Recitals

A. AKD operates a wireless telecommunications business within the State of Alaska.

B. Denali PCS, LLC, an Alaska limited liability company ("Denali"),

holds licenses for certain spectrum capacity.

C. GCI desires to purchase all of the outstanding Denali Interests (as defined below) from the Denali Members and contribute cash and the Denali Interests to AKD in exchange for AKD Common Units (as defined below) and to purchase additional AKD Common Units from AKD all in accordance with the terms and conditions of this Agreement.

### Agreement

Accordingly, in consideration of the mutual covenants in this Agreement, and for other good and valuable consideration, the receipt and sufficiency of which are acknowledged, the Parties agree as follows:

## SECTION 1 DEFINITIONS

As used in this Agreement, the following terms have the indicated meanings:

"Accountants" has the meaning given in Section 2.3.2.

"Acquisition Transaction" has the meaning given in Section 6.1.

"Adverse Consequences" means all actions, suits, proceedings, investigations, complaints, claims, demands, orders, decrees, rulings, injunctions, judgments, directives, notices of violation, Liabilities, liens, losses (including loss of value), damages, penalties, fines, settlements, costs, remediation costs, expenses and fees (including court costs and reasonable fees and expenses of counsel and other experts), plus interest at a rate equal to two percentage points above the prime rate quoted by AKD's principal lender from time to time accrued from the date any Adverse Consequence becomes a liability of or is otherwise recognized by the party suffering the Adverse Consequence as determined in accordance with GAAP. As used in this Agreement, Adverse Consequences are not limited to matters asserted by third parties, but include Adverse Consequences incurred or sustained by a Party hereto other than as a result of claims by third persons.

1

"Affiliate" means, as to any Person, another Person that controls, is controlled by or is under common control with any Person and, as to any natural Person, any relative by blood, marriage or adoption. For that purpose, "control" means the power, directly or indirectly, by stock ownership, contract, family relationship, employment, position or otherwise, to significantly influence the business decisions of another Person.

"Agreement" has the meaning given in the Preamble.

"AKD" has the meaning given in the Preamble.

"AKD Balance Sheet Date" has the meaning given in Section 3.4.1.

"AKD Business Assets" means all assets and properties of AKD, whether real or personal, tangible or intangible, including, without limitation, (a) the AKD Licenses, (b) all furniture, office equipment, other equipment, automobiles and other tangible personal property listed in Schedule 3.3.1, (c) all of AKD's rights under the AKD Contracts, (d) all inventory, fixed assets and leasehold improvements, (e) all notes and accounts receivable of AKD, (f) all customer deposits, advance payments, prepaid items and expenses, deferred charges, rights of offset and credits and claims for refund, (g) all claims, rights and causes in action against third parties and all rights to insurance proceeds relating to any damage, destruction or impairment of the AKD Business Assets, (h) all Intellectual Property of AKD, including without limitation, those items listed in Schedule 3.3.5, (i) all books of account and all customer and supplier lists and other records related to AKD's business, (j) all goodwill associated with the AKD Business Assets and (k) all of AKD's right, title and ownership interest in Properties.

"AKD Common Unit" means a Common Unit (as defined in the Operating Agreement) of membership interest in AKD.

"AKD Contracts" has the meaning given in Section 3.5.3.

"AKD Disclosure Schedule" means the schedule delivered by AKD to GCI and attached to this Agreement setting forth exceptions to the representations and warranties of AKD and the AKD Members in this Agreement.

"AKD Financial Statements" has the meaning given in Section 3.4.1.

"AKD Group" means AKD and its Subsidiaries.

"AKD Interest-Bearing Obligations" means the interest-bearing

obligations of the AKD Group, including, without limitation, those obligations which are listed on Schedule 1.1 and are reflected on the Closing Adjustment Schedule.

"AKD Licenses" has the meaning given in Section 3.8.1.

"AKD Members" has the meaning given in the Preamble.

"AKD Members' Agent" has the meaning given in Section 12.14.

2

"AKD Net Asset Value" has the meaning given in Section 2.3.1.

"AKD Net Working Capital" means the excess of the accounts receivable and other current assets over the accounts payable, accrued expenses, property taxes and other current liabilities of the AKD Group (other than the current portion of any AKD Interest Bearing Obligation), in each case, as reflected on the Closing Adjustment Schedule.

"AKD Plans" has the meaning given in Section 3.6.1.

"AKD Profits Interest Unit" means a Profits Interest Unit (as defined in the Operating Agreement) of membership interest in AKD.

"AKD Redemption Price" has the meaning given in Section 2.4.

"Business Day" means any day other than a Saturday, a Sunday or a day on which banking institutions in Anchorage, Alaska are required or authorized to be closed.

"Capitalized Refinancing Costs" means the fees, expenses and costs incurred by AKD to consummate one or more credit facilities the proceeds of which will be used to fully pay the obligations of AKD to CoBank, ACB, but only to the extent that such fees, expenses and costs are approved by GCI, which approval shall not be unreasonably withheld or delayed.

"Claim" has the meaning given in Section 10.6.

"Closing" has the meaning given in Section 2.7.

"Closing Adjustment Schedule" has the meaning given in Section 2.3.2.

"Closing Date" has the meaning given in Section 2.7.

"Code" means the United States Internal Revenue Code of 1986,

as amended.

"Communications Act" means the Communications Act of 1934, as

amended.

"Communications Laws" means the Communications Act and the rules, regulations, published policies, published decisions, published orders, published rulings, and published notices of the FCC promulgated thereunder.

"Contract" means any contract, agreement, deed, mortgage, lease (whether or not capitalized), license, instrument, commitment, sales order, purchase order, quotation, bid, undertaking, arrangement or understanding, whether written or oral.

"Denali" has the meaning given in Recital B.

"Denali Balance Sheet Date" has the meaning given in Section 4.4.1.  $\,$ 

"Denali Contracts" has the meaning given in Section 4.5.3.

3

"Denali Disclosure Schedule" means the schedule delivered by the Denali Members to GCI and attached to this Agreement setting forth exceptions to the representations and warranties of the Denali Members in this Agreement.

"Denali Financial Statements" has the meaning given in Section 4.4.1.

"Denali Interest" means a membership interest in Denali.

"Denali Licenses" has the meaning given in Section 4.8.1.

"Denali Members" has the meaning given in the Preamble.

"Disagreement" has the meaning given in Section 2.3.2.

"Disputing Party" has the meaning given in Section 12.1.

"Encumbrance" means any conditional sale agreement, default of title, easement, encroachment, encumbrance, hypothecation, infringement, lien, mortgage, pledge, reservation, restriction, security interest, title retention or other security arrangement, any right to possession or use, any option or other right to acquire title to or the right to possession or use, or any adverse right or interest, charge or claim of any nature whatsoever of, on or with respect to any asset, capital stock, property or property interest.

"Environmental Law" means all present and future statutes, ordinances, codes, common law principles, rules, regulations, orders, decrees, standards, procedures, permit or license requirements or other requirements of any governmental authority relating to land use, public health, safety, welfare or the environment, including, without limitation, the Resource Conservation and Recovery Act, as amended, and the Comprehensive Environmental Response, Compensation, and Liability Act, as amended.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended and the regulations and orders promulgated thereunder.

"FCC" means the Federal Communications Commission.

"FCC Rules" means the rules, regulations, policies, instructions and orders of the FCC.

"Final Order" means any action or decision of the FCC as to which (i) no request for a stay or similar request is pending, no stay is in effect, the action or decision has not been vacated, reversed, set aside, annulled or suspended and any deadline for filing such request that may be designated by statute or regulation has passed without the filing of any such request, (ii) no petition for rehearing or reconsideration or application for review is pending and the time for the filing of any such petition or application has passed, (iii) the FCC does not have the action or decision under reconsideration on its own motion and the time within which it may effect such reconsideration has passed and (iv) no appeal is pending including other administrative or judicial review, or in effect and any deadline for filing any such appeal that may be designated by statute or rule has passed.

4

"Fire Lake" means Fire Lake Partners, L.L.C., an Alaska limited liability company.

"GAAP" means accounting principles generally accepted in the United States, consistently applied.

"GCI" has the meaning given in the Preamble.

"GCI Indemnitee" has the meaning given in Section 10.1.

"Governmental Entity" means any United States or foreign governmental or regulatory agency, commission, court, body, entity or authority.

"Graystone" has the meaning given in the Preamble.

"Hazardous Substance" means any material, substance, chemical, element, compound, mixture, pollutant, contaminant or toxic or hazardous material, substance or waste that is designated, defined, listed, classified or regulated by any Environmental Law.

"Indemnifying Party" has the meaning given in Section 10.6(a).

"Indemnitee" has the meaning given in Section 10.6.

"Intellectual Property" means any right, license or other claim to or of ownership, authorship or invention, or the right to use, any trade mark, trade name, service mark, patent, copyright, work, Internet domain name, Internet website, know-how, trade secret, formula, pattern, compilation, method, technique, confidential information, client list, technical information, program, specification or plan, whether or not registered or filed with any governmental authority, and all filings, registrations or applications relating to any of the foregoing.

"Interests" has the meaning given in Section 5.5

"Interim Capital Loans" means any loans in the principal amount of up to \$3 million made by CoBank, ACB or its successors to AKD pursuant to that certain Promissory Note dated March 17, 2006.

"Knowledge" of a Party has the meaning given in Section 12.6.

"Leased Real Property" means real property leased by AKD pursuant to the Real Property Leases.

"Liability" means any liability, debt or obligation, whether known or unknown, absolute or contingent, arising under contract, in tort, by

statute or regulation or otherwise, accrued or unaccrued, liquidated or unliquidated and due or to become due, and whether for the payment of money, the provision of goods or services or the performance of any other obligation.

_

"Management Agreement" means the Management Agreement to be entered into at the Closing between AKD and Fire Lake in the form attached as Exhibit A.

"Non-Competition Agreements" means the Non-Competition Agreements to be entered into at the Closing between AKD and the persons specified in Schedule 7.1(n) in the form attached as Exhibit E.

"Operating Agreement" means the Second Amended and Restated Operating Agreement of AKD to be entered into at the Closing among GCI, Parent, Fire Lake and the AKD Members in the form attached as Exhibit B.

"Pacificom" has the meaning given in the Preamble.

"Parent" has the meaning given in the Preamble.

"Party" means, individually or collectively, AKD, any AKD Member, any Denali Member, Parent or GCI.

"Person" means any individual, corporation, partnership, trust, limited liability company, association, governmental authority or any other entity.

"Premises" means the real property and improvements located at 3127 Commercial Drive, Anchorage, Alaska 99501, which are owned by Properties and leased to AKD.

"Properties" means Pacificom Properties, LLC, an Alaska limited liability company.

"Real Property Leases" has the meaning given in Section 3.3.3.

"RCA" means the Regulatory Commission of Alaska.

"Red River" has the meaning given in the Preamble.

"Regulatory Consents" has the meaning given in Section 6.2.

"Sections" has the meaning given in Section 12.7.

"Subscriber Contract" means any Contract pursuant to which AKD provides wireless telephone services to any customer.

"Subsidiary" shall mean, with respect to any Person, any corporation, limited liability company, limited liability partnership or other limited or general partnership, trust, association or other business entity of which an aggregate of at least a majority of the outstanding stock or other interests entitled to vote in the election of the board of directors of such corporation, managers, trustees or other controlling persons, or an equivalent controlling interest therein, of such Person is, at the time, directly or indirectly, owned by such Person and/or one or more subsidiaries of such Person.

# 6 **** CONFIDENTIAL TREATMENT

"Tax" means any federal, state or local tax or any foreign tax, including, without limitation, any net income, gross income, profits, premium, estimated, excise, sales, value added, services, use, occupancy, gross receipts, franchise, license, ad valorem, severance, capital levy, production, stamp, transfer, withholding, employment, unemployment, social security (including FICA), payroll or property tax, customs duty, or any other governmental charge or assessment, together with any interest, addition to tax or penalty.

"Tax Return" means any return, declaration, report, claim for refund, or information return or statement relating to Taxes, including any schedule or attachment thereto, and including any amendment thereof.

"Third Party" means a Person who is not a Party or an Affiliate of any Party.

"Transaction Agreements" means this Agreement, the Operating Agreement, the Management Agreement, the Non-Competition Agreements and all other instruments and agreements executed and delivered pursuant to this Agreement.

# SECTION 2 TERMS OF THE REORGANIZATION

2.1. Purchase of Denali. At the Closing, GCI shall purchase, and

the Denali Members shall sell to GCI, all of the outstanding Denali Interests, for aggregate consideration of \$**** in cash or immediately available funds, by wire transfer to accounts designated by each Denali Member in written instructions delivered to GCI not less than five Business Days before the Closing Date. GCI shall pay each Denali Member an amount of cash equal to \$**** multiplied by the Denali Interest owned by such Denali Member (expressed as a percentage of all Denali Interests outstanding as of the Closing Date). Each Denali Member shall transfer, convey and deliver all of the Denali Interest owned by such Denali Member to GCI, with full warranty of title, free and clear of all Encumbrances.

2.2. Contribution to Parent; Conversion to Units. At or before the Closing, each AKD Member shall contribute ***** of such AKD Member's membership interest in AKD to Parent in exchange for a proportionate membership interest in Parent. Immediately thereafter, all of the issued and outstanding membership interests of AKD shall be converted and reclassified into an aggregate of **** AKD Common Units, which shall be allocated among Parent and the AKD Members as follows:

Member	AKD Common Units
Parent	**** Units
Pacificom	**** Units
Red River	**** Units
Graystone	**** Units

The capital accounts of Parent and each AKD Member will be adjusted pursuant to Treasury Regulation ss. 1.704-1(b)(2)(iv)(f)-(g). The Parties agree that such adjustments will result in aggregate capital account balances for Parent and the AKD Members totaling the AKD Net Asset Value (as determined in accordance with Section 2.3). Immediately prior to the Closing, AKD shall distribute to Parent

# 7 **** CONFIDENTIAL TREATMENT

and the AKD Members all cash and cash equivalents of AKD as of the Closing Date. Parent and the AKD Members shall be responsible for any taxes arising from such distributions.

2.3. Contributions by GCI. At the Closing, GCI shall contribute to AKD (a)  $\****  in cash or immediately available funds and (b) all of the Denali Interest purchased by GCI pursuant to Section 2.1, and AKD shall sell and issue to GCI a number of AKD Common Units equal to (x)  $\****  minus the AKD Net Asset Value (as determined in accordance with this Section 2.3) divided by (y)  $\**** . GCI shall be credited with a capital account balance in AKD of  $\**** , representing the amount of cash and the agreed value of the Denali Interest contributed by GCI.

2.3.1. AKD Net Asset Value. The "AKD Net Asset Value" shall be an amount equal to the sum of (i) \$****, plus (ii) the capitalized cost of capital improvements funded with the proceeds of Interim Capital Loans that have been authorized by GCI in accordance with Section 6.10 and capitalized on the books of AKD in accordance with GAAP, plus (iii) the capitalized cost of capital improvements approved by GCI that are funded with the proceeds of any loan other than the Interim Capital Loans and capitalized on the books of AKD in accordance with GAAP, plus (iv) the Capitalized Refinancing Costs; provided, that (a) if the aggregate amount of AKD Interest-Bearing Obligations as of the Closing Date (as determined in accordance with Section 2.3.2) is greater or less than \$****, then the AKD Net Asset Value shall be decreased by the amount of the excess or increased by the positive amount of the deficit, as applicable, and (b) if the amount of AKD Net Working Capital as of the Closing Date (as determined in accordance with Section 2.3.2) is greater or less than \$****, then the AKD Net Asset Value shall be decreased by the positive amount of the deficit or increased by the amount of the excess, as applicable.

## 2.3.2. Closing Adjustment Schedule.

(a) Not less than five Business Days before the Closing Date, the AKD Members shall deliver to GCI a schedule signed by AKD and the AKD Members (the "Closing Adjustment Schedule") setting forth in detail the AKD Interest-Bearing Obligations and the AKD Net Working Capital as of the Closing Date. The Closing Adjustment Schedule shall be prepared by AKD and the AKD Members in good faith and in a manner consistent with GAAP and the balance sheets included in the AKD Financial Statements. AKD and the AKD Members shall provide GCI with such other information concerning the Closing Adjustment Schedule as GCI may reasonably request. The Closing Adjustment Schedule will be prepared as of the Closing Date but will be consistent in form with the AKD Interest-Bearing Obligations and the AKD Net Working Capital as of May 31, 2006 set forth on Schedule 2.3.2 attached hereto and incorporated herein by this reference.

(b) The Closing Adjustment Schedule shall be final and binding upon the Parties hereto for all purposes, unless GCI shall notify the AKD Members' Agent in writing, not later than thirty (30) days from the Closing Date, of a disagreement with the AKD Interest-Bearing Obligations or AKD Net

Working Capital reflected in the Closing Adjustment Schedule, in which event the provisions of Section 2.3.2(c) below shall apply. Such notice of disagreement shall specify all items as to which there is disagreement, and an explanation of the basis for any disagreement. During the 30-day review period, GCI shall have full access to the AKD Group's books and records during normal business hours

## 8 **** CONFIDENTIAL TREATMENT

and upon reasonable notice, and to the employees, representatives and agents of AKD who prepared, or assisted in the preparation of, the Closing Adjustment Schedule. GCI's failure to timely notify AKD in writing of the existence of such a disagreement shall be deemed, for all purposes, GCI's acceptance of the Closing Adjustment Schedule.

(c) In the event and to the extent that GCI shall timely notify the AKD Members' Agent in writing, as provided in Section 2.3.2(b) above, of a disagreement with the Closing Adjustment Schedule (the "Disagreement"), GCI and the AKD Members' Agent shall attempt, in good faith, to resolve such Disagreement. In the event that the parties are unable to resolve such Disagreement within ten Business Days from the date of receipt by the AKD Members' Agent of notice from GCI of the Disagreement, GCI and the AKD Members' Agent shall jointly select one of the "Big Four" accounting firms, or any successors thereto, to resolve the Disagreement (the "Accountants"). Each of GCI and the AKD Members' Agent shall submit to the Accountants its proposal concerning what the Closing Adjustment Schedule should be, and the parties shall submit to the Accountants all relevant financial data, and the Disagreement shall be submitted for final and binding arbitration and resolution before representatives of the Accountants. In resolving the Disagreement, the Accountants shall only consider those items or amounts in the Closing Adjustment Schedule as to which GCI has disagreed. After completing their review of the Disagreement, the Accountants shall resolve each item in dispute and confirm their conclusion (and the resulting Closing Adjustment Schedule) in writing to the AKD Members' Agent and GCI, and the decision of the Accountants regarding such adjustment shall be final and binding upon the parties hereto for all purposes and enforceable in any court of competent jurisdiction. The fees and costs of the Accountants, if any, in connection with such arbitration shall be paid by the nonprevailing party (either the AKD Members or GCI), whose identity shall be determined by the Accountants.

(d) Upon determination of the final Closing Adjustment Schedule, appropriate adjustments shall be made to the AKD Net Asset Value, the number of AKD Common Units issued pursuant to Section 2.3 and the capital account balances of Parent and the AKD Members under Section 2.2 to offset any changes from the Closing Adjustment Schedule. Schedule 1.3 of the Operating Agreement will thereafter be amended to reflect the adjusted amount of AKD Common Units issued to GCI and Schedule 4.1 of the Operating Agreement will thereafter be amended to reflect the adjusted capital account balances of Parent and the AKD Members.

2.3.3. Examples. The examples set forth on Schedule 2.3.3 illustrate the impact of a post-Closing adjustment to the AKD Net Asset Value on the number of AKD Common Units issued to GCI pursuant to this Section 2.3.

2.4. Purchase and Redemption of Units. At the Closing, GCI shall purchase from AKD **** AKD Common Units, and AKD shall use the proceeds from the sale of such additional AKD Common Units to redeem a like number of AKD Common Units from Parent and the AKD Members in accordance with this Section 2.4. AKD shall first redeem the fractional AKD Common Units issued to the AKD Members pursuant to Section 2.2 and shall use the balance of such proceeds to redeem AKD Common Units from Parent. The price for the purchase and redemption of AKD Common Units pursuant to this Section 2.4 shall be an amount per AKD Common Unit

# 9 **** CONFIDENTIAL TREATMENT

equal to the AKD Net Asset Value (as adjusted pursuant to Section 2.3.2) divided by **** (the "AKD Redemption Price"), payable in cash or immediately available funds to an account designated by AKD (in the case of the purchase by GCI) and by Parent and the AKD Members (in the case of the redemption by AKD) not less than five Business Days before the Closing Date.

2.4.1. Post-Closing Adjustment. If the AKD Redemption Price is increased or decreased as a result of a post-Closing adjustment to the AKD Net Asset Value pursuant to Section 2.3, then the amount of cash paid by GCI to AKD and by AKD to Parent and the AKD Members pursuant to this Section 2.4 shall be proportionately adjusted with appropriate payments to be made by GCI or Parent and the AKD Members within five Business Days after the Closing Adjustment Schedule has been finalized in accordance with Section 2.3.

2.4.2. Examples. The examples set forth on Schedule 2.3.3 illustrate the impact of a post-Closing adjustment to the AKD Net Asset Value on the AKD Redemption Price paid pursuant to this Section 2.4.

 $2.4.3.\ \text{Tax}$  Election. The parties agree to cause AKD to make an election under Section 754 of the Code for its taxable year that includes the Closing Date.

- 2.5. Management Agreement; Profits Interest. At the Closing, AKD and Fire Lake shall enter into the Management Agreement. AKD shall grant and issue to Fire Lake a number of AKD Profits Interest Units constituting a ****% interest (as adjusted pursuant to the terms and conditions of the Operating Agreement) in the profits and losses of AKD following the Closing Date, after giving effect to the issuance of AKD Common Units to the AKD Members and GCI pursuant to Sections 2.2 and 2.3. Fire Lake shall execute a counterpart of the Operating Agreement, and Schedule 1.3 of the Operating Agreement shall reflect the AKD Profits Interest Units of Fire Lake and the admission of Fire Lake as a member of AKD. If the number of AKD Common Units issued to GCI is increased or decreased as a result of a post-Closing adjustment to the AKD Net Asset Value pursuant to Section 2.2, then the number of AKD Profits Interest Units issued to Fire Lake pursuant to this Section 2.5 shall be proportionately adjusted. The examples set forth on Schedule 2.3.3 illustrate the impact of a post-Closing adjustment to the AKD Net Asset Value on the number of AKD Profits Interest Units issued to Fire Lake pursuant to this Section 2.5.
- 2.6. Operating Agreement. At the Closing, GCI, Parent, Fire Lake and the AKD Members shall enter into the Operating Agreement. Schedule 1.3 and Schedule 4.1 of the Operating Agreement shall reflect the number of AKD Common Units owned and capital account balances of GCI and Parent and the AKD Profits Interest Units owned by Fire Lake after giving effect to the transactions described in Sections 2.1, 2.2, 2.3, 2.4 and 2.5.
- 2.7. Closing. The consummation of the transaction contemplated by this Agreement (the "Closing") will be held by e-mail or facsimile transmission and wire transfer at 10:00 a.m. Alaska time on the third Business Day after the satisfaction or waiver of all of the conditions set forth in Section 8, unless another date or time is agreed by the Parties. At or before the Closing, each Party will deliver to the others by e-mail or facsimile transmission scanned executed originals of each Transaction Agreement to which it is a party and will deliver to the other Parties paper copies, facsimiles or

## 10 **** CONFIDENTIAL TREATMENT

e-mails of scanned copies of all other documents and instruments that it is required to deliver at or before the Closing. The documents and agreements so delivered shall for all purposes be deemed originals thereof and the signatures of the Parties thereon shall be deemed original signatures. Promptly following the Closing, each Party shall deliver by overnight courier to the appropriate other Party or Parties the executed originals of all Transaction Agreements and all other original documents required to be delivered by it at or before the Closing. The failure of any Party to deliver executed originals after the Closing shall not affect the validity of any action taken at the Closing. The date on which the Closing actually occurs shall be the "Closing Date." The transactions described in Sections 2.1, 2.2, 2.3 and 2.4 shall be deemed to occur in immediate succession and shall be effective as of 12:01 a.m., Alaska time, on the Closing Date.

# SECTION 3 REPRESENTATIONS AND WARRANTIES OF AKD, PARENT AND THE AKD MEMBERS

AKD, Parent and the AKD Members jointly and severally represent and warrant to GCI that, except as disclosed in the AKD Disclosure Schedule (which is numbered to correspond to the section numbers in this Section 3):

### 3.1. Organization; Standing and Capitalization

3.1.1. Organization and Standing of AKD. AKD is a limited liability company duly organized, validly existing and in good standing under the laws of the State of Alaska. AKD is qualified to do business and in good standing as a foreign limited liability company in the jurisdictions listed in Schedule 3.1.1, which are all of the jurisdictions where such qualification is required. AKD has the requisite limited liability company power and authority to own its assets and carry on its business as presently being conducted. Complete and correct copies of the articles of organization, operating agreement, minute books and membership records of AKD have been delivered to GCI.

3.1.2. Organization and Standing of Properties. Properties is a limited liability company duly organized, validly existing and in good standing under the laws of the State of Alaska. Properties is not required to qualify to do business as a foreign limited liability company in any jurisdiction. Properties has the requisite limited liability company power and authority to own its assets and carry on its business as presently being conducted. Complete and correct copies of the articles of organization, operating agreement, minute books and membership records of Properties have been delivered to GCI.

3.1.3. Capitalization. As of the date of this Agreement, all of the issued and outstanding membership interests of AKD are duly authorized, validly issued, fully paid and nonassessable and are owned of record and beneficially by the AKD Members, free and clear of Encumbrances, as follows:

 Member
 Points
 Membership Interest

 ---- ----- 

 Pacificom
 ****

 Red River
 ****

 ***** Class A Membership Interest

 ***** Class A Membership Interest

**** CONFIDENTIAL TREATMENT

Graystone **** *** Class B Membership Interest

As of the Closing Date, after giving effect to the reclassification of membership interests contemplated by Section 2.2 but before giving effect to the other transactions contemplated by Article 2, all of the issued and outstanding AKD Common Units will be duly authorized, validly issued, fully paid and nonassessable and owned of record and beneficially by Parent and the AKD Members, free and clean of Encumbrances, as described in Section 2.2. Except as disclosed on Schedule 3.1.3, there are no outstanding options, warrants, convertible securities or other rights to acquire any membership interest or any other security from AKD. No membership interests or other securities of AKD have been issued in violation of any preemptive or similar right of any Person or have been transferred in violation of, or are currently subject to, any right of first refusal or similar right of any Person. All of the issued and outstanding ownership interests of AKD were issued in compliance with applicable law, including, without limitation, federal and state securities laws. No securities of AKD are subject to any voting trust or other voting agreement.

3.1.4. Subsidiaries. AKD owns all of the issued and outstanding membership interest of Properties, free and clear of any Encumbrance. There are no outstanding options, warrants, convertible securities or other rights to acquire any membership interest or other security from Properties. Other than Properties, AKD does not own, directly or indirectly, any capital stock of, any partnership, equity or other ownership interest in or any security issued by any other corporation, organization, association, entity, business enterprise or other Person. Properties has no material assets other than its ownership of the Premises, has no material Liabilities other than the AKD Interest-Bearing Obligations and has not conducted any sort of business other than leasing the Premises to AKD.

### 3.2. The Transaction Agreements.

3.2.1. Execution and Validity. AKD has the requisite power and authority to execute and deliver this Agreement and to perform its obligations hereunder, and the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby have been duly and validly authorized and approved by all necessary action of the members and managers of AKD as required by the articles of organization and operating agreement of AKD and the Alaska Limited Liability Company Act. Parent and each AKD Member has the requisite power and authority to execute and deliver each of the Transaction Agreements to which it is a party and to perform its obligations thereunder, and the execution, delivery and performance by Parent or such AKD Member of this Agreement and the other Transaction Agreements to which it is a party and the consummation by Parent or such AKD Member of the transactions contemplated hereby and thereby have been duly and validly authorized and approved by all necessary action of the owners and managers of Parent and such AKD Member as required by applicable documents and law. This Agreement has been duly executed and delivered by AKD, Parent and each AKD Member and constitutes, and each of the other Transaction Agreements to which AKD, Parent or any AKD Member is a party will be duly executed and delivered by AKD, Parent or such AKD Member at Closing and will constitute, the legal, valid and binding obligation of AKD, Parent or such AKD Member, enforceable in accordance with its terms, subject to bankruptcy, insolvency, moratorium, reorganization and similar laws

12

of general applicability affecting the rights and remedies of creditors and to general principles of equity.

3.2.2. No Violation or Approval. The execution, delivery and performance by AKD, Parent and the AKD Members of the Transaction Agreements to which they are parties and the consummation of the transactions contemplated by the Transaction Agreements do not and will not constitute or result in (a) a violation of any order, judgment or decree of any court or governmental agency or body having jurisdiction over any entity in the AKD Group, Parent, any AKD Member, or any of the AKD Business Assets, or (b) a breach of or default under, or the acceleration of any obligation or creation of any Encumbrance under (whether immediately, upon the passage of time or after the giving of notice), or otherwise require a consent or waiver under, any agreement, instrument, lease, contract, mortgage, deed or license to which any entity in the AKD Group, Parent or any AKD Member is a party or by which any entity in the AKD Group, Parent or any AKD Member or any of their assets are bound or affected or (c) a violation of or a conflict with the articles of organization or operating

agreement of any entity in the AKD Group, Parent or any AKD Member. Except as described on Schedule 3.2.2, no notice to, or consent, approval, order or authorization of, or declaration or filing with, any governmental authority or entity or other Person is required to be obtained or made by any entity in the AKD Group, Parent or any AKD Member in connection with the execution, delivery and performance of or the consummation of the transactions contemplated by any of the Transaction Agreements.

### 3.3. AKD Business Assets.

3.3.1. Description. The AKD Business Assets and the Leased Real Property constitute all of the assets, properties and rights used by any entity in the AKD Group to conduct its business and necessary to conduct its business as currently conducted. Schedule 3.3.1 is a true and complete list of all depreciable tangible AKD Business Assets (including tangible AKD Business Assets leased by any entity in the AKD Group under leases that are required to be capitalized for accounting purposes), that reflects the in-service dates of such tangible AKD Business Assets, the depreciation methods and periods of such tangible AKD Business Assets and the net book value of such tangible AKD Business Assets as of May 31, 2006. Except as described in Schedule 3.3.1, neither any AKD Member nor any Affiliate of an AKD Member owns or leases any assets used in AKD's business.

3.3.2. Title. AKD has good and marketable title to all of the AKD Business Assets, free and clear of Encumbrances except Encumbrances securing current Taxes not yet due and payable.

3.3.3. Real Property. Schedule 3.3.3 contains a true and complete list of all leases and other agreements or arrangements pursuant to which any entity in the AKD Group occupies or uses any real property (the "Real Property Leases") and a description of the real property subject to each. All of the Real Property Leases are in full force and effect, and will continue to be in full force and effect following the consummation of the transactions contemplated hereby, and neither AKD nor, to knowledge of the AKD Members, any other Person is in default under any Real Property Lease. Without limiting the generality of the foregoing, AKD is current in the performance of its maintenance obligations under all Real Property Leases. The Leased Real Property

13

constitutes all of the real property, buildings and improvements used by AKD in its business. Properties has good and marketable title to the Premises, free and clear of all Encumbrances. AKD owns no real property (other than its indirect ownership of the Premises through Properties). Each parcel of Leased Real Property is supplied with utilities and other services necessary for the operation thereof. The Leased Real Property is free from material defects, has been maintained in accordance with normal industry practice, is in good operating condition and repair, subject to ordinary wear and tear reasonably to be expected in a business of the type operated by AKD, and is suitable for the purposes for which it presently is used. The Leased Real Property complies in all material respects with applicable laws, rules and regulations and all applicable declarations and covenants, has received all approvals of governmental authorities (including permits) required in connection with the occupation and operation thereof and has been occupied, operated and maintained in accordance with applicable law. AKD enjoys peaceful and undisturbed possession of all Leased Real Property.

3.3.4. Condition. The tangible AKD Business Assets, taken as a whole, are in good operating condition and repair, subject to ordinary wear and tear reasonably to be expected in a business of the type operated by AKD, and are suitable for the purposes for which they are currently used.

3.3.5. Intellectual Property. Schedule 3.3.5 contains a true and complete list of all Intellectual Property used in or related to any entity in the AKD Group's business and all registrations or filings with respect thereto. Each entity in the AKD Group owns or possesses adequate licenses or other rights to use, free and clear of any Encumbrance, all Intellectual Property listed in Schedule 3.3.5, which includes all Intellectual Property necessary to operate each entity in the AKD Group's business as currently conducted. Any Intellectual Property provided to any entity in the AKD Group by a technology vendor is used under a valid license and AKD is in compliance with such license. Each entity in the AKD Group has taken commercially reasonable actions to maintain and protect each item of Intellectual Property that it owns or uses. No entity in the AKD Group has interfered with, infringed upon, misappropriated or otherwise violated any other Person's Intellectual Property, and, to the knowledge of each entity in the AKD Group and the AKD Members, no Person has any Intellectual Property that interferes or would be likely to interfere with any entity in the AKD Group's use of any of its Intellectual Property. No proceeding is pending or, to the knowledge of any entity in the AKD Group or any AKD Member, threatened, alleging any such interference. No royalty or similar fee of any kind is payable by any entity in the AKD Group for the use of any of the Intellectual Property listed on Schedule 3.3.5, and no entity in the AKD Group has granted any Person any interest, as licensee or otherwise, in or to any of such Intellectual Property.

3.4.1. Financial Statements. Attached to this Agreement as Schedule 3.4.1 are (a) the audited consolidated balance sheet of the AKD Group as of December 31, 2005 (the "AKD Balance Sheet Date"), and the related audited statements of income, members' equity and cash flows for the fiscal year then ended, and (b) the unaudited consolidated balance sheet of the AKD Group as of May 31, 2006, and the related unaudited statements of income, members' equity and cash flows for the period covered thereby (collectively, the "AKD Financial

14

Statements"). The AKD Financial Statements were prepared from the books and records of AKD, which are correct and complete. The AKD Financial Statements present fairly and accurately the financial position of the AKD Group and the results of its operations as of the respective dates and for the periods presented therein and have been prepared in accordance with GAAP.

- 3.4.2. No Undisclosed Liabilities. The AKD Group have no Liabilities except (i) Liabilities set forth on the balance sheets included in the AKD Financial Statements, and (ii) Liabilities which have arisen after the AKD Balance Sheet Date in the ordinary course of business, consistent with historical practice (none of which Liabilities arises out of or relates to any breach of contract, breach of warranty, tort, infringement or violation of law). No entity in the AKD Group is a guarantor or otherwise liable for any Liability of any other Person.
- 3.4.3. Absence of Changes. Since the AKD Balance Sheet Date, the AKD Group has not undergone any adverse change in its business, condition (financial or otherwise) or prospects, or suffered any damage, destruction or loss (whether or not covered by insurance). Since the Balance Sheet Date, the AKD Group has operated only in the ordinary course of business, consistent in all respects with historical practice, and no change has been made or transaction entered into in anticipation of the transactions contemplated hereby. Without limiting the generality of the foregoing, since the AKD Balance Sheet Date, the AKD Group has not:
- (a) made any loan, advance or other extension of credit to any AKD Member or any officer, director or employee of any entity in the AKD Group or any Affiliate of any thereof;
- (b) increased or experienced any adverse change in any assumption underlying any method of calculating bad debts, contingencies or other reserves from that reflected in the AKD Financial Statements;
- (c) cancelled, compromised, written down, written off or waived any claim or right of substantial value;
- (d) sold, transferred, distributed or otherwise disposed of any of its assets except for sales of equipment for fair consideration in the ordinary course of business which has either been replaced by comparable equipment or is no longer necessary for the operation of AKD's business;
- (e) made any capital expenditure or commitment for additions to property, plant or equipment, except for capital expenditures or commitments (i) set forth on Schedule 3.4.3(e) or (ii) approved in advance by GCI, which approval shall not be unreasonably withheld or delayed;
- (f) made or agreed to make any increase in the compensation payable or benefits provided to any of the officers or directors of any entity in the AKD Group;
- (g) lost any key employee or key sales representative or consultant of AKD:

15

- (h) paid any severance or termination pay to any officer, director or employee of any entity in the AKD Group;
- (i) entered into, added to or modified any Plan or any other arrangement or practice relating to employees, other than (A) contributions made in accordance with its normal practice or (B) the extension of coverage to employees who became eligible after the AKD Balance Sheet Date;
- (j) changed the methods of accounting or accounting principles or practices of any entity in the AKD Group set forth in or reflected by the AKD Financial Statements;
- (k) entered into any contract or received any payment as a result of which any entity in the AKD Group would be required to provide goods or services to any Person after the Closing without receiving full payment for those goods or services at or after the time they are provided;
- (1) entered into any transaction or contract, or amended or terminated any transaction or contract, with respect to the business of any entity in the AKD Group, except normal transactions or contracts consistent in

nature and scope with prior practices and entered into in the ordinary course of business in arm's-length transactions, none of which transactions or contracts, or amendments or terminations thereof, could reasonably be expected to have a material adverse effect upon the business of any entity in the AKD Group or the financial condition or prospects thereof;

(m) terminated or been advised of the termination of or material reduction in its relationship with any material customer or supplier;

(n) changed in any material respect the business policies or practices of any entity in the AKD Group or failed to operate the business of any entity in the AKD Group in good faith and in the ordinary course; or

(o) agreed, whether in writing or not, to do any of the

foregoing.

3.4.4. Taxes. Each entity in the AKD Group has filed all Tax Returns that it has been required to file. All such Tax Returns were correct and complete in all material respects. All Taxes owed by each entity in the AKD Group (whether or not required to be shown on a Tax Return) have been paid. No entity in the AKD Group is currently the beneficiary of any extension of time within which to file any Tax Return. No entity in the AKD Group has received notice from an authority in a jurisdiction where it does not file Tax Returns that it may be subject to taxation by that jurisdiction. Each entity in the AKD Group has withheld and paid all Taxes required to have been withheld and paid in connection with amounts paid or owing to any employee, independent contractor, creditor, stockholder or other Person. Schedule 3.4.4 lists all Tax Returns filed by each entity in the AKD Group since January 1, 1999 (complete copies of which have been delivered to GCI) and indicates those Tax Returns that have been audited or are currently being audited or for which any member of the AKD Group has received notice of a proposed audit. No entity in the AKD Group has waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency. No entity in the AKD Group

16

has any liability for the Taxes of any Person under Treasury Regulation Section 1.1502-6 (or any similar provision of state, local or foreign law), as a transferee or successor, by contract, or otherwise. No entity in the AKD Group has been a member of an affiliated group filing a consolidated federal income Tax Return (or any other consolidated, combined or unitary income Tax Return). The reserves for Taxes in the AKD Financial Statements are adequate. No entity in the AKD Group is subject to any agreements that could result in any "excess parachute payments" under Code Section 280G. No entity in the AKD Group has agreed to or is otherwise required to make any adjustments pursuant to Code Section 481(a) or any similar provision of state, local or foreign law by reason of a change in accounting method. No taxing authority has proposed any such adjustment or change in accounting method with respect to any entity in the AKD Group. There is no application pending with any taxing authority requesting permission for any change in accounting method of any entity in the AKD Group. No entity in the AKD Group has participated in any investment or transaction that (i) constituted a "tax shelter" within the meaning of Section 6111(c)(1) of the Code; or (ii) would constitute a "reportable transaction" within the meaning of Treasury Regulation Section 1.6011-4, whether entered into before or after the effective date of such Treasury Regulation. None of the assets of any entity in the AKD Group are subject to any liens in respect of Taxes, except liens that are expressly permitted under the terms of this Agreement.

3.4.5. Accounts Receivable. The accounts receivable of each entity in the AKD Group (i) are validly existing, (ii) are enforceable by the AKD Group in accordance with the terms of the instruments or documents creating them, subject to bankruptcy, insolvency, moratorium, reorganization and similar laws of general applicability affecting the rights and remedies of creditors and to general principles of equity, (iii) are owned by the applicable entity in the AKD Group free and clear of all Encumbrances, (iv) represent monies due for, and have arisen solely out of, bona fide performance of services and other business transactions in the ordinary course of business consistent with past practices and (v) are collectible within one year after the Closing Date at the full recorded amount thereof less any allowance for uncollectible accounts receivable that is reflected in the Final Closing Adjustment Schedule. There are no refunds, discounts or other adjustments payable with respect to any such accounts receivable, and there are no defenses, rights of set-off, counterclaims, assignments, restrictions, encumbrances, or conditions enforceable by third parties on or affecting any account receivable.

### 3.5. Operational Matters.

3.5.1. Compliance With Law. Each entity in the AKD Group has conducted its operations in material compliance with applicable laws. The AKD Members have no knowledge of and the AKD Group has not received notice of any violations of law relating to any entity in the AKD Group, AKD's operations or the AKD Business Assets. Neither any AKD Member, any entity in the AKD Group nor, to the knowledge of AKD, Parent and the AKD Members any officer, employee or agent of any entity in the AKD Group has directly or indirectly given or agreed to give any gift, contribution, payment or similar benefit to any

supplier, customer, governmental official or employee or other Person who was, is or may be in a position to help or hinder any entity in the AKD Group or made or agreed to make any contribution, or reimbursed any political gift or contribution made by any other Person, to any candidate for United States federal, state, local or foreign public office, in any case, which would subject

# 17 **** CONFIDENTIAL TREATMENT

any entity in the AKD Group to any Liability or the failure to make which in the future could adversely affect the business or prospects of any entity in the AKD Group.

3.5.2. Litigation. There are no actions, claims, suits, audits, examinations, investigations or proceedings pending or, to the knowledge of AKD or any of the AKD Members, threatened against any entity in the AKD Group, whether by a private Person or a governmental agency or body, nor is there any reasonable basis for any such action, claim, suit, audit, examination, proceeding or investigation. No judgments, orders, decrees, citations, fines or penalties have been entered or assessed against any entity in the AKD Group.

3.5.3. Contracts. Schedule 3.5.3 contains a true and complete list of all Contracts to which or by which any entity in the AKD Group is a party or otherwise bound (other than the Real Property Leases and Subscriber Contracts) that: (a) have a duration of twelve (12) months or more and that are not terminable without penalty upon 30 days or less prior written notice; (b) require or could reasonably be expected to require any party thereto to pay \$5,000 or more; (c) are between any entity in the AKD Group and any Governmental Entity; (d) have or may have the effect of prohibiting or impairing any business practice of any entity in the AKD Group; (e) under which any entity in the AKD Group is restricted from providing services to customers or potential customers in any geographic area, during any period of time or in any segment of a market; or (f) contain any restrictive covenant or confidential or secrecy agreement other than such an agreement relating solely to information about a customer's business or any entity in the AKD Group's services to such customer (Contracts described in (a) through (f) above are referred to collectively as the "AKD Contracts"). The AKD Members have made available to GCI a true and complete copy of each AKD Contract. Neither AKD nor, to the knowledge of AKD or any AKD Member, any other party is in default under or in breach or violation of any AKD Contract, nor has an event occurred that (with or without notice, lapse of time or both) would constitute a default or breach or violation by AKD, or to the knowledge of AKD or any AKD Member, any other party, under any AKD Contract. Subject to applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent conveyance or similar laws affecting the enforcement of creditors rights generally and to general principles of equity, each AKD Contract is legal, valid, binding, enforceable and in full force and effect, and subject to applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent conveyance or similar laws affecting the enforcement of creditors rights generally and to general principles of equity, will continue to be legal, valid, binding, enforceable and in full force and effect following the consummation of the transactions contemplated hereby.

3.5.4. Subscriber Contracts. As of May 31, 2006, AKD had legal, valid, binding and enforceable Subscriber Contracts with **** customers to provide wireless telephone services (of which Subscriber Contracts, **** were for pre-paid wireless services and **** were for post-paid wireless services). As of the Closing Date, AKD will have legal, valid, binding and enforceable Subscriber Contracts with no less than **** customers to provide wireless telephone services (of which Subscriber Contracts, at least **** will be for pre-paid wireless services and **** will be for post-paid wireless services). The consummation of the transactions contemplated hereby will have no effect on whether each such Subscriber Contract in effect as of the Closing Date will continue to be legal, valid, binding, enforceable and in full force and effect following the Closing Date. The AKD Members have made available to GCI a true and complete copy of each general form of Subscriber Contract used by AKD (and

18

not each Subscriber Contract with every individual customer). At the Closing, AKD shall deliver a certificate signed on behalf of AKD by an authorized person stating (i) the total number of wireless telephone subscribers of AKD, (ii) the number of subscribers for pre-paid wireless services and (iii) the number of subscribers for post-paid wireless services, in each case, as of the end of the month immediately preceding the Closing Date.

3.5.5. Transactions With Affiliates. Except as disclosed on Schedule 3.5.5, no AKD Member, no director, officer or employee of any entity in the AKD Group and no Affiliate of any thereof is currently a party to any AKD Contract or has been a party to any material transaction with any entity in the AKD Group since December 31, 2004.

3.5.6. Insurance. Schedule 3.5.6 is a true and complete list of all insurance policies (including self-insurance arrangements) maintained by any entity in the AKD Group (including coverage limits, deductibles, named insureds and policy periods), all of which policies are in full force and effect. No insurance company with which any entity in the AKD

Group has had a policy has ever terminated or declined to renew such insurance.

3.5.7. Books and Records. The AKD Members have made available to GCI true and correct copies of all books and records of each entity in the AKD Group.

3.5.8. Bank Accounts; Powers of Attorney. Schedule 3.5.8 is a true and complete list of all bank accounts, securities accounts and other financial accounts maintained by or for the benefit of any entity in the AKD Group, including the name of the institution at which the account is maintained, the name and number of the account, the purpose of the account and the names and capacities of all persons authorized to sign on the account. Schedule 3.5.8 also lists all powers of attorney or similar instruments signed by any entity in the AKD Group authorizing any person to act on its behalf with respect to any matter.

### 3.6. Employee Matters.

3.6.1. AKD Plans. Schedule 3.6.1 is a true and complete list of all of the following that are now or have ever been maintained or contributed to by AKD or any entity in the AKD Group or any other entity that is a member of a "controlled group" with AKD, as determined under ERISA Section 4001(a)(14) (collectively, the "AKD Plans"): (a) any nonqualified deferred compensation or retirement plan or arrangement which is an "Employee Pension Benefit Plan" as defined in Section 3(2) of ERISA; (b) any qualified defined contribution retirement plan or arrangement which is an Employee Pension Benefit Plan; (c) any qualified defined benefit retirement plan or arrangement which is an Employee Pension Benefit Plan (including any "Multiemployer Plan", as defined in Section 3(37) of ERISA); or (d) any "Employee Welfare Benefit Plan" as defined in Section 3(1) of ERISA. No AKD Plan is a Multiemployer Plan, and no AKD Plan is subject to the provisions of Title IV of ERISA. No AKD Plan is maintained in connection with any trust described in Code Section 501(c)(9). There have been no prohibited transactions with respect to any AKD Plan. No "Fiduciary" (as defined in ERISA Section 3(21)) has any Liability for breach of fiduciary duty or any other failure to act or comply in connection with the administration or investment of the assets of any such AKD Plan. No entity in the AKD Group currently maintains or contributes to any AKD Plan providing health or medical benefits for current or future retired or terminated

1 0

employees, their spouses or their dependents (other than in accordance with Code Section 4980B). Each AKD Plan has been maintained and administered in compliance with its terms and with all applicable legal requirements, including but not limited to ERISA, the Code, and the Consolidated Omnibus Budget Reconciliation Act of 1985. Nothing done or omitted to be done and no transaction or holding of any asset under or in connection with any AKD Plan has made or will make any entity in the AKD Group, or any employee, officer or director of any entity in the AKD Group, or any fiduciary with respect to such AKD Plan, subject to any Liability under Title I of ERISA or any Liability for any tax under Code Section 4972 or Code Sections 4975 through 4980B, inclusive. No condition exists that would prevent any entity in the AKD Group from amending or terminating any AKD Plan to the extent permitted by applicable law. To the knowledge of the AKD Members and based upon the requirements of Code Section 409A and the quidance issued by the Internal Revenue Service, including Notice 2005-1 and the proposed regulations published on October 4, 2005, each AKD Plan that is a "nonqualified deferred compensation plan" as defined in Code Section 409A(d)(1) has been operated in material compliance with Code Section 409A. Each entity in the AKD Group has established and implemented such policies, programs, procedures, contracts and systems as are necessary to bring each entity in the AKD Group into compliance with HIPAA; Title II, Subtitle F, Sections 261-264, Public Law 104-91; and the Standards for Privacy of Individually Identifiable Health Information, 45 C.F.R. Parts 160-164 as of the effective dates of such laws, except where the failure to do so would not reasonably be expected to have a material adverse effect.

3.6.2. Employees. Schedule 3.6.2 is a true and complete list of all employees of any entity in the AKD Group and shows for each such employee: (i) his or her position and title; (ii) his or her date of hire; (iii) his or her salary; (iv) his or her unpaid wages, accrued vacation time and accrued personal time as of May 31, 2006; and (v) any bonuses paid to him or her with respect to the fiscal year ended December 31, 2005 or earned by or promised to him or her with respect to the current fiscal year.

3.6.3. Labor Relations. There is no dispute or controversy between any entity in the AKD Group and any of its employees. No entity in the AKD Group is a party to any collective bargaining agreement with respect to any of its employees, none of its employees is represented by a labor union and, to the knowledge of AKD and the AKD Members, there is no labor union organizing activity by or among its employees.

3.6.4. Certain Agreements. Neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated hereby will (a) result in any payment (including, without limitation, severance, unemployment compensation, parachute payment, bonus or otherwise) becoming due to any employee or any other Person from any entity in the AKD Group under any

AKD Plan, agreement or otherwise, (b) increase any benefits otherwise payable to any employee under any AKD Plan or agreement, or (c) result in the acceleration of the time of payment or vesting of any such benefits. None of the employees of any entity in the AKD Group is subject to any covenant against competition or similar agreement that would limit his or her ability to participate in all aspects of any entity in the AKD Group's business at any present or future location.

20

### 3.7. Environmental Matters.

3.7.1. Compliance. Each entity in the AKD Group is conducting and at all times has conducted its business and operations, and has occupied, used and operated all real property and facilities presently or previously owned, occupied, used or operated by any entity in the AKD Group, in compliance (in all material respects) with all Environmental Laws and so as not to give rise to Liability under any Environmental Laws or to any adverse impact on any entity in the AKD Group's business or activities. Neither any entity in the AKD Group nor the AKD Members have any knowledge of pending or proposed changes to any Environmental Laws that would require any changes in any of any entity in the AKD Group's premises, facilities, equipment, operations or procedures or affect any entity in the AKD Group's business or its cost of conducting its business as now conducted. To the knowledge of AKD, Parent and the AKD Members, no conditions, circumstances or activities have existed or currently exist (including, without limitation, off-site disposal or treatment of Hazardous Substances) which could give rise to any Liability pursuant to any Environmental Laws. Other than the Leased Real Property, neither any entity in the AKD Group nor its predecessors have at any time owned, occupied, used or operated any real property or facilities.

3.7.2. Waste Materials. Any chemicals and chemical compounds and mixtures which are included among the assets of any entity in the AKD Group are required for the conduct of any entity in the AKD Group's business, have not been and are not intended to be discarded or abandoned, and are not waste or waste materials. No entity in the AKD Group has generated, handled, used, transported or disposed of Hazardous Substances. All waste materials which are generated as part of the business of any entity in the AKD Group are handled, stored, treated and disposed of in accordance with applicable Environmental Laws.

3.7.3. Tanks; Asbestos. Any underground storage tanks ever located at the Leased Real Property have been removed in compliance with all applicable Environmental Laws, all remediation required in connection with such removal has been completed in accordance with applicable Environmental Laws and all governmental agencies having jurisdictions have approved such removal and remediation and issued appropriate certificates reflecting that no further action is required. All above ground storage tanks located at the Leased Real Property comply with applicable Environmental Laws and are appropriate and adequate for the conduct of each entity in the AKD Group's business. No real properties or facilities presently or previously owned, occupied, used or operated by any entity in the AKD Group or any predecessor have been used at any time as a gasoline service station or as a facility for storing, pumping, dispensing or producing gasoline or any other petroleum products or any other Hazardous Substances. No building or other structure on any of the real property owned, occupied, used or operated by any entity in the AKD Group contains asbestos or asbestos-containing materials. There are not nor have there been any incinerators, septic tanks, leach fields, cesspools or wells (including without limitation dry, drinking, industrial, agricultural and monitoring wells) on any real property owned, occupied, used or operated by any entity in the AKD Group.

3.7.4. Environmental Assessments. The AKD Members have delivered to GCI correct and complete copies of all documents, correspondence, reports or other materials in their possession or control concerning the environmental condition of any real property currently or formerly used or

21

occupied by any entity in the AKD Group, including, without limitation, all environmental site assessments.

### 3.8. Communications Regulatory Matters.

3.8.1. AKD Licenses. AKD is fully qualified under the Communications Laws to be an FCC licensee. Schedule 3.8.1 lists all licenses and authorizations issued by the FCC or the RCA to any entity in the AKD Group (the "AKD Licenses"), together with the name of the licensee or authorization holder, the expiration date of the AKD Licenses and, where applicable, the relevant FCC market designation. AKD validly holds the AKD Licenses which represent all the FCC authorizations required in connection with the ownership and operation of the AKD wireless telecommunications business as it is presently being conducted. The AKD Licenses are not subject to any conditions outside of the ordinary course. No person other than AKD has any right, title or interest (legal or beneficial) in or to, or any right or license to use, the AKD Licenses. The AKD Licenses have been granted to AKD by Final Order and are in full force and effect.

3.8.2. Compliance. Except as disclosed in Schedule 3.8.2, AKD is in material compliance with the Communications Laws, including without limitation those relating to: (i) the Communications Assistance for Law Enforcement Act (CALEA); (ii) E-911 Phase I and Phase II compliance; (iii) number porting, number pooling and related number usage and utilization reports; (iv) Telecommunications Relay Service obligations; (v) universal service obligations; (vi) the payment of regulatory fees; (vii) Text Telephone Devices (TTY); (viii) the submission of quarterly, semi-annual, annual or other periodic reports or filings with the FCC or other Governmental Entity or administrative body (e.g. the National Exchange Carrier Association (NECA) and the Universal Service Administrative Company (USAC)); (ix) compliance with the National Environmental Protection Act (NEPA) provisions applicable to telecommunications carriers; (x) compliance with any spectrum clearing or incumbent relocation cost sharing obligations; and (xi) compliance with FCC and FAA antenna registration and painting and lighting requirements.

3.8.3. Proceedings. There are no objections, petitions to deny, complaints (formal or informal) competing applications or other proceedings pending before the FCC or any other Governmental Entity having jurisdiction over AKD or the AKD Licenses relating to AKD or the AKD Licenses. AKD has not received any notice of any claim of default with respect to any of the AKD Licenses. Except for proceedings affecting the wireless industry generally, and except as disclosed on Schedule 3.8.3, there is not pending or, to the knowledge of AKD, threatened against AKD or the AKD Licenses any action, petition, objection or other pleading, or any proceeding with the FCC or any other Governmental Entity, which contests the validity of, or seeks the revocation, forfeiture, non-renewal modification or suspension of, the AKD Licenses, or which would adversely affect the ability of AKD to consummate the transactions contemplated by this Agreement.

3.8.4. Filings. All documents required to be filed during the ownership of the AKD Licenses by AKD with the FCC or any other Governmental Entity have been timely filed or the time period for such filing has not lapsed, except where such failure to timely file would not reasonably be expected to result in the revocation, cancellation, forfeiture, non-renewal or

22

suspension of any authorization or license or the imposition of any monetary forfeiture. All of such filings were complete and correct in all material respects when filed.

3.8.5. Build-Out. AKD is not in breach or otherwise in violation of any FCC build-out requirements with respect to any FCC authorization held by it. Each FCC licensed station has been built out at least to the minimum extent required by the Communications Laws. Any and all FCC notifications or filings associated with the build-out were timely filed and were true complete and correct when filed. There has been no discontinuance of service subsequent to the completion of construction and certification that would cause the AKD Licenses to be deemed forfeited or automatically cancelled by the FCC.

### 3.9. Transactional Matters.

3.9.1. Brokers, Finders, etc. All negotiations relating to this Agreement and the transactions contemplated hereby have been carried on without the intervention of any Person acting on behalf of any entity in the AKD Group or any AKD Member in such manner as to give rise to any valid claim against any entity in the AKD Group, any AKD Member or GCI for any brokerage or finder's commission, fee or similar compensation.

3.9.2. Disclosure. The representations and warranties of each entity in the AKD Group and the AKD Members in this Agreement, including the Schedules hereto, taken as a whole, do not contain any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements contained not misleading.

### SECTION 4

REPRESENTATIONS AND WARRANTIES OF THE DENALI MEMBERS

The Denali Members jointly and severally represent and warrant to GCI that, except as disclosed in the Denali Disclosure Schedule (which is numbered to correspond to the section numbers in this Section 4):

4.1. Organization and Standing; Capitalization.

4.1.1. Organization and Standing of Denali. Denali is a limited liability company duly organized, validly existing and in good standing under the laws of the State of Alaska. Denali is qualified to do business and in good standing as a foreign limited liability company in the jurisdictions listed in Schedule 4.1.1, which are all of the jurisdictions where such qualification is required. Denali has the requisite limited liability company power and authority to own its assets and carry on its business as presently being conducted. Complete and correct copies of the articles of organization, operating agreement and membership records of Denali have been delivered to GCI.

4.1.2. Subsidiaries. Denali does not own, directly or indirectly, any capital stock of, any partnership, equity or other ownership interest in or any security issued by any other corporation, organization, association, entity, business enterprise or other Person.

## 23 **** CONFIDENTIAL TREATMENT

4.1.3. Capitalization. All of the issued and outstanding Denali Interests are duly authorized, validly issued, fully paid and nonassessable and are owned of record and beneficially by the Denali Members, free and clear of Encumbrances, as follows:

There are no outstanding options, warrants, convertible securities or other rights to acquire any membership interest or any other security from Denali. No membership interests or other securities of Denali have been issued in violation of any preemptive or similar right of any Person or have been transferred in violation of, or are currently subject to, any right of first refusal or similar right of any Person. All of the issued and outstanding ownership interests of Denali were issued in compliance with applicable law, including, without limitation, federal and state securities laws. No securities of Denali are subject to any voting trust or other voting agreement.

### 4.2. The Transaction Agreements.

4.2.1. Execution and Validity. Each Denali Member has the requisite power and authority to execute and deliver each of the Transaction Agreements to which it is a party and to perform its obligations thereunder, and the execution, delivery and performance by such Denali Member of this Agreement and the other Transaction Agreements to which it is a party and the consummation by such Denali Member of the transactions contemplated hereby and thereby have been duly and validly authorized and approved by all necessary action of the owners and managers of such Denali Member as required by applicable documents and law. This Agreement has been duly executed and delivered by each Denali Member and constitutes, and each of the other Transaction Agreements to which any Denali Member is a party will be duly executed and delivered by such Denali Member at Closing and will constitute, the legal, valid and binding obligation of such Denali Member, enforceable in accordance with its terms, subject to bankruptcy, insolvency, moratorium, reorganization and similar laws of general applicability affecting the rights and remedies of creditors and to general principles of equity.

4.2.2. No Violation or Approval. The execution, delivery and performance by each Denali Member of the Transaction Agreements to which such Denali Member is a party and the consummation of the transactions contemplated by the Transaction Agreements do not and will not constitute or result in (a) a violation of any order, judgment or decree of any court or governmental agency or body having jurisdiction over Denali, any Denali Member, or any of their respective assets, or (b) a breach of or default under, or the acceleration of any obligation or creation of any Encumbrance under (whether immediately, upon the passage of time or after the giving of notice), or otherwise require a consent or waiver under, any agreement, instrument, lease, contract, mortgage, deed or license to which Denali or any Denali Member is a party or by which Denali or any Denali Member or any of their assets are bound or affected or (c) a violation of or a conflict with the articles of organization or operating agreement of Denali. Except as described on Schedule 4.2.2, no notice to, or consent, approval, order or authorization of, or declaration or filing with, any governmental authority or entity or other Person is required to be obtained or made by Denali or any Denali Member in connection

24

with the execution, delivery and performance of or the consummation of the transactions contemplated by any of the Transaction Agreements.

 $4.3.\ \mbox{No Assets.}$  Denali does not own, lease or use any assets or properties other than the Denali Contracts and the Denali Licenses.

### 4.4. Financial Matters.

4.4.1. Financial Statements. Attached to this Agreement as Schedule 4.4.1 are (a) the unaudited balance sheet of Denali as of December 31, 2005 (the "Denali Balance Sheet Date"), and the related audited statements of income, members' equity and cash flows for the fiscal year then ended, and (b) the unaudited balance sheets of Denali as of May 31, 2006, and the related audited statement of income, members' equity and cash flows for the quarter then ended (collectively, the "Denali Financial Statements"). The Denali Financial Statements were prepared from the books and records of Denali, which are correct and complete. The Denali Financial Statements present fairly and accurately the financial position of Denali and the results of its operations as of the respective dates and for the periods presented therein and have been prepared in

accordance with GAAP.

- 4.4.2. No Undisclosed Liabilities. Denali has no Liabilities except (i) Liabilities set forth on the balance sheet included in the Denali Financial Statements, and (ii) Liabilities which have arisen after the Denali Balance Sheet Date in the ordinary course of business, consistent with historical practice (none of which Liabilities arises out of or relates to any breach of contract, breach of warranty, tort, infringement or violation of law). Denali is not a guarantor or otherwise liable for any Liability of any other Person.
- 4.4.3. Absence of Changes. Since the Denali Balance Sheet Date, Denali has not undergone any adverse change in its business, condition (financial or otherwise) or prospects, or suffered any damage, destruction or loss (whether or not covered by insurance). Since the Denali Balance Sheet Date, Denali has operated only in the ordinary course of business, consistent in all respects with historical practice, and no change has been made or transaction entered into in anticipation of the transactions contemplated hereby. Without limiting the generality of the foregoing, since the Denali Balance Sheet Date, Denali has not:
- (a) made any loan, advance or other extension of credit to any Denali Member or any officer, director or employee of Denali or any Affiliate of any thereof;
- (b) increased or experienced any adverse change in any assumption underlying any method of calculating bad debts, contingencies or other reserves from that reflected in the Denali Financial Statements;
- (c) cancelled, compromised, written down, written off or waived any claim or right of substantial value;
- (d) sold, transferred, distributed or otherwise disposed of any of its assets except for sales of equipment for fair consideration in the ordinary course of business which has either been replaced by comparable equipment or is no longer necessary for the operation of Denali's business;

2.5

- (e) made any capital expenditure or commitment for additions to property, plant or equipment;
- $\hbox{ (f) made or agreed to make any increase in the compensation payable or benefits provided to any of the officers or directors of Denali;}$
- (g) paid any severance or termination pay to any officer, director or employee of Denali;
- (h) changed the methods of accounting or accounting principles or practices of Denali set forth in or reflected by the Financial Statements:
- (i) entered into any contract or received any payment as a result of which Denali would be required to provide goods or services to any Person after the Closing without receiving full payment for those goods or services at or after the time they are provided;
- (j) entered into any transaction or contract, or amended or terminated any transaction or contract, with respect to the business of Denali, except normal transactions or contracts consistent in nature and scope with prior practices and entered into in the ordinary course of business in arm's-length transactions, none of which transactions or contracts, or amendments or terminations thereof, could reasonably be expected to have a material adverse effect upon the business of Denali or the financial condition or prospects thereof;
- (k) terminated or been advised of the termination of or material reduction in its relationship with any material customer or supplier;
- (1) changed in any material respect the business policies or practices of Denali or failed to operate the business of Denali in good faith and in the ordinary course; or
- $$\left(m\right)$$  agreed, whether in writing or not, to do any of the foregoing.
- 4.4.4. Taxes. Denali has filed all Tax Returns that it has been required to file. All such Tax Returns were correct and complete in all material respects. All Taxes owed by Denali (whether or not required to be shown on a Tax Return) have been paid. Denali is not currently the beneficiary of any extension of time within which to file any Tax Return. Denali has not received notice from an authority in a jurisdiction where it does not file Tax Returns that it may be subject to taxation by that jurisdiction. Denali has withheld and paid all Taxes required to have been withheld and paid in connection with amounts paid or owing to any employee, independent contractor, creditor,

stockholder or other Person. Schedule 4.4.4 lists all Tax Returns filed by Denali since January 1, 1999 (complete copies of which have been delivered to GCI) and indicates those Tax Returns that have been audited or are currently being audited or for which Denali has received notice of a proposed audit. Denali has not waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency. Denali has no liability for the Taxes of any Person under Treasury Regulation Section 1.1502-6 (or any similar provision of state, local or foreign law), as a transferee or successor, by contract, or otherwise. Denali has never been a member of an affiliated group filing a consolidated federal income Tax Return (or any other consolidated, combined or unitary income Tax Return). The reserves for Taxes in the Denali Financial Statements are adequate. Denali is not subject

2.6

to any agreements that could result in any "excess parachute payments" under Code Section 280G. Denali has not agreed to and is not otherwise required to make any adjustments pursuant to Code Section 481(a) or any similar provision of state, local or foreign law by reason of a change in accounting method. No taxing authority has proposed any such adjustment or change in accounting method with respect to Denali. There is no application pending with any taxing authority requesting permission for any change in accounting method of Denali. Denali has not participated in any investment or transaction that (i) constituted a "tax shelter" within the meaning of Section 6111(c)(1) of the Code; or (ii) would constitute a "reportable transaction" within the meaning of Treasury Regulation Section 1.6011-4, whether entered into before or after the effective date of such Treasury Regulation. None of Denali's assets are subject to any liens in respect of Taxes, except liens that are expressly permitted under the terms of this Agreement.

4.4.5. Accounts Receivable. The accounts receivable of Denali (i) are validly existing, (ii) are enforceable by Denali in accordance with the terms of the instruments or documents creating them, (iii) are owned by Denali free and clear of all Encumbrances, (iv) represent monies due for, and have arisen solely out of, bona fide performance of services and other business transactions in the ordinary course of business consistent with past practices and (v) are collectible within one year after the Closing Date at the full recorded amount thereof less any allowance for uncollectible accounts receivable that is reflected in the Final Closing Adjustment Schedule. There are no refunds, discounts or other adjustments payable with respect to any such accounts receivable, and there are no defenses, rights of set-off, counterclaims, assignments, restrictions, encumbrances, or conditions enforceable by third parties on or affecting any account receivable.

## 4.5. Operational Matters.

4.5.1. Compliance With Law. Denali has conducted its operations in material compliance with applicable laws. The Denali Members have no knowledge of and Denali has not received notice of any violations of law relating to Denali, Denali's operations or the Denali Contributed Assets. Neither any Denali Member, Denali nor any officer, employee or agent of Denali has directly or indirectly given or agreed to give any gift, contribution, payment or similar benefit to any supplier, customer, governmental official or employee or other Person who was, is or may be in a position to help or hinder Denali or made or agreed to make any contribution, or reimbursed any political gift or contribution made by any other Person, to any candidate for United States federal, state, local or foreign public office, in any case, which would subject Denali to any Liability or the failure to make which in the future could adversely affect the business or prospects of Denali.

4.5.2. Litigation. There are no actions, claims, suits, audits, examinations, investigations or proceedings pending or, to the knowledge of Denali or the Denali Members, threatened against Denali, whether by a private Person or a governmental agency or body, nor is there any reasonable basis for any such action, claim, suit, audit, examination, proceeding or investigation. No judgments, orders, decrees, citations, fines or penalties have been entered or assessed against Denali.

2.

4.5.3. Contracts. Schedule 4.5.3 contains a true and complete list of all Contracts to which or by which Denali is a party or otherwise bound that: (a) have a duration of twelve (12) months or more and that are not terminable without penalty upon 30 days or less prior written notice; (b) require or could reasonably be expected to require any party thereto to pay \$5,000 or more; (c) are between Denali and any Governmental Entity; (d) have or may have the effect of prohibiting or impairing any business practice of Denali or any entity in the AKD Group; (e) under which Denali or any entity in the AKD Group is restricted from providing services to customers or potential customers in any geographic area, during any period of time or in any segment of a market; or (f) contain any restrictive covenant or confidential or secrecy agreement other than such an agreement relating solely to information about a customer's business or Denali or any entity in the AKD Group's services to such customer (Contracts described in (a) through (f) above are referred to collectively as the "Denali Contracts"). The Denali Members have made available to GCI a true and complete copy of each Contract. Neither Denali nor, to the knowledge of Denali or any Denali Member, any other party is in default under or in breach or violation of any Denali Contract, nor has an event occurred that (with or without notice, lapse of time or both) would constitute a default or breach or violation by Denali, or to the knowledge of Denali or any Denali Members, any other party, under any Denali Contract. Subject to applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent conveyance or such laws affecting the enforcement of creditors rights generally and to general principles of equity, each Denali Contract is legal, valid, binding, enforceable and in full force and effect, and subject to applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent conveyance or similar laws affecting the enforcement of creditors rights generally and to general principles of equity, will continue to be legal, valid, binding, enforceable and in full force and effect following the consummation of the transactions contemplated hereby.

- 4.5.4. Transactions With Affiliates. No Denali Member, no director, officer, manager or employee of Denali and no Affiliate of any thereof is currently a party to any Denali Contract or has been a party to any material transaction with Denali since December 31, 2004.
- 4.5.5. Insurance. Schedule 4.5.5 is a true and complete list of all insurance policies (including self-insurance arrangements) maintained by Denali (including coverage limits, deductibles, named insureds and policy periods), all of which policies are in full force and effect. No insurance company with which Denali has had a policy has ever terminated or declined to renew such insurance
- 4.5.6. Books and Records. The Denali Members have made available to GCI true and correct copies of all books and records of Denali.
- 4.5.7. Bank Accounts; Powers of Attorney. Schedule 4.5.7 is a true and complete list of all bank accounts, securities accounts and other financial accounts maintained by or for the benefit of Denali, including the name of the institution at which the account is maintained, the name and number of the account, the purpose of the account and the names and capacities of all persons authorized to sign on the account. Schedule 4.5.7 also lists all powers of attorney or similar instruments signed by Denali authorizing any person to act on its behalf with respect to any matter.

2.8

- 4.6. Employee Matters. Denali has no employees and has never had any employees since its inception. Neither Denali nor any other entity that is a member of a controlled group with Denali, as determined under ERISA Section 4001(a)(14), has ever maintained or contributed to any of the following: (a) any nonqualified deferred compensation or retirement plan or arrangement which is an "Employee Pension Benefit Plan" as defined in Section 3(2) of ERISA; (b) any qualified defined contribution retirement plan or arrangement which is an Employee Pension Benefit Plan; (c) any qualified defined benefit retirement plan or arrangement which is an Employee Pension Benefit Plan; (c) any qualified defined benefit retirement plan or arrangement which is an Employee Pension Benefit Plan (including any "Multiemployer Plan", as defined in Section 3(37) of ERISA); or (d) any "Employee Welfare Benefit Plan" as defined in Section 3(1) of ERISA.
- 4.7. Environmental Matters. Denali is conducting and at all times has conducted its business and operations, and has occupied, used and operated all real property and facilities presently or previously owned, occupied, used or operated by Denali, in compliance (in all material respects) with all Environmental Laws and so as not to give rise to Liability under any Environmental Laws or to any adverse impact on Denali's business or activities. Neither Denali nor any Denali Members have any knowledge of pending or proposed changes to any Environmental Laws that would require any changes in any of Denali's premises, facilities, equipment, operations or procedures or affect Denali's business or its cost of conducting its business as now conducted. No conditions, circumstances or activities have existed or currently exist (including, without limitation, off-site disposal or treatment of Hazardous Substances) which could give rise to any Liability pursuant to any Environmental Laws. Neither Denali nor its predecessors have at any time owned, occupied, used or operated any real property or facilities.

## 4.8. Communications Regulatory Matters.

4.8.1. Denali Licenses. Denali is fully qualified under the Communications Laws to be an FCC licensee. Schedule 4.8.1 lists all licenses and authorizations issued by the FCC or the RCA to Denali (the "Denali Licenses"), together with the name of the licensee or authorization holder, the expiration date of the Denali Licenses and, where applicable, the relevant FCC market designation. Denali validly holds the Denali Licenses which represent all the FCC authorizations required in connection with the ownership and operation of the Denali wireless telecommunications business as it is presently being conducted. The Denali Licenses are not subject to any conditions outside of the ordinary course. No person other than Denali has any right, title or interest (legal or beneficial) in or to, or any right or license to use, the Denali Licenses. The Denali Licenses have been granted to Denali by Final Order and are in full force and effect.

4.8.2. Compliance. Denali is in material compliance with the Communications Laws, including without limitation those relating to: (i) the Communications Assistance for Law Enforcement Act (CALEA); (ii) E-911 Phase I

and Phase II compliance; (iii) number porting, number pooling and related number usage and utilization reports; (iv) Telecommunications Relay Service obligations; (v) universal service obligations; (vi) the payment of regulatory fees; (vii) Text Telephone Devices (TTY); (viii) the submission of quarterly, semi-annual, annual or other periodic reports or filings with the FCC or other Governmental Entity or administrative body (e.g. the National Exchange Carrier

29

Association (NECA) and the Universal Service Administrative Company (USAC)); (ix) compliance with the National Environmental Protection Act (NEPA) provisions applicable to telecommunications carriers; (x) compliance with any spectrum clearing or incumbent relocation cost sharing obligations; and (xi) compliance with FCC and FAA antenna registration and painting and lighting requirements.

4.8.3. Proceedings. There are no objections, petitions to deny, complaints (formal or informal) competing applications or other proceedings pending before the FCC or any other Governmental Entity having jurisdiction over Denali or the Denali Licenses relating to Denali or the Denali Licenses. Denali has not received any notice of any claim of default with respect to any of the Denali Licenses. Except for proceedings affecting the wireless industry generally, and except as disclosed on Schedule 4.8.3, there is not pending or, to the knowledge of Denali, threatened against Denali or the Denali Licenses any action, petition, objection or other pleading, or any proceeding with the FCC or any other Governmental Entity, which contests the validity of, or seeks the revocation, forfeiture, non-renewal modification or suspension of, the Denali Licenses, or which would adversely affect the ability of Denali to consummate the transactions contemplated by this Agreement.

4.8.4. Filings. All documents required to be filed during the ownership of the Denali Licenses by Denali with the FCC or any other Governmental Entity have been timely filed or the time period for such filing has not lapsed, except where such failure to timely file would not reasonably be expected to result in the revocation, cancellation, forfeiture, non-renewal or suspension of any authorization or license or the imposition of any monetary forfeiture. All of such filings were complete and correct in all material respects when filed.

4.8.5. Build-Out. Denali is not in breach or otherwise in violation of any FCC build-out requirements with respect to any FCC authorization held by it. Each FCC licensed station has been built out at least to the minimum extent required by the Communications Laws. Any and all FCC notifications or filings associated with the build-out were timely filed and were true complete and correct when filed. There has been no discontinuance of service subsequent to the completion of construction and certification that would cause the Denali Licenses to be deemed forfeited or automatically cancelled by the FCC.

### 4.9. Transactional Matters.

4.9.1. Brokers, Finders, etc. All negotiations relating to this Agreement and the transactions contemplated hereby have been carried on without the intervention of any Person acting on behalf of Denali or any Denali Member in such manner as to give rise to any valid claim against Denali, any Denali Member or GCI for any brokerage or finder's commission, fee or similar compensation.

4.9.2. Disclosure. The representations and warranties of the Denali Members in this Agreement, including the Schedules hereto, taken as a whole, do not contain any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements contained not misleading.

### 30 SECTION 5 REPRESENTATIONS AND WARRANTIES OF GCI

 $\,$  GCI represents and warrants to AKD, Parent, the AKD Members and the Denali Members that:

- 5.1. Organization and Standing. GCI is a corporation duly incorporated, validly existing and in good standing under the laws of the State of Alaska. GCI has the requisite corporate power and authority to execute and deliver each of the Transaction Agreements to which it is a party and to perform its obligations thereunder. The execution, delivery and performance by GCI of this Agreement and the other Transaction Agreements to which it is a party, and the consummation by GCI of the transactions contemplated hereby and thereby, have been duly and validly authorized and approved by all necessary corporate action on the part of GCI.
- 5.2. Execution and Validity of Agreements. This Agreement has been duly executed and delivered by GCI and constitutes, and each of the other Transaction Agreements to which GCI is a party will be duly executed and delivered at Closing and will constitute, the legal, valid and binding obligation of GCI, enforceable in accordance with its terms, subject to bankruptcy, insolvency, moratorium, reorganization and similar laws of general

applicability affecting the rights and remedies of creditors and to general principles of equity, regardless of whether enforcement is sought in proceedings in equity or at law.

- 5.3. No Violation or Approval. The execution, delivery and performance by GCI of each of the Transaction Agreements to which it is a party and the consummation of the transactions contemplated thereby does not and will not result in (a) a violation of any law, rule or regulation, order, judgment or decree applicable to GCI or any order, judgment or decree of any court or any governmental agency or body having jurisdiction over GCI or its properties or assets, (b) a breach or a default under (whether immediately, upon the passage of time or after giving notice), or the acceleration of any payment under any material agreement, instrument, lease, contract, mortgage, or license to which GCI is a party or by which it or any of its properties or assets is bound, or (c) a violation of or a conflict with its charter or bylaws. No consent, approval, order or authorization of, or declaration or filing with, any governmental authority or entity or other party is required to be, and has not been, obtained or made by GCI in connection with the execution, delivery and performance of or the consummation of the transactions contemplated by any of the Transaction Agreements.
- 5.4. Brokers, Finders, etc. All negotiations relating to this Agreement and the transactions contemplated hereby have been carried on without the intervention of any Person acting on behalf of GCI in such manner as to give rise to any valid claim against any AKD, the AKD Members or the Denali Members for any brokerage or finder's commission, fee or similar compensation.
- 5.5. No Oral Warranties. Except for the warranties and representations expressly set forth in this Agreement, GCI is relying solely on its own expertise and that of its consultants, employees, agents, servants and representatives in connection with its acquisition of the Denali Interests and the AKD Common Units (collectively, the "Interests"). GCI has or will conduct

31

such investigation, analysis and inspection of AKD, Denali and their respective businesses as it deems necessary or appropriate and shall rely upon such investigation, analysis and inspection in connection with the acquisition of the Interests. GCI acknowledges and agrees that neither Denali, AKD, Parent, the AKD Members and/or the Denali Members, on the one hand, nor their respective officers, directors, members, employees, servants, agents, attorneys or representatives have made any oral agreements, warranties or representations, and that none of Denali, AKD, Parent, the AKD Members and/or the Denali Members are liable or bound by any oral or written statement, agreement, information, warranty or representation with respect to the Interests, AKD, Denali of the business or assts of AKD or Denali made by any Person whatsoever, unless such statement, agreement, information, warranty or representation is expressly set forth in this Agreement. In connection with its acquisition of the Interests, GCI is not relying on any oral statements or representations made or given by any Person whomsoever. The parties agree that this Section 5.5 shall in no way diminish or otherwise impair GCI's ability (a) to seek indemnification for any breach of any representation or warranty herein or (b) to enforce its rights under this Agreement or the documents and instruments delivered pursuant to this Agreement.

## SECTION 6

6.1. Exclusivity; Acquisition Proposals. Unless and until this Agreement shall have been terminated pursuant to Section 11, none of any entity in the AKD Group, Parent, the AKD Members or the Denali Members shall take or cause or permit to be taken, directly or indirectly, any of the following actions with any Person other than GCI and its designees or agents: (a) solicit, encourage, initiate or participate in any negotiations, inquiries or discussions with respect to any offer or proposal to acquire the business or assets of any entity in the AKD Group or Denali, whether by merger, consolidation, other business combination, purchase of assets or stock, tender or exchange offer or otherwise (each of the foregoing an "Acquisition Transaction"); (b) disclose any information not customarily disclosed to any Person who is or may be requesting such information for purposes of a possible Acquisition Transaction; (c) agree to or execute any letter of intent, terms sheet or agreement relating to an Acquisition Transaction; or (d) make or authorize any public statement or solicitation with respect to any Acquisition Transaction or any offer or proposal relating to an Acquisition Transaction other than with respect to the transactions contemplated hereby.

### 6.2. Notices and Consents.

(a) Each entity in the AKD Group, Parent, the AKD Members and the Denali Members shall cooperate with each other and use, and shall cause their respective Affiliates to use, their respective reasonable best efforts to prepare and file as promptly as reasonably practicable all documentation to effect all necessary notices, reports and other filings (including all applications required to be filed with the FCC and the RCA) and to obtain as promptly as reasonably practicable all consents, registrations, approvals, permits and authorizations necessary or advisable to be obtained from

any Third Party and/or any Governmental Entity in order to consummate the transactions contemplated by this Agreement (collectively, the "Regulatory Consents"). GCI will provide reasonable cooperation to the forgoing Parties to obtain such Regulatory Consents. None of GCI, any entity in the AKD Group,

32

Parent, the AKD Members or the Denali Members shall agree to participate in any substantive meeting or discussion with any such Governmental Entity in respect of any filing, investigation or inquiry concerning this Agreement or the transactions contemplated by this Agreement unless it consults with the other Parties reasonably in advance and, to the extent permitted by such Governmental Entity, gives the other Parties the opportunity to attend and participate.

(b) GCI, each entity in the AKD Group, Parent, the AKD Members and the Denali Members each shall, upon request by the other, furnish the other with all information concerning itself, its Subsidiaries, directors, officers, members and members and such other matters as may be reasonably necessary or advisable in connection with any statement, filing, notice or application made by or on behalf of any entity in the AKD Group or Denali to any Governmental Entity in connection with the transactions contemplated by this Agreement.

(c) Subject to applicable Law and the instructions of any Governmental Entity, GCI, each entity in the AKD Group, Parent, the AKD Members and the Denali Members each shall keep the others apprised of the status of matters relating to completion of the transactions contemplated hereby, including promptly furnishing the other with copies of notices or other communications received by such Person from any Governmental Entity with respect to such transactions.

- 6.3. Preparation for Closing. Each of the Parties will use commercially reasonable efforts to take all actions necessary, proper or advisable in order to consummate and make effective the transactions contemplated by this Agreement (including the satisfaction, but not the waiver, of the conditions precedent set forth in Section 7) and the other Transaction Agreements.
- 6.4. Notification of Certain Matters. Between the date of this Agreement and the Closing Date, each Party shall give prompt notice in writing to the other Parties of: (a) any information that indicates that any of such Party's representations or warranties contained herein was not true and correct in all material respects as of the date hereof or will not be true and correct in all material respects at and as of the Closing Date (except for changes permitted or contemplated by this Agreement), (b) the occurrence of any event that will result, or has a reasonable prospect of resulting, in the failure of any condition specified in Section 8 hereof to be satisfied, (c) any notice or other communication from any Person indicating that such Person will not or may not grant any consent or approval required in connection with the transactions contemplated by this Agreement or that such transactions otherwise may violate the rights of or confer remedies upon such Person and (d) any other material development that occurs after the date of this Agreement and affects the accuracy of the representations, warranties, covenants or Disclosure Schedule contained herein. No notice given under this Section 6.4 will be deemed to amend or supplement any Disclosure Schedule or to prevent or cure any misrepresentation, breach of warranty or breach of covenant of any Party.
- 6.5. Other Limitations on Conduct of AKD's and Properties' Businesses. AKD, Parent and the AKD Members hereby covenant and agree with GCI that, prior to the earlier of the termination of this Agreement pursuant to Section 11 or the Closing Date, unless the prior written consent of GCI shall

33

have been obtained and except as otherwise contemplated herein, they shall operate the business of each entity in the AKD Group only in the usual, regular and ordinary course of business consistent with historical practice, incurring only ordinary and necessary business expenses consistent with historical practice. Without limiting the foregoing, unless the prior written consent of GCI shall have been obtained and except as otherwise contemplated herein, prior to the earlier of the termination of this Agreement pursuant to Section 11 or the Closing Date AKD, Parent and the AKD Members shall: (a) use commercially reasonable efforts to preserve intact the business and assets of each entity in the AKD Group, including present operations, facilities, employee relationships and relationships with lessors, licensors, suppliers and customers; (b) comply with all applicable laws and regulations, and maintain the licenses, registrations, rights and franchises of each entity in the AKD Group; (c) not declare, set aside or pay any dividend or make any distribution with respect to the membership interests of any entity in the AKD Group or redeem, purchase, or otherwise acquire any such membership interests, except for the distributions contemplated by Section 2.2; (d) not enter into any material contracts, obligations, commitments or agreements (including without limitation any material modification or amendment to any such contract, commitment or agreement) other than in the ordinary course of business; (e) not sell, dispose of or otherwise transfer or encumber any assets, except in the ordinary course of business consistent with past practice; (f) not increase the compensation, benefits or any other form of remuneration to officers or employees, other than

in connection with promotions and performance reviews in accordance with past practice, and not award any bonuses or other extraordinary compensation to any officers or employees, except such as are paid in full in cash prior to the Closing; (g) not modify, alter or amend the terms or provisions of any AKD Plan; (h) maintain in full force and effect all insurance policies in effect on the date hereof or replacement policies providing comparable coverage if available on commercially reasonable terms; (i) not authorize for issuance, issue or obligate any entity in the AKD Group to issue any membership interests or any options, warrants, convertible securities or other rights to acquire any membership interests; (j) cause any entity in the AKD Group to pay and discharge all liabilities when due and not cause any entity in the AKD Group to accelerate the collections of receivables beyond their normal, stated terms; (k) not allow any entity in the AKD Group to hire additional management employees; and (1) not knowingly take any action that would (i) adversely affect the ability of any entity in the AKD Group to obtain any necessary approvals of any third parties or any governmental authorities required for the transactions contemplated hereby or materially increase the period of time necessary to obtain such approvals, or (ii) adversely affect any entity in the AKD Group's ability to perform its covenants and agreements under this Agreement or any other Transaction Agreement. Prior to Closing, AKD intends to acquire the interest of all Persons who are participants under that certain Profits Participation Plan and to terminate such Profit Participation Plan, and nothing contained in this Section 6.5 shall restrict the right of AKD to take such actions or require the consent of GCI prior to taking such actions.

6.6. Other Limitations on Conduct of Denali's Business. The Denali Members hereby covenant and agree with GCI that, prior to the earlier of the termination of this Agreement pursuant to Section 11 or the Closing Date, unless the prior written consent of GCI shall have been obtained and except as otherwise contemplated herein, they shall cause Denali to operate its business only in the usual, regular and ordinary course of business consistent with historical practice, incurring only ordinary and necessary business expenses consistent with historical practice. Without limiting the foregoing, unless the

34

prior written consent of GCI shall have been obtained and except as otherwise contemplated herein, prior to the earlier of the termination of this Agreement pursuant to Section 11 or the Closing Date the Denali Members shall cause Denali to: (a) use commercially reasonable efforts to preserve intact its business and assets, including present operations, facilities, employee relationships and relationships with lessors, licensors, suppliers and customers; (b) comply with all applicable laws and regulations, and maintain its licenses, registrations, rights and franchises; (c) not declare, set aside or pay any dividend or make any distribution with respect to its membership interests or redeem, purchase, or otherwise acquire any such membership interests; (d) not enter into any material contracts, obligations, commitments or agreements (including without limitation any material modification or amendment to any such contract, commitment or agreement) other than in the ordinary course of business; (e) not sell, dispose of or otherwise transfer or encumber any assets, except in the ordinary course of business consistent with past practice; (f) not increase the compensation, benefits or any other form of remuneration to its officers or employees, other than in connection with promotions and performance reviews in accordance with past practice, and not award any bonuses or other extraordinary compensation to any officers or employees, except such as are paid in full in cash prior to the Closing; (q) maintain in full force and effect all insurance policies in effect on the date hereof or replacement policies providing comparable coverage if available on commercially reasonable terms; (h) not authorize for issuance, issue or obligate itself to issue any membership interests or any options, warrants, convertible securities or other rights to acquire any membership interests; (i) pay and discharge all liabilities when due and not accelerate the collections of receivables beyond their normal, stated terms; and (j) not knowingly take any action that would (i) adversely affect its ability to obtain any necessary approvals of any third parties or any governmental authorities required for the transactions contemplated hereby or materially increase the period of time necessary to obtain such approvals, or (ii) adversely affect its ability to perform its covenants and agreements under this Agreement or any other Transaction Agreement.

6.7. Access to Information. Each entity in the AKD Group shall afford, and the Denali Members shall cause Denali to afford, GCI and its accountants, counsel and other representatives reasonable access during normal business hours prior to the Closing Date to (a) all of their financial statements, properties, books, contracts, commitments and records and (b) all other information concerning their business and assets as GCI may reasonably request. No information or knowledge obtained after the date hereof in any investigation pursuant to this Section 6.7 shall affect or be deemed to modify any representation or warranty contained herein or the conditions to the obligations of the Parties to consummate the transactions contemplated hereby.

6.8. Cooperation. Each of the Parties, upon the reasonable request from time to time of any other Party, shall take and cooperate with the other Parties in taking such actions as may be reasonably necessary or desirable to consummate the transactions contemplated hereby and to comply with the terms of this Agreement.

6.9. Announcements. Prior to the Closing, except as may be required by law or applicable stock exchange rules, no Party to this Agreement shall issue any press release or make any public announcement relating to the subject matter of this Agreement without the prior written approval of the other Parties, which approval will not be unreasonably withheld or delayed. If GCI believes that it is required by law or applicable stock exchange rules to make

# 35 **** CONFIDENTIAL TREATMENT

such a public announcement, it shall promptly advise the other Parties and use reasonable efforts, consistent with its legal obligations or obligations under stock exchange rules, to allow the other Parties an opportunity to review and comment upon the announcement before the announcement is made.

6.10. Use of Proceeds of Interim Capital Loans. All proceeds from the **** will be used by **** for **** to be made by **** by ****, which ****. **** will not make **** on the **** following the **** without ****.

## SECTION 7 CLOSING DELIVERIES

- 7.1. Of AKD, Parent and the AKD Members. At the Closing, AKD, Parent and the AKD Members shall make the following deliveries, unless waived by GCI:
- (a) Parent and each AKD Member shall deliver a counterpart of the Operating Agreement, duly executed by such Person and duly executed by Fire Lake.
- (b) AKD shall deliver to GCI an amended version of Schedule 1.3 and Schedule 4.1 of the Operating Agreement reflecting the AKD Common Units issued to GCI pursuant to Sections 2.3 and 2.4.
- (c) Parent and each AKD Member shall deliver instruments of assignment in form and substance reasonably satisfactory to GCI transferring to AKD the number of AKD Common Units to be redeemed by AKD from Parent or such AKD Member pursuant to Section 2.4, with full warranty of title, duly executed by such Person.
- (d) AKD shall deliver to Parent and the AKD Members cash or immediately available funds, by wire transfer as contemplated by Section 2.4
- (e) The AKD Members shall cause Fire Lake to deliver a counterpart of the Management Agreement, duly executed by Fire Lake.
- (f) AKD shall deliver evidence in form and substance reasonably satisfactory to GCI that the Management Agreement between AKD and Poplar Associates LLC has been terminated effective as of the Closing Date, without Liability or further obligation of AKD.
- $\mbox{\sc (g)}$  AKD shall deliver a counterpart of the Management Agreement, duly executed by AKD.
- (h) AKD shall deliver payoff letters in form and substance reasonably satisfactory to GCI with respect to any Liabilities of AKD that are to be paid off at the Closing with the GCI capital contributions made pursuant to Section 2.3.
- (i) AKD, Parent and the AKD Members shall deliver counterparts of any other Transaction Agreements to which AKD, Parent or the AKD Members are a party, duly executed by such Party.

36

- (j) AKD shall deliver an opinion of The Bogatin Law Firm, PLC, counsel to AKD, the AKD Members and the Denali Members, covering the matters set forth on the attached Exhibit C. Such opinion shall be dated as of the Closing Date and shall be in form and substance reasonably satisfactory to GCI and its counsel.
- (k) AKD shall deliver an opinion of Lukas, Nace, Gutierrez & Sachs, Chartered, counsel to AKD, the AKD Members and the Denali Members, covering the matters set forth on the attached Exhibit D. Such opinion shall be dated as of the Closing Date and shall be in form and substance reasonably satisfactory to GCI and its counsel.
- (1) AKD, Parent and each AKD Member shall deliver a certificate signed on behalf of AKD, Parent or such AKD Member by an authorized person, confirming that the conditions precedent to the obligations of GCI under Section 8.2.2 have been fulfilled, insofar as those conditions relate to the representations and warranties made by or to the obligations and covenants of AKD, Parent or such AKD Member.
- $\,$  (m) AKD shall deliver the certificate regarding wireless telephone subscribers contemplated by Section 3.5.4.

- (n) AKD shall deliver a Non-Competition Agreement executed by each of the persons specified in Schedule 7.1(n) in the form attached as Exhibit E.
- (o) AKD shall deliver such certificates or other documents as may be reasonably requested by GCI, including without limitation certificates of legal existence, good standing and certified charter documents from the Alaska Department of Commerce, Community, and Economic Development and certificates of officers or member AKD with respect to minutes, resolutions, bylaws and any other relevant matters concerning the authorization of the transactions contemplated hereby.
- 7.2. Of the Denali Members. At the Closing, each Denali Member shall make the following deliveries, unless waived by GCI:
- (a) Each Denali Member shall deliver instruments of assignment in form and substance reasonably satisfactory to GCI transferring to GCI such Denali Member's Denali Interest, with full warranty of title, duly executed by such Denali Member.
- (b) The Denali Members shall cause Denali to deliver instruments of assignment in form and substance reasonably satisfactory to GCI transferring to AKD the Denali Licenses, with full warranty of title, duly executed by Denali.
- (c) Each Denali Member shall deliver counterparts of any other Transaction Agreements to which such Denali Member is a party, duly executed by such Denali Member.
- (d) Each Denali Member shall deliver a certificate signed by such Denali Member on behalf of such Denali Member by an authorized person, confirming that the conditions precedent to the obligations of GCI under

37

- Section 8.2.3 have been fulfilled, insofar as those conditions relate to the representations and warranties made by or to the obligations and covenants of such Denali Member.
- (e) The Denali Members shall deliver such certificates or other documents as may be reasonably requested by GCI, including without limitation certificates of legal existence, good standing and certified charter documents from the Alaska Department of Commerce, Community, and Economic Development and certificates of officers or member Denali with respect to minutes, resolutions, bylaws and any other relevant matters concerning the authorization of the transactions contemplated hereby.
- 7.3. Of GCI. At the Closing, GCI shall make the following deliveries, unless waived by the AKD Members' Agent and the Denali Members' Agent:
- (a) GCI shall deliver to the Denali Members cash or immediately available funds, by wire transfer as contemplated by Section 2.1.
- (b) GCI shall deliver to AKD cash or immediately available funds, by wire transfer as contemplated by Section 2.3.
- (c) GCI shall deliver to AKD instruments of assignment in form and substance reasonably satisfactory to the AKD Members' Agent transferring to AKD the Denali Interest purchased by GCI pursuant to Section 2.1, with full warranty of title, duly executed by GCI.
- (d) GCI shall deliver to AKD cash or immediately available funds, by wire transfer as contemplated by Section 2.4.
- (e) GCI shall deliver counterparts of any other Transaction Agreements to which GCI a party, duly executed by GCI.
- (f) GCI shall deliver a certificate signed by GCI by an authorized officer, confirming that the conditions precedent to the obligations of AKD, the AKD Members and the Denali Members under Section 8.1.2 have been fulfilled.
- (g) GCI shall deliver such certificates or other documents as may be reasonably requested by the AKD Members' Agent or the Denali Members' Agent, including without limitation certificates of legal existence, good standing and certified charter documents from the Alaska Department of Commerce, Community, and Economic Development and certificates of officers of GCI with respect to minutes, resolutions, bylaws and any other relevant matters concerning the authorization of the transactions contemplated hereby.

SECTION 8
CONDITIONS PRECEDENT

and the Denali Members The obligations of AKD, Parent, the AKD Members and the Denali Members to consummate the transactions contemplated by this Agreement are subject to the satisfaction, prior to or contemporaneously with the Closing, of the following conditions, which may be waived in whole or in part by a written

3.8

instrument signed by the AKD Members' Agent and the Denali
Members' Agent:

- 8.1.1. Closing Deliveries by GCI. GCI shall have tendered delivery of all items required to be delivered by GCI under Section 7.
- 8.1.2. Representations; Covenants; Closing Certificates. The representations and warranties of GCI contained in Section 5, if qualified by reference to materiality, shall be correct and complete, and if not so qualified, shall be correct and complete in all material respects, as of the Closing Date. GCI shall have complied with all covenants required by this Agreement to be performed or complied with by it on or prior to the Closing Date.
- $\,$  8.1.3. General. All instruments and legal and corporate proceedings on the part of GCI in connection with the transactions contemplated by this Agreement and the other Transaction Agreements shall be reasonably satisfactory in form and substance to AKD, the AKD Members and the Denali Members.
- 8.2. Conditions to Obligations of GCI. The obligations of GCI to consummate the transactions contemplated by this Agreement shall be subject to the satisfaction, prior to or contemporaneously with the Closing, of the following conditions, which may be waived in whole or in part by GCI in writing:
- 8.2.1. Closing Deliveries by AKD, Parent, the AKD Members and the Denali Members. AKD, Parent, the AKD Members and the Denali Members shall have tendered delivery of all items required to be delivered by them under Section 7.
- 8.2.2. Representations and Covenants of AKD, Parent and the AKD Members. The representations and warranties of AKD, Parent and the AKD Members contained in Section 3, if qualified by reference to materiality, shall be correct and complete, and if not so qualified, shall be correct and complete in all material respects, as of the Closing Date. AKD, Parent and each AKD Member shall have complied with all covenants required by this Agreement to be performed or complied with by it on or prior to the Closing Date.
- 8.2.3. Representations and Covenants of the Denali Members. The representations and warranties of the Denali Members contained in Section 4, if qualified by reference to materiality, shall be correct and complete, and if not so qualified, shall be correct and complete in all material respects, as of the Closing Date. Each Denali Member shall have complied with all covenants required by this Agreement to be performed or complied with by it on or prior to the Closing Date.
- 8.2.4. General. All instruments and legal and limited liability company proceedings on the part of AKD, Parent, the AKD Members and the Denali Members in connection with the transactions contemplated by this Agreement and the other Transaction Agreements shall be reasonably satisfactory in form and substance to GCI.

39

- 8.3. Conditions to the Obligations of Each Party. The obligations of each of the Parties to consummate the transactions contemplated by this Agreement shall be subject to the following additional conditions:
- 8.3.1. Regulatory Consents. All Regulatory Consents shall have been granted without the imposition of any adverse condition on GCI, any entity in the AKD Group, Denali or any of their Affiliates and all such Regulatory Consents shall be in full force and effect as of the Closing Date.
- 8.3.2. Other Consents. Each entity in the AKD Group, Parent, the AKD Members and the Denali Members shall have given, made or obtained all notices, consents, approvals, orders, authorizations, declarations or filings to, of or with any Person required to be given, made or obtained by any such Party in connection with the execution, delivery and performance of, or consummation of the transactions contemplated by, any of the Transaction Agreements.
- 8.3.3. No Litigation. No temporary restraining order, preliminary or permanent injunction or other order by any federal or state court or governmental body prohibiting, preventing or materially restraining the consummation of the transactions contemplated by this Agreement shall have been issued and shall not have expired or been withdrawn or reversed and there shall be no pending or threatened litigation or other proceeding seeking to prohibit, prevent or materially restrict or impose any material limitations on the consummation of such transactions. The Parties hereto shall use their commercially reasonable efforts to cause any such temporary restraining order,

preliminary or permanent injunction or other order to be vacated or lifted as promptly as possible.

## SECTION 9 POST-CLOSING COVENANTS

9.1. Further Assurances. Each of the Parties, before, at and after the Closing Date, upon the reasonable request from time to time of any other Party and without further consideration, shall do each and every act and thing as may be necessary or reasonably desirable to fully and effectively consummate, evidence and confirm the transactions contemplated hereby, including: (a) executing, acknowledging and delivering assurances, assignments, powers of attorney and other documents and instruments; (b) furnishing information and copies of documents, books and records; (c) filing reports, returns, applications, filings and other documents and instruments with governmental authorities; (d) assisting in responding to any inquiry or claim of any Person with respect to a transaction occurring prior to the Closing; (e) assisting in good faith in any litigation, threatened litigation or claim and cooperating therein with other Parties and their advisors and representatives, including providing relevant documents and evidence and maintaining confidentiality in connection with such litigation or threatened litigation or claims, in each case, other than litigation, threatened litigation or claims in which the Party requesting such cooperation is adverse to the Party from whom such cooperation is requested; and (f) cooperating with the other Parties (at such other Parties' expense except as provided in this Agreement) in exercising any right or pursuing any claim, whether by litigation or otherwise, other than rights and claims running against the Party from which such cooperation is requested.

40

9.2. Announcements. Following the Closing, except as may be required by law, no Party to this Agreement shall issue any press release or make any public announcement relating to the subject matter of this Agreement without the prior written approval of the other Parties, which approval will not be unreasonably withheld or delayed. If GCI believes that it is required by law to make such a public announcement, it shall promptly advise the other Parties and use reasonable efforts, consistent with its legal obligations, to allow the other Parties an opportunity to review and comment upon the announcement before the announcement is made.

#### SECTION 10 INDEMNIFICATION

- 10.1. Indemnification by AKD, Parent and the AKD Members. Subject to the limitations set forth in this Section 10, AKD, Parent and the AKD Members hereby agree, jointly and severally, to indemnify and hold harmless GCI, AKD, any employee benefit plan of GCI or AKD and each of their respective officers, directors, employees and Affiliates (each, in such capacity, a "GCI Indemnitee"), from, against and in respect of any and all Adverse Consequences arising from, or otherwise related to, directly or indirectly, any of the following:
- (a) Any breach of any representation or warranty made by AKD, Parent or the AKD Members in this Agreement or any other Transaction Agreement (as each such representation or warranty would be read if all qualifications as to materiality or knowledge or words of similar import were deleted therefrom); and
- (b) Any breach or default in performance by AKD, Parent or any AKD Member of any covenant or other agreement in this Agreement or any other Transaction Agreement.
- 10.2. Indemnification by the Denali Members. Subject to the limitations set forth in this Section 10, the Denali Members hereby agree, jointly and severally, to indemnify and hold harmless each GCI Indemnitee from, against and in respect of any and all Adverse Consequences arising from, or otherwise related to, directly or indirectly, any of the following:
- (a) Any breach of any representation or warranty made by the Denali Members in this Agreement or any other Transaction Agreement (as each such representation or warranty would be read if all qualifications as to materiality or knowledge or words of similar import were deleted therefrom); and
- (b) Any breach or default in performance by any Denali Member of any covenant or other agreement in this Agreement or any other Transaction Agreement.
- 10.3. Indemnification by GCI. Subject to the limitations set forth in this Section 10, GCI hereby agrees to indemnify and hold harmless AKD, Parent, the AKD Members and the Denali Members, and each of their respective officers, directors, employees and Affiliates, from, against and in respect of any and all Adverse Consequences arising from, or otherwise related to, directly or indirectly, any of the following:
  - (a) Any breach of any representation or warranty made by

GCI in this Agreement or any other Transaction Agreement (as each such representation or warranty would be read if all qualifications as to materiality

## **** CONFIDENTIAL TREATMENT

or knowledge or words of similar import were deleted therefrom); and

(b) Any breach or default in performance by GCI of any covenant or other agreement in this Agreement or any other Transaction Agreement.

10.4. Survival; Time Limits for Indemnification. The representations and warranties made in this Agreement, or in any certificate or other document delivered pursuant to this Agreement or in connection with this Agreement, will survive the Closing Date as follows (even if the damaged party knew or had reason to know of any misrepresentation or breach of warranty at the time of Closing): (a) the representations and warranties contained in Sections 3.1, 3.2, 3.3.2, 3.9.2, 4.1, 4.2, 4.3 and 4.9.2 hereof shall not terminate and shall survive the Closing Date indefinitely; (b) the representations and warranties contained in Sections 3.4.4, 3.6, 3.7, 3.8, 4.4.4, 4.6, 4.7 and 4.8 hereof shall survive the Closing Date for a period of three years and shall terminate on the third anniversary of the Closing Date; and (c) all other representations and warranties in this Agreement or in any certificate or other document delivered pursuant to this Agreement or in connection with this Agreement survive the Closing Date for a period of two years and shall terminate on the second anniversary of the Closing Date. The covenants of the Parties made in this Agreement will survive the Closing Date indefinitely. Each Party shall promptly notify the others of any facts or other circumstances of which such Party becomes aware or has any knowledge that could give rise to a claim for indemnification under this Agreement by any Party. No Party will have any obligation to indemnify any Person pursuant to this Agreement with respect to any breach of a representation or warranty unless a notice of such breach is given to the Party against whom indemnification is sought on or prior to the last day of the applicable survival period, except that if a Party has a reasonable basis to believe that an indemnifiable claim will arise and gives notice to the other Party concerning such matter within the survival period, then all rights of such Party to seek indemnification with respect to such matter will survive the expiration of such period for a period of 180 days. If an indemnifiable claim has not arisen prior to the expiration of that 180-day period but the Party continues to have a reasonable basis to believe that an indemnifiable claim will arise and gives notice to such effect to the other party prior to the end of such 180-day period, then all rights of the Party to seek indemnification with respect to such matter will survive for one additional period of 180 days. If an indemnifiable claim does not arise prior to the end of the second 180-day period, the rights of the Party to seek indemnification will terminate at the expiration thereof. If a Party is obligated to indemnify another Party against a particular breach, the indemnity obligation shall extend to all Adverse Consequences, whether occurring before or after the survival period.

### 10.5. Basket and Cap.

10.5.1. Tipping Basket. AKD, Parent, the AKD Members and the Denali Members (as a group) will have no obligation to indemnify any GCI Indemnitee from and against any Adverse Consequences under Section 10.1 or 10.2 until the GCI Indemnitees (as a group) have suffered Adverse Consequences in the aggregate amount of \$**** or more arising from, or otherwise related to, directly or indirectly, any of the items set forth in Section 10.1 or 10.2. If and when the aggregate of such Adverse Consequences exceeds \$****, the GCI Indemnitees shall be entitled to indemnification against all Adverse Consequences incurred under Section 10.1 or 10.2, including the initial \$**** of Adverse Consequences.

# 42 **** CONFIDENTIAL TREATMENT

10.5.2. Indemnification Cap. The aggregate indemnification obligations of AKD, Parent, the AKD Members and the Denali Members under Sections 10.1 and 10.2 shall not exceed the sum of (x) \$**** plus (y) the aggregate purchase price for the AKD Common Units purchased by GCI from AKD pursuant to Section 2.4.

10.5.3. Exceptions. Notwithstanding anything to the contrary in this Agreement, there shall be no limitation on any GCI Indemnitee's right to indemnification from and against any Adverse Consequences arising directly or indirectly out of (a) any breach of any representation, warranty or covenant of AKD, Parent, an AKD Member or a Denali Member that involves an intentional misrepresentation or the commission of fraud by AKD, Parent or such AKD Member or Denali Member or (b) any act or omission or other matter arising prior to Closing that involves a claim by a third party for material misrepresentation, fraud, gross negligence or intentional misconduct, and each GCI Indemnitee shall have all remedies available to it at law and in equity with respect to any such breach, act, omission or matter.

10.6. Defense of Claims. The procedures to be followed with respect to the defense and settlement of any claim made by a Third Party which, if true, would give rise to a right on the part of a Party to be indemnified against resulting Adverse Consequences (an "Indemnitee"), in whole or in part, under this Section 10 (a "Claim") shall be as follows:

(a) Unless in the reasonable judgment of the Party seeking indemnification under this Section 10 (i) there is a conflict between the positions of the Party against whom indemnification is sought under this Section 10 (the "Indemnifying Party") and the Indemnitee in conducting the defense of such Claim or (ii) legitimate legal or business considerations would require the Indemnitee to defend or respond to such Claim in a manner different from the Indemnifying Party, the Indemnifying Party shall, by giving notice thereof to the Indemnitee confirming the Indemnifying Party's obligation under this Section 10 to indemnify the Indemnitee in respect of such Claim, be entitled to assume and control such defense with counsel chosen by it. The Indemnitee shall be entitled to participate therein after such assumption, but the costs of such participation (other than the costs of providing witnesses or documents at the request of the Indemnifying Party or in response to legal process) following such assumption shall be at the expense of the Indemnitee. Upon assuming such defense, the Indemnifying Party shall have full right to enter into any compromise or settlement which is dispositive of the matter involved; provided that, except for the settlement of a Claim that involves no obligation of the Indemnitee other than the payment of money for which full indemnification is provided hereunder, the Indemnifying Party shall not settle or compromise any Claim without the prior written consent of the Indemnitee, which consent will not be unreasonably withheld or delayed; and provided, further, that the Indemnifying Party may not consent to entry of any judgment or enter into any settlement in respect of a Claim which does not include an unconditional release of the Indemnitee from all liability in respect of such Claim.

(b) With respect to a Claim as to which the Indemnifying Party does not have the right to assume the defense under Section 10.6(a) or shall not have exercised its right to assume the defense, the Indemnitee shall assume and control the defense of such Claim with counsel chosen by it and the Indemnifying Party shall be obligated to pay all reasonable attorneys' fees and expenses of the Indemnitee incurred in connection with such defense. The Indemnifying Party shall be entitled to participate in the defense of such Claim

43

at its own expense. Notwithstanding the foregoing, the Indemnitee shall not be required to defend any Claim under this Section 10.6(b) unless (i) the Indemnifying Party confirms its obligation under this Section 10 to indemnify the Indemnitee in respect of such Claim by written notice to the Indemnitee and (ii) if requested by the Indemnitee, the Indemnifying Party provides reasonable assurance to the Indemnitee of the Indemnifying Party's financial ability to indemnify the Indemnitee against the costs of defense and any liability that may result from such Claim, including providing a bond or other security therefor if reasonably requested by the Indemnitee. If the Indemnitee is not required to defend any Claim under the immediately preceding sentence, it shall owe no duties to the Indemnifying Party with respect to such Claim, and may defend, fail to defend or settle such Claim without affecting its right to indemnity hereunder. If the Indemnitee is required to defend any such Claims under this Section 10.6(b), it shall not settle or compromise the Claim without the prior written consent of the Indemnifying Party, which consent will not be unreasonably withheld or delayed.

(c) If in the reasonable judgment of the Indemnitee it would be materially harmed or otherwise prejudiced by not entering into a proposed settlement or compromise as to which the Indemnifying Party withholds consent, the Indemnitee may enter into such settlement or compromise, but such settlement or compromise shall not be conclusive as to the existence or amount of the liability of the Indemnifying Party to the Indemnitee or any Third Party.

(d) Both the Indemnifying Party and the Indemnitee shall cooperate fully with one another in connection with the defense, compromise or settlement of any Claim, including without limitation making available to the other all pertinent information and witnesses within its control at reasonable intervals during normal business hours.

## SECTION 11

 $11.1.\ \mathrm{Right}$  to Terminate. The parties may terminate this Agreement as provided below:

(a) GCI, the AKD Members' Agent and the Denali Members' Agent may terminate this Agreement by mutual written consent at any time prior to the Closing.

(b) GCI may terminate this Agreement by giving written notice to the AKD Members' Agent and the Denali Members' Agent at any time prior to the Closing (i) in the event AKD, any AKD Member or any Denali Member has breached any representation, warranty or covenant contained in this Agreement in

a way that would result in the nonfulfillment of the conditions to the obligations of GCI hereunder, GCI has notified such Party of the breach, and the breach has not been cured within 10 days after the notice of breach or such longer period as agreed by the parties or (ii) if the Closing has not occurred on or before June 1, 2007 because of the failure of any condition precedent to the obligations of GCI to consummate the Closing (unless the failure results primarily from GCI breaching any representation, warranty or covenant contained in this Agreement).

44

(c) The AKD Members' Agent or the Denali Members' Agent may terminate this Agreement by giving written notice to GCI at any time prior to the Closing (i) in the event GCI has breached any representation, warranty or covenant contained in this Agreement in a way that would result in the nonfulfillment of the conditions to the obligations of AKD, the AKD Members and the Denali Members hereunder, the AKD Members' Agent or the Denali Members' Agent has notified GCI of the breach, and the breach has not been cured within 10 days after the notice of breach or such longer period as agreed by the parties or (ii) if the Closing has not occurred on or before June 1, 2007 because of the failure of any condition precedent to the obligations of AKD, Parent, the AKD Members or the Denali Members to consummate the Closing (unless the failure results primarily from AKD, any AKD Member or any Denali Member breaching any representation, warranty or covenant contained in this Agreement).

11.2. Effect of Termination. The termination of this Agreement by a party pursuant to Section 11.1(b) or (c) will in no way limit any obligation or liability of any other Party based on or arising from a breach or default by such other Party with respect to any of its representations, warranties, covenants or agreements contained in this Agreement prior to the termination, and the terminating party will be entitled to seek all relief to which it is entitled under applicable law.

11.3. Specific Performance. In view of (a) the complexities and uncertainties in measuring actual damages to be sustained by reason of the failure of a Party to perform this Agreement strictly in accordance with the specific terms hereof, and (b) the uniqueness of the transactions contemplated herein, each of the Parties acknowledges and agrees that: (i) the remedy at law for a breach of this Agreement would be inadequate, and (ii) the other Parties would be irreparably damaged if any Party fails to perform the provisions of this Agreement strictly in accordance with their specific terms. Therefore, it is expressly agreed that, in addition to any other remedy to which a nonbreaching Party may be entitled, at law or in equity, the nonbreaching Parties shall be entitled to injunctive relief to prevent breaches of the provisions of this Agreement and to specifically enforce the terms and provisions hereof in any action instituted in any court of the United States or any state thereof having subject matter jurisdiction thereof.

### SECTION 12 MISCELLANEOUS

12.1. Arbitration. The Parties will attempt in good faith to resolve any controversy or claim arising out of or relating to this Agreement or any other Transaction Agreement through discussions between the senior management of the Parties. If these discussions are unsuccessful, the Parties agree that any action asserting a claim by one Party against any other Party or Parties hereto (collectively, the "Disputing Parties") arising out of or relating to this Agreement or any other Transaction Agreement shall, on the written notice by one Disputing Party to the others, be submitted to binding arbitration to be held in Seattle, Washington. The arbitration shall be conducted by and in accordance with the Commercial Arbitration Rules of the American Arbitration Association. The Disputing Parties shall hold an initial meeting within thirty (30) days from receipt of notice from the requesting Party of a request for arbitration. Unless otherwise agreed in writing, they will

45

jointly appoint a mutually acceptable arbitrator not affiliated with any of the Disputing Parties. If they are unable to agree upon such appointment within thirty (30) days of the initial meeting, the Disputing Parties shall obtain an odd numbered list of not less than five (5) potential arbitrators from the Superior Court for the Third Judicial District, State of Alaska. Each Disputing Party shall alternatively strike a single name from the list until only one name remains, with such person to be the arbitrator. The Disputing Party requesting the arbitration shall strike the first name. Each Disputing Party shall pay an equal share of the costs related to the arbitration, unless the arbitrator's decision provides otherwise. Each Disputing Party shall bear its own costs to prepare for and participate in the arbitration. Each Disputing Party shall produce at the request of any other Disputing Party, at least thirty (30) days in advance of the hearing, all documents to be submitted at the hearing and such other documents as are relevant to the issues or likely to lead to relevant information. The arbitrator shall promptly render a written decision, in accordance with Alaska law and supported by substantial evidence in the record. The prevailing Party shall be entitled to recover reasonable attorneys' fees, costs, charges and expended or incurred therein, if the arbitrator's decision so provides. Failure to apply Alaska law, or entry of a decision that is not based on substantial evidence in the record, shall be additional grounds for modifying or vacating an arbitration decision. Judgment on any arbitration award shall be entered in any court of competent jurisdiction. In any subsequent arbitration, the decision in any prior arbitration of this Agreement shall not be deemed conclusive of the rights among the parties hereunder.

- 12.2. Governing Law. This Agreement shall be governed by and construed in accordance with the laws of the State of Alaska without giving effect to any choice or conflict of laws rule or provision that would cause the application of the domestic substantive laws of any other jurisdiction.
- 12.3. Notices. All notices and other communications required or permitted hereunder shall be in writing (including any facsimile transmission or similar writing), and may be given by any means selected by the sender. Each such notice or other communication shall be effective (i) if sent by facsimile to the recipient's fax number given below, when such facsimile is transmitted and the sender's facsimile machine confirms transmission, (ii) if sent by reputable overnight courier to the recipient's address given below, one Business Day after being delivered to such courier or (iii) if sent by any other means, when actually received.

To GCI:

General Communication, Inc. 2550 Denali Street, Suite 1000 Anchorage, AK 99503 Attention : Corporate Counsel Fax No. : (907) 868-5676

With a copy to:

General Communication, Inc. 2550 Denali Street, Suite 1000 Anchorage, AK 99503 Attention : General Manager & Executive Vice President Fax No. : (907) 868-5676

46

To AKD, Parent or any AKD Member:

Pacificom Holdings, LLC c/o Wireless Partners 5350 Poplar Avenue, Suite 875 Memphis, TN 38117 Attention: Stephen M. Roberts Fax No.: (901) 763-3369

With a copy to:

The Bogatin Law Firm, PLC 1661 International Place Drive, Suite 300 Memphis, Tennessee 38120-1431 Attention: Jack S. Magids Fax No. (901) 767-2803

To any Denali Member:

Pacificom Holdings, LLC c/o Wireless Partners 5350 Poplar Avenue, Suite 875 Memphis, TN 38117 Attention: Stephen M. Roberts Fax No.: (901) 763-3369

With a copy to:

The Bogatin Law Firm, PLC 1661 International Place Drive, Suite 300 Memphis, Tennessee 38120-1431 Attention: Jack S. Magids Fax No. (901) 767-2803

Any Party may change its address or facsimile number to be used for purposes of this Section 12.3 by notice to the other Parties; provided that no such change shall require any Party to give any notice to any AKD Member other than to the AKD Members' Agent or to any Denali Member other than to the Denali Members' Agent.

12.4. Entire Agreement, Assignability, Etc. This Agreement (including the Schedules and Exhibits attached hereto) (i) constitutes the entire agreement, and supersedes all other prior agreements and understandings, both written and oral, among the Parties, or any of them, with respect to the transactions and matters contemplated hereby, including the Amended and Restated Memorandum of Understanding dated January 26, 2006 among GCI, AKD and Denali, as

amended, and (ii) is not intended to confer upon any Person other than the Parties hereto any rights or remedies hereunder. No Party may assign its rights or delegate its duties under this Agreement without the consent of the other Parties, and any purported assignment or delegation without such consent shall be null and void.

- 12.5. Counterparts. This Agreement may be executed in any number of counterparts, no one of which need be signed by all Parties, each of which shall be deemed an original and all of which together shall constitute one and the same instrument. This Agreement may be executed or delivered by facsimile.
- 12.6. Representations as to Knowledge. Whenever any statement in this Agreement is made "to its knowledge" or words of similar intent or effect, (a) an individual will be deemed to have knowledge of a particular fact or other matter if (i) that individual is actually aware of that fact or matter, or (ii) a prudent individual could be expected to discover or otherwise become aware of that fact or matter in the course of conducting a reasonably comprehensive investigation regarding the accuracy of any representation or warranty contained in this Agreement; and (b) a Person (other than an individual) will be deemed to have knowledge of a particular fact or other matter if any individual who is serving, or who has at any time served, as a director, officer, general partner, registered agent, general manager, director of engineering, executor or trustee of that Person (or in any similar capacity) has, or at any time had, knowledge of that fact or other matter (as set forth in (i) and (ii) above), and any such individual (and any Party to this Agreement) will be deemed to have conducted a reasonably comprehensive investigation regarding the accuracy of the representations and warranties made herein by that Person or individual.
- 12.7. Headings, Terms. The section headings contained in this Agreement are inserted for convenience only and will not affect in any way the meaning or interpretation of this Agreement. Defined terms are applicable to both singular and plural forms. All pronouns will be deemed to refer to the masculine, feminine or neuter, as the identity of the Person may require. The singular or plural includes the other, as the context requires or permits. The word include (and any variation) is used in an illustrative sense rather than a limiting sense. The word day means a calendar day. All references to "Sections" are to sections of this Agreement unless indicated otherwise.
- $12.8.\ \text{Amendments}.$  This Agreement may be amended only by a written instrument signed by GCI, the AKD Members' Agent and the Denali Members' Agent.
- 12.9. Waivers. No waiver by any Party of any default, misrepresentation or breach of warranty or covenant hereunder, whether intentional or not, will be deemed to extend to any prior or subsequent default, misrepresentation or breach of warranty or covenant hereunder or affect in any way any rights arising by virtue of any prior or subsequent such occurrence, and no waiver will be effective unless set forth in writing and signed by the Party against whom such waiver is asserted.
- 12.10. Severability. The invalidity or unenforceability of any term or provision of this Agreement shall not affect the validity or enforceability of the remaining terms and provisions hereof.

### 48

- 12.11. Remedies Cumulative. Subject to Section 12.1, all remedies of the Parties under this Agreement or any other Transaction Agreement are cumulative with each other and with any other remedies available at law, in equity or by contract. Any decision to pursue one remedy shall not prevent a Party from pursuing any other remedy at the same or any subsequent time.
- 12.12. Expenses. Except as otherwise provided in this Agreement, each Party will bear all costs and expenses (including all legal, accounting ad tax related fees and expenses) incurred by it in connection with this Agreement and the transactions contemplated hereby.
- 12.13. Construction. The Parties have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement will be construed as if drafted jointly by the Parties and no presumption or burden of proof will arise favoring or disfavoring any Party by virtue of the authorship of any of the provisions of this Agreement. The Parties intend that each representation, warranty and covenant contained herein will have independent significance. If any Party breaches any representation, warranty or covenant contained herein in any respect, the fact that there exists another representation, warranty or covenant relating to the same subject matter (regardless of the relative levels of specificity) which the Party has not breached will not detract from or mitigate the fact that the Party is in breach of the first representation, warranty or covenant.
- 12.14. AKD Members' Agent and Denali Members' Agent. Pacificom shall act as the "AKD Members' Agent," and Pacificom shall act as the Denali Members' Agent under this Agreement. Parent and each AKD Member hereby authorizes and appoints the AKD Members' Agent, and each Denali Member hereby

authorizes and appoints the Denali Members' Agent, as his or its exclusive agent and attorney-in-fact to act on behalf of such Person with respect to all matters which are the subject of this Agreement, including, without limitation, (a) receiving or giving all notices, instructions, other communications, consents or agreements that may be necessary, required or given hereunder, (b) agreeing to any amendment of this Agreement or any other Transaction Agreement or waiving any right or remedy of Parent and the AKD Members or the Denali Members (as the case may be) hereunder or thereunder, and (c) asserting, settling, compromising, waiving or defending, or determining not to assert, settle, compromise or defend, (i) any claim which Parent or any AKD Member or Denali Member (as the case may be) may assert, or have the right to assert, against GCI, or (ii) any claim which GCI may assert, or have the right to assert, against Parent or any AKD Member or any Denali Member (as the case may be). Neither Parent nor any AKD Member shall act with respect to any of the matters which are the subject of this Agreement except through the AKD Members' Agent, and no Denali Member shall act with respect to any of the matters which are the subject of this Agreement except though the Denali Members' Agent.

12.15. Incorporation of Exhibits. The Exhibits and Schedules identified in this Agreement are incorporated herein by reference and made a part hereof.

12.16. No Tax Advice. Each Party hereby acknowledges that any tax advice express or implicit in the provisions of this Agreement is not intended or written to be used, and cannot be used, by any taxpayer for the purpose of avoiding penalties that may be imposed on any taxpayer by the

49

Internal Revenue Service. Each Party should seek advice based on its particular circumstances from an independent tax advisor.

50

IN WITNESS WHEREOF, the Parties have duly executed this Reorganization Agreement as of the date first above written.

GENERAL COMMUNICATION, INC.

By: /s/ William C. Behnke Name: William C. Behnke Title: Senior Vice President

ALASKA DIGITEL, LLC

By: /s/ William M Yardell, III Name: William M Yardell, III Title: President

AKD HOLDINGS, LLC

By: /s/ William M Yardell, III Name: William M Yardell, III Title: President

PACIFICOM HOLDINGS, LLC

By: /s/ William M Yardell, III Name: William M Yardell, III Title: Chief Manager

RED RIVER WIRELESS, LLC

By: /s/ William M Yardell, III Name: William M Yardell, III Title: Chief Manager

GRAYSTONE HOLDINGS, LLC By: /s/ J. Michael Keenan Name: J. Michael Keenan Title: Exec. V.P.

51

Exhibit A

[This exhibit does not contain information material to an investment decision that is not otherwise disclosed in the agreement and is not included in the filing of the agreement with the SEC as being not material to investor decisions.]

#### ALASKA DIGITEL, LLC

This Second Amended and Restated Operating Agreement is made as of the [] day of [], 2006 (the "Effective Date"), by and among General Communication, Inc., an Alaska corporation ("GCI"), Pacificom Holdings, L.L.C., a Delaware limited liability company ("Pacificom"), Red River Wireless, LLC, a Delaware limited liability company ("Red River"), Graystone Holdings, LLC, an Alaska limited liability company ("Graystone"), AKD Holdings, LLC, a Delaware limited liability company ("AKD Holdings") and Fire Lake Partners, L.L.C., an Alaska limited liability company ("Fire Lake"), such parties being all of the Members of Alaska Digitel, LLC, an Alaska limited liability company (the "Company"), on the Effective Date.

In consideration of the mutual covenants contained in this Agreement and intending to be legally bound, the parties agree as follows:

#### ARTICLE 1. FORMATION AND DEFINITIONS

- 1.1 Formation. The Company was formed on September 17, 1996 by filing Articles of Organization with the Alaska Department of Commerce, Community, and Economic Development under the Act.
- 1.2 Name. The name of the Company is Alaska Digitel, LLC. The business of the Company will be conducted under such name, and any other name or names as the Company may from time to time determine.
- 1.3 Members and Units. The name and address of each Member and its number of Units held as of the Effective Date are set forth in Schedule 1.3.
  - 1.4 Office and Agent.
  - [a] The registered office of the Company in Alaska is 508 West 2nd Avenue, Third Floor, Anchorage, Alaska 99501 and its registered agent is John H. Tindall. The Company may change its registered office or registered agent in Alaska in accordance with the Act.
  - [b] The principal office of the Company is at 3127 Commercial Drive, Anchorage, Alaska 99501. The Company may change its principal office upon the affirmative Vote of Members owning more than 50% of the outstanding Units.
- 1.5 Foreign Qualification. The Company will apply for a certificate of authority to do business in any other state or jurisdiction, as required or
- appropriate from time to time, and will file such other certificates and instruments as may be required or appropriate from time to time in connection with its formation, existence and operation.
- 1.6 Term. The Company became effective as a limited liability company on the date its Articles of Organization were filed with the Alaska Department of Commerce, Community, and Economic Development and will continue in effect, unless and until a Dissolution occurs and Articles of Dissolution are filed in accordance with the Act.
- 1.7 Effective Date; Amendment and Restatement. This Agreement will become effective on the Effective Date and as of such date, amends and restates in its entirety the Amended and Restated Operating Agreement of the Company dated March 31, 2004, as amended by amendments dated April 7, 2004, January 9, 2006 and [], 2006 (as amended, the "Former Operating Agreement"), which shall be null, void and of no further force or effect.
- 1.8 Definitions. The following terms used in this  $\,$  Agreement have the corresponding meanings set forth below:
  - Act: the Alaska Revised Limited Liability Company Act, as amended from time to time.
  - Additional Contribution: a capital contribution (other than the Initial Contribution) that a Member makes to the Company with respect to any Units issued to such Member, as described in Section 4.2.
  - Adjusted Capital Account Deficit: as to any Member, the deficit balance (if any) in such Member's Capital Account as of the end of the Fiscal Year, after [a] crediting to such Capital Account any amount that such Member is obligated to restore pursuant to this Agreement or is deemed obligated to restore pursuant to the minimum gain chargeback provisions of the ss. 704(b) Regulations and [b] charging to such Capital Account any adjustments, allocations or distributions described in the qualified income offset provisions of the ss. 704(b) Regulations that are required to be charged to such Capital Account pursuant to this Agreement.

Adjusted Initial Capital Account: as defined in Section 14.1[a].

Affiliate: with respect to any Person, any Person that directly or indirectly Controls, is Controlled by, or is under common Control with, such Person. Notwithstanding the foregoing, for purposes of this Agreement (except for purposes of the definition of Third Party), neither the Company nor any of its direct or indirect

2

subsidiaries will be deemed to be an Affiliate of any Member.

Agreement: this Operating Agreement, also known as a limited liability company agreement, as amended from time to time.

AKD Call Period: as defined in Section 14.1[d].

AKD Holdings: as defined in the preamble.

AKD Holdings Contribution Agreement: as defined in Section 4.1.

AKD Holdings Member: AKD Holdings and any successor to or Transferee of Units from AKD Holdings who is admitted as a Member pursuant to Article 13. If at any time there is more than one AKD Holdings Member, any allocations and Distributions to the AKD Holdings Member under this Agreement or any other rights or obligations of the AKD Holdings Member under this Agreement will be allocated among such AKD Holdings Members based upon the number of Units owned by each AKD Holdings Member as a percentage of the total number of Units owned by all AKD Holdings Members.

AKD Net Asset Value as defined in the Reorganization Agreement.

AKD Redemption Price: as defined in the Reorganization Agreement.

Annual Budget: shall mean, as at any time, the Company's then-effective annual operating and capital budget approved or otherwise in effect pursuant to Section 7.5.

Appraised Unit Value as defined in Section 14.3.

Articles: the Articles of Organization referred to in Section 1.1, as amended from time to time.

3

Authorized Valuation Methodology: as defined in Section 14.3[a].

Available Cash: as of any relevant date, the aggregate cash or cash equivalents of the Company on hand or on deposit with banks or financial institutions or held in any other form by the Company, less [i] the current portion of indebtedness of the Company, [ii] payments required to be paid by the Company within one year after the date of calculation, and [iii] reasonable reserves for working capital and contingent liabilities of the Company as reasonably determined by the Board of Managers.

Bankruptcy: a Person will be deemed bankrupt if: [a] any proceeding is commenced against such Person as "debtor" for any relief under bankruptcy or insolvency laws, or laws relating to the relief of debtors, reorganizations, civil rehabilitations, arrangements, compositions, or extensions, or such Person becomes subject to procedures for provisional or final attachment in respect of all or a material portion of its assets, and [i] such proceeding is not dismissed or stayed within 120 days after such proceeding has commenced, or [ii] an order for relief against such Person is granted, or [b] such Person commences any proceeding for relief under bankruptcy or insolvency laws or laws relating to the relief of debtors, reorganizations, civil rehabilitations, arrangements, compositions, or extensions.

Board of Managers: the governing body of the Company that has the power and authority set forth in Section 7.1, comprised of all of the Managers, as and when acting in their capacity as the Board of Managers of the Company as provided in this Agreement.

Budget Committee: as defined in Section 7.5.

Business Day: any day other than a Saturday, a Sunday or a day on which banking institutions in Anchorage, Alaska are required or authorized to be closed.

Call Notice: as defined in Section 14.1[a].

Capital Account: the capital account of a Member established and

maintained in accordance with Section 4.3.

- Capital Contribution: any contribution (or deemed contribution) of money or property by a Member to the Company that is either an Initial Contribution or an Additional Contribution.
- Change of Control: with respect to any subject Person means the first to occur of the following events: [i] any sale, lease, exchange, or other transfer (in one or a series of related transactions) of all or substantially all of the assets of the subject Person to any Person or group of related Persons as determined pursuant to Section 13(d) of the Securities Exchange Act of 1934, other than to one or more of its Affiliates; or [ii] any event pursuant to which any Person other than any Affiliates of the subject Person acquires beneficial ownership (within the meaning of Rule 13d-3 promulgated under the Securities Exchange Act of 1934) of more than 50% of the combined voting power of the then-outstanding voting securities of the subject Person; provided that no Change of Control with respect to the Company shall occur as a result of any of the transactions contemplated by the Reorganization Agreement.
- Code: The Internal Revenue Code of 1986, as amended from time to time, and the corresponding provisions of any subsequent revenue laws.
- Commercially Reasonable Efforts: reasonable efforts made by any party that will not require such party to undertake extraordinary or unreasonable measures to obtain any consents, approvals or other authorizations or to achieve other desired results, including requiring such party to make any material expenditures (other than normal filing fees or the like) or to accept any material changes in the terms of a contract, license or other instrument for which a consent, approval or authorization is sought.
- Common Units: Units denominated as such and having the rights set forth in Section 3.2.
- Company: as defined in the preamble.
- Confidential Information: as defined in Section 15.3[b].

5

Control: including with correlative meanings, the terms "controlling",
 "controlled by" and "under common control with", as used with
 respect to any Person means the possession, directly or
 indirectly, of the power to direct or cause the direction of the
 management and policies of a Person, whether through the
 ownership of voting securities, by contract or otherwise.

Damages: as defined in Section 9.1.

- Dissolution: the happening of any of the events described in Section  $11.1. \ \ \,$
- Distribution: the amount of any money or the Fair Market Value of any property distributed by the Company to a Member as an operating or liquidating Distribution in accordance with this Agreement, reduced by the amount of any Company liabilities assumed by the distributee or to which the distributed property is subject.
- EBITDA Multiplier: as defined in the Management Agreement and as adjusted from time to time pursuant to Section 5.1(d) of the Management Agreement.

Effective Date: as defined in the preamble.

Fair Market Value: the cash price at which a willing seller would sell and a willing buyer would buy, both having full knowledge of the relevant facts and being under no compulsion to buy or sell, in an arm's-length transaction without time constraints. The Fair Market Value any asset will be the fair market value thereof as determined in good faith by the Board of Managers or, if requested by Notice given by any Member, as determined in accordance with the valuation procedure set forth in Section 14.3, which determination will be conclusive and binding; provided that any determination of the Fair Market Value of Units for purposes of determining the Appraised Unit Value of such Units will be governed in all respects by the provisions of Article 14.

15% Coupon: as defined in Section 14.1[a].

Fire Lake: as defined in the preamble.

Fire Lake Member: Fire Lake and any successor to or Transferee of Units from Fire Lake who is admitted as a Member pursuant to Article 13. If at any time there is more than one Fire Lake Member, any allocations and Distributions to the Fire Lake Member under this Agreement or any other rights or obligations of the Fire Lake Member under this Agreement will be allocated among such Fire Lake Members based upon the number of Units owned by each Fire Lake Member as a percentage of the total number of Units owned by all Fire Lake Members.

First GCI Call Period: as defined in Section 14.1[a].

Fiscal Year: the fiscal and taxable year of the Company, including both 12-month and short fiscal or taxable years; until changed as provided in this Agreement, each Fiscal Year will begin on January 1 of each year and end on December 31 of such year, provided that the first Fiscal Year will begin on the Effective Date and the last Fiscal Year will end on the date on which Liquidation of the Company is completed.

Former Operating Agreement: as defined in Section 1.7.

GAAP: generally accepted accounting principles as in effect from time to time in the United States of America, consistently applied.

GCI: as defined in the preamble.

GCI Member: GCI and any successor to or Transferee of Units from GCI who is admitted as a Member pursuant to Article 13. If at any time there is more than one GCI Member, any allocations and Distributions to the GCI Member under this Agreement or any other rights or obligations of the GCI Member under this Agreement will be allocated among such GCI Members based upon the number of Units owned by each GCI Member as a percentage of the total number of Units owned by all GCI Members.

GCI Requested Financial Information: as defined in Section 10.5.

7

Governmental Approvals: any consent, approval or authorization of, notice to, declaration of, or filing with, any Governmental Authority.

Governmental Authority: any foreign, domestic, federal, territorial, state or local governmental authority, quasi-governmental authority, court, government or self-regulatory organization, commission, tribunal, organization or any regulatory, administrative or other agency, or any political or other subdivision, department or branch of any of the foregoing.

Graystone: as defined in the preamble.

Graystone Member: Graystone and any successor to or Transferee of Units from Graystone who is admitted as a Member pursuant to Article 13. If at any time there is more than one Graystone Member, any allocations and Distributions to the Graystone Member under this Agreement or any other rights or obligations of the Graystone Member under this Agreement will be allocated among such Graystone Members based upon the number of Units owned by each Graystone Member as a percentage of the total number of Units owned by all Graystone Members.

Indemnified Persons: as defined in Section 9.1.

Initial Contribution: the initial Capital Contribution made (or deemed made) by a Member to the Company with respect to any Units issued to such Member, as described in Schedule 4.1.

IRS Notice: as defined in Section 3.4[b].

Lien: a mortgage, lien, pledge, collateral assignment, charge, title retention agreement, levy, execution, attachment, garnishment, security interest or other encumbrance.

Limited Owner: as defined in Section 11.3.

Liquidation: the process of winding up and terminating the Company after its Dissolution.

Losses: as defined in Section 5.1.

8

Management Agreement: The Management Agreement entered into by and between the Company and Fire Lake as of the date hereof in the

form attached hereto as Exhibit C.

- Manager: an individual appointed to serve on the Board of Managers in accordance with Section 7.2.
- Members: each of the GCI Member, the Pacificom Member, the Red River Member, the Graystone Member, the Fire Lake Member and any other Person admitted as a Member pursuant to Article 13.
- Non-GCI Members: each of the AKD Holdings Member, the Pacificom Member, the Red River Member and the Graystone Member.
- Notice: any written notice actually delivered pursuant to Section 16.12 or deemed delivered pursuant to Section 16.13.
- Original Profits Interest Units: as defined in Section 3.3[a].
- Pacificom: as defined in the preamble.
- Pacificom Member: Pacificom and any successor to or Transferee of Units from Pacificom who is admitted as a Member pursuant to Article 13. If at any time there is more than one Pacificom Member, any allocations and Distributions to the Pacificom Member under this Agreement or any other rights or obligations of the Pacificom Member under this Agreement will be allocated among such Pacificom Members based upon the number of Units owned by each Pacificom Member as a percentage of the total number of Units owned by all Pacificom Members.
- Person: an individual, corporation, partnership, limited liability company, trust, unincorporated organization or other entity.
- Presumed Tax Liability: for any Member for a Fiscal Year and any Fiscal Year prior thereto, an amount equal to the product of [a] the amount of taxable income allocated to such Member for that Fiscal

9

Year and all Fiscal Years prior thereto and [b] the Presumed Tax Rate.

- Presumed Tax Rate: the highest combined federal and state income tax rate applicable during such Fiscal Year to a natural person residing in Alaska, taxable at the highest marginal federal income tax rate and the highest marginal Alaska income tax rates, determined without regard to the adjustments provided for in Sections 67 and 68 of the Code.
- Proceeding: any threatened, pending, ongoing or completed action, suit or proceeding, whether formal or informal, and whether civil, administrative, investigative or criminal.
- Profits: as defined in Section 5.1.
- Profits Interest Units: Units denominated as such and having the rights set forth in Section 3.3.
- Put Notice: as defined in Section 14.2[a].
- Qualified Appraiser as defined in Section 14.3[a].
- Red River: as defined in the preamble.
- Red River Member: Red River and any successor to or Transferee of Units from Red River who is admitted as a Member pursuant to Article 13. If at any time there is more than one Red River Member, any allocations and Distributions to the Red River Member under this Agreement or any other rights or obligations of the Red River Member under this Agreement will be allocated among such Red River Members based upon the number of Units owned by each Red River Member as a percentage of the total number of Units owned by all Red River Members.
- Regulations: the Treasury Regulations (including temporary or proposed regulations) promulgated under the Code, as amended from time to time (including corresponding provisions of succeeding regulations).
- Regulatory Allocations: as defined in Section 5.12.

10

Reorganization Agreement: the Reorganization Agreement dated as of June  $[\ ]$ , 2006 among the initial Members, the Company, and the members of Denali PCS, LLC.

Representatives: as defined in Section 15.3[a].

- Safe Harbor Units: as defined in Section 3.4[b].
- Second GCI Call Period: as defined in Section 14.1[b].
- Tax Distribution: as defined in Section 6.5.
- Tax Matters Partner: as defined in Section 10.9.
- Third Party: means, with respect to any Person, any other Person that is not an Affiliate of such Person.
- Transfer: a direct or indirect sale, exchange, assignment, transfer or other disposition of a Unit or any interest therein (including the creation of a Lien on all or any part of a Unit), whether voluntary, involuntary or by operation of law.
- Transferee: a Person to whom a Unit is Transferred in compliance with this Agreement.
- Transferor: a Person who Transfers a Unit in compliance with this  $\mbox{\sc Agreement.}$
- Unit: as defined in Section 3.1.
- Vote: the action of the Members made in accordance with the voting requirements set forth in Article 7 or any other applicable provision of this Agreement, either at a meeting or by written consent without a meeting.
- Withdrawal: the occurrence of an event which terminates membership in the Company, as provided in Section 11.2.
- ARTICLE 2. PURPOSES AND POWERS
- 2.1 Principal Purposes. Subject to the provisions of this Agreement, the purpose of the Company is [a] to provide wireless communication services; and

## 11 **** CONFIDENTIAL TREATMENT

- [b] to do any and all other acts or things that may be incidental, advisable or necessary to carry on the business of the Company as contemplated by this Agreement.
- 2.2 Other Purposes. The Company may engage in activities related or incidental to its principal purpose, as well as any other lawful business or investment activity as may be approved by the unanimous Vote of the Members.
- 2.3 Powers. Subject to the provisions of this Agreement, the Company has all the powers granted to a limited liability company under the Act, as well as all powers necessary or convenient to or for the furtherance of the purpose specified in Section 2.1 that are not expressly prohibited to the Company by applicable law.

## ARTICLE 3. UNITS

3.1 Classes of Units. The ownership of the Company shall be divided into and represented by Units. There shall be two classes of Units: Common Units and Profits Interest Units. References in this Agreement to "Units" shall include all Units outstanding as of the relevant date, without regard to class. The Company shall be authorized to issue a total of 3,830 Units, consisting of **** Common Units and **** Profits Interest Units, subject to the adjustments provided for in Sections 3.2 and 3.3 below. The Units owned by each Member are set forth in Schedule 1.3 (subject to the adjustments contemplated by Section 3.2 and 3.3 below). Each time Units are issued, cancelled, forfeited or transferred in accordance with the terms and conditions of this Agreement, the Company shall attach a revised Schedule 1.3 to this Agreement and send a copy thereof to all Members. No consent of any Member shall be needed to make such revisions to Schedule 1.3.

### 3.2 Common Units.

[a] Issuance. The Company has issued on the Effective Date **** Common Units to the GCI Member, **** Common Units to AKD Holdings, **** Common Units to the Pacificom Member, **** Common Units to the Red River Member and **** Common Units to the Graystone Member for the Capital Contributions set forth in Section 4.1; provided that Section 2.3 of the Reorganization Agreement provides for an adjustment to the number of Common Units issued on the Effective Date to the GCI Member based on the value of the Company as of the Effective Date as determined in accordance with the terms and conditions of the Reorganization Agreement (such value is referred to in the Reorganization Agreement as the AKD Net Asset Value); and provided further that Section 2.4 of the Reorganization Agreement provides that at the closing of the transactions contemplated by the Reorganization

Agreement, GCI shall purchase from the Company an additional **** Common Units, with the proceeds from the sale of such additional Common Units to be used by the Company to redeem a like number of Common Units first from the Pacificom Member, the Red River Member and the Graystone Member and then from the AKD Holdings Member. The number of Common Units held by the Non-GCI Members and the GCI Member following the transactions and adjustments contemplated by Sections 2.2, 2.3 and 2.4 of the Reorganization Agreement shall automatically be adjusted as appropriate to reflect that total number of Common Units finally determined to be held by the Non-GCI

## 12 **** CONFIDENTIAL TREATMENT

Members and the GCI Member pursuant to the terms and conditions of Sections 2.2, 2.3 and 2.4 of the Reorganization Agreement. The adjustments to the number of Common Units issued to the GCI Member based on the AKD Net Asset Value pursuant to Section 2.3 of the Reorganization Agreement shall be made to be effective retroactively as of the date of the original issuance. The Company shall make appropriate revisions to Schedule 1.3 to reflect the number of Common Units held by the Non-GCI Members or the GCI Member following the transactions contemplated by Sections 2.2, 2.3 and 2.4 of the Reorganization Agreement, which revisions do not need to be approved by the Members. Exhibit D provides some examples of the operation of the Sections 2.2, 2.3 and 2.4 of the Reorganization Agreement and the accompanying adjustments to the total number of Common Units held by each Member. Except as set forth above, the Company shall have no authority to issue additional Common Units.

[b] Rights. The holders of the Common Units shall have the voting and economic rights set forth in the other Articles of this Agreement.

### 3.3 Profits Interest Units.

- [a] Issuance. The Company has issued on the Effective Date a total of **** Profits Interest Units to Fire Lake (the "Original Profits Interest Units"); provided that if the transactions and adjustments contemplated by Sections 2.2, 2.3 and 2.4 of the Reorganization Agreement result in the Company having more or less than **** Common Units issued and outstanding, then the number of the Original Profits Interest Units shall be automatically adjusted to equal a number of Profits Interest Units equal to [i] the total adjusted number of Common Units issued and outstanding divided by **** (rounded to two decimal places), minus [ii] the total adjusted number of Common Units issued and outstanding. Exhibit D provides some examples of the operation of the Sections 2.2, 2.3 and 2.4 of the Reorganization Agreement and the accompanying adjustments to the number of Original Profits Interest Units. Except for the adjustment contemplated by Section 3.3[c] below, the Company shall have no authority to grant additional Profits Interest Units.
- [b] Rights. Profits Interest Units shall constitute an interest in the future Profits and Losses of the Company and shall not entitle the holder thereof to any portion of the value of the Company as of the date that the Profits Interest Units are issued. The holders of Profits Interest Units shall have the voting and economic rights set forth in the other Articles of this Agreement. It is the intent of the Members and the Company that Profits Interest Units shall represent an interest in the Company that qualifies as a "profits interest" within the meaning of IRS Revenue Procedure 93-27, 1993-2 C.B. 343, or any successor authority thereto, and such profits interest shall have no Capital Account as of the date that any such Profits Interest Units are issued.
- [c] Adjustment of Profits Interest Units. In the event that the Management Agreement terminates for any reason, GCI exercises its rights to purchase the Units of the Non-GCI Members and the Fire Lake Member pursuant to Section 14.1 below or if there is a Change of Control transaction with

1.3

respect to the Company, then subject to any forfeiture provisions contained in Section 14.1 below, the number of Profits Interest Units issued to Fire Lake pursuant to Section 3.3[a] shall be deemed to have been retroactively adjusted as of the date of the original issuance to equal the amount of the Original Profits Interest Units multiplied by the applicable EBITDA Multiplier. By way of example, assuming that the number of Original Profits Interest Units remains at 230 pursuant to Section 3.3[a], then if there is an adjusting event and the EBITDA Multiplier is equal to 1.5, then Fire Lake will be deemed to have owned 345 Profits Interest Units from the Effective Date. The Company will then make appropriate adjustments to the Capital Accounts of the Members to reflect this adjustment, which adjustments will not need to be approved by the Members. The Company shall also make appropriate revisions to Schedule 1.3, which revisions will not need to be approved by the Members.

- 3.4 Code Section 83(b) Election and Safe Harbor.
- [a] General. Each Member that acquires an interest in the Company that is

subject to vesting agrees that within 30 days after such Person becomes a Member, the Person who is performing the services to which the vesting requirement relates shall file an election with the Internal Revenue Service under Section 83(b) of the Code and the regulations promulgated thereunder with respect to that interest (and with respect to any other interest in the Company that is subject to vesting based upon services to be performed by that Person).

- [b] Safe Harbor Election. By executing this Agreement, each Member authorizes and directs the Company to elect to have the "Safe Harbor" described in the proposed Revenue Procedure set forth in Internal Revenue Service Notice 2005-43 (the "IRS Notice") apply to any Units transferred to a service provider by the Company on or after the effective date of such Revenue Procedure (or any substantially similar Revenue Procedure or other guidance issued by the Internal Revenue Service) in connection with services provided to the Company. For purposes of making such Safe Harbor election, the Board of Managers is hereby designated as the "partner who has responsibility for federal income tax reporting" by the Company and, accordingly, that execution of such Safe Harbor election by the Board of Managers constitutes execution of a "Safe Harbor Election" in accordance with Section 3.03(1) of the IRS Notice with respect to such Units ("Safe Harbor Units"). The Company and each Member hereby agree to comply with all requirements of the Safe Harbor described in the IRS Notice (and any substantially similar Revenue Procedure or other guidance issued by the Internal Revenue Service), including, without limitation, the requirement that each Member shall prepare and file all federal income tax returns reporting the income tax effects of each Safe Harbor Units issued by the Company in a manner consistent with the requirements of the IRS Notice (an any substantially similar Revenue Procedure or other guidance issued by the Internal Revenue Service).
- [c] Failure to Comply. Any Member or former Member that fails to comply with the requirements set forth in Section 3.4[a] or Section 3.4[b] shall indemnify and hold harmless the Company and each adversely affected Member and former Member from and against any and all losses, liabilities, taxes, damages, judgments, fines, costs, penalties, amounts paid in settlement and

14

reasonable out-of-pocket costs and expenses incurred in connection therewith (including, without limitation, costs and expenses of suits and proceedings, and reasonable fees and disbursements of counsel), in each case resulting from such Member's or former Member's failure to comply with such requirements. The Board of Managers may offset distributions to which a Person is otherwise entitled under this Agreement against such Person's obligation to indemnify the Company and any other Person under this Section 3.4[c] (and any amount so offset with respect to such Person's obligation to indemnify a Person other than the Company shall be paid over to such other Person by the Company). A Member's obligations to comply with the requirements of Section 3.4[a] or Section 3.4[b] and to indemnify the Company and any Member or former Member under this Section 3.4[c] shall survive such Member's ceasing to be a Member of the Company and/or the termination, dissolution, liquidation and winding up of the Company, and, for purposes of this Section 3.4[c], the Company shall be treated as continuing in existence. The Company and any Member or former Member may pursue and enforce all rights and remedies it may have against each Member or former Member under this Section 3.4[c], including [i] instituting a lawsuit to collect such indemnification and contribution, with interest calculated at a rate equal to the prime rate plus three percentage points per annum (but not in excess of the highest rate per annum permitted by law), compounded on the last day of each fiscal quarter and [ii] specific performance and/or immediate injunctive or other equitable relief from any court of competent jurisdiction (without the necessity of showing actual money damages, or posting any bond or other security) in order to enforce or prevent any violation of the provisions of Section 3.4[a] or Section 3.4[b].

- [d] Certain Amendments. Each Member authorizes the Board of Managers to amend Section 3.4[a] and Section 3.4[b] to the extent necessary to achieve substantially the same tax treatment with respect to any interest in the Company transferred by the Company to a service provider in connection with services provided to the Company as set forth in Section 4 of the IRS Notice (e.g., to reflect changes from the rules set forth in the IRS Notice in subsequent Internal Revenue Service guidance), provided that such amendment is not materially adverse to any Member (as compared with the after-tax consequences that would result if the provisions of the IRS Notice applied to all interests in the Company transferred to a service provider by the Company in connection with services provided to the Company).
- 3.5 Reacquired Units. Any Units reacquired by the Company shall automatically be cancelled and shall not be deemed issued or outstanding. Units reacquired by the Company shall not be available for reissuance.
- 3.6 Certificates. The Company may issue certificates representing any or all of the outstanding Units, in the discretion of the Board of Managers.

4.1 Initial Contributions. By execution of this Agreement: [a] AKD Holdings acknowledges and agrees that all of the Class A Membership Interest and Class B Membership Interest and accompanying Class A Points, Class B Points and Voting Points (as such terms are defined in the Former Operating Agreement) granted to Pacificom, Red River and Graystone pursuant to the Former Operating

## 15 **** CONFIDENTIAL TREATMENT

Agreement and transferred to AKD Holdings pursuant to that certain Contribution Agreement dated as of the date hereof by and among the Non-GCI Members (the "AKD Holdings Contribution Agreement") are hereby converted and reclassified into the **** Common Units (and accompanying voting and economic rights set forth in this Agreement) issued to AKD Holdings pursuant to Section 3.2[a]; [b] Red River acknowledges and agrees that all of its Class A Membership Interest and accompanying Class A Points and Voting Points (as such terms are defined in the Former Operating Agreement) granted pursuant to the Former Operating Agreement and not transferred to AKD Holdings pursuant to the AKD Holdings Contribution Agreement are hereby converted and reclassified into the ****  $\bar{\text{Common}}$  Units (and accompanying voting and economic rights set forth in this Agreement) issued to Red River pursuant to Section 3.2[a]; [c] Pacificom acknowledges and agrees that all of its Class A Membership Interest and accompanying Class A Points and Voting Points (as such terms are defined in the Former Operating Agreement) granted pursuant to the Former Operating Agreement and not transferred to AKD Holdings pursuant to the AKD Holdings Contribution Agreement are hereby converted and reclassified into the **** Common Units (and accompanying voting and economic rights set forth in this Agreement) issued to Pacificom pursuant to Section 3.2[a]; and [d] Graystone acknowledges and agrees that all of its Class B Membership Interest and accompanying Class B Points and Voting Points (as such terms are defined in the Former Operating Agreement) granted pursuant to the Former Operating Agreement and not transferred to AKD Holdings pursuant to the AKD Holdings Contribution Agreement are hereby converted and reclassified into the **** Common Units (and accompanying voting and economic rights set forth in this Agreement) issued to Graystone pursuant to Section 3.2[a]. Contemporaneously with or prior to the execution of this Agreement, GCI will make or will have made the Capital Contribution to the Company contemplated to be made by GCI with respect to its Units pursuant to Section 2.3 of the Reorganization Agreement. The agreed Fair Market Value of GCI's contribution as specified in Schedule 4.1 will be credited to GCI's Capital Account with respect to such Units, and such agreed Fair Market Value will be deemed to be the amount of GCI's Initial Capital Contribution. The Capital Account of the Non-GCI Members will be adjusted as of the Effective Date pursuant to Treasury Regulation ss. 1.704-1(b)(2)(iv)(f)-(g) which will result in the Capital Account balances specified in Schedule 4.1, and such agreed amounts will be deemed to be the amount of such Non-GCI Member's Initial Capital Contribution. Notwithstanding the forgoing, Schedule 4.1 shall be amended as appropriate to adjust the Non-GCI Members' and the GCI Member's Capital Account as appropriate to reflect [i] the adjustments made to the AKD Net Asset Value pursuant to Section 2.3 of the Reorganization Agreement and the resulting change in the number of Common Units owned by the GCI Member pursuant to the terms and conditions of Section 2.3 of the Reorganization Agreement and [ii] the issuance of Common Units and accompanying redemptions pursuant to Section 2.4 of the Reorganization Agreement and any adjustments made to the AKD Redemption Price (as defined in the Reorganization Agreement) pursuant to Section 2.4 of the Reorganization Agreement and any resulting change in the number of Common Units owned by the Members. Exhibit D provides some examples of the operation of Sections 2.2, 2.3 and 2.4 of the Reorganization Agreement and the accompanying adjustments to the Capital Accounts of each Member.

16

- 4.2 Additional Contributions Except as required by law and except for the adjustments contemplated by Section 4.1 above, no Additional Contributions will be required or permitted to be made by any Member except upon the unanimous Vote of the Members.
- 4.3 Capital Accounts. A Capital Account will be maintained for each Member and credited, charged and otherwise adjusted as required by ss. 704(b) of the Code and the ss. 704(b) Regulations. Each Member's Capital Account will be:
  - [a] Credited with [i] the amount of money contributed by the Member as an Initial Contribution or an Additional Contribution, [ii] the Fair Market Value of assets contributed by the Member as an Initial Contribution or Additional Contribution (net of liabilities that the Company assumes or takes subject to that were not taken into account in determining the Fair Market Value of such assets), [iii] the Member's allocable share of Profits and [iv] all other items properly credited to such Capital Account as required by the ss. 704(b) Regulations;
  - [b] Charged with [i] the amount of money distributed to the Member by the Company, [ii] the Fair Market Value of assets distributed to the Member by the Company (net of liabilities that the Member assumes or takes subject to that were not taken into account in determining the Fair Market Value of

such assets), [iii] the Member's allocable share of Losses and [iv] all other items properly charged to such Capital Account as required by the ss. 704(b) Regulations; and

[c] Otherwise adjusted as required by the ss. 704(b) Regulations.

Any unrealized appreciation or depreciation with respect to any asset distributed in kind will be allocated among the Members in accordance with the provisions of Article 4 as though such asset had been sold for its Fair Market Value on the date of Distribution and the Members' Capital Accounts will be adjusted to reflect both the deemed realization of such appreciation or depreciation and the Distribution of such property.

The foregoing provisions and the other provisions of this Agreement relating to the maintenance of Capital Accounts are intended to comply with the ss.  $704\,(b)$  Regulations and will be interpreted and applied in a manner consistent with such Regulations and any amendment or successor provision thereto. The Members will cause appropriate modifications to be made if unanticipated events might otherwise cause this Agreement not to comply with the ss.  $704\,(b)$  Regulations, so long as such modifications do not cause a material change in the relative economic benefits of the Members under this Agreement.

4.4 Adjustments. The Members intend to comply with the ss.  $704\,(b)$  Regulations in all respects, and the Capital Accounts of the Members will be adjusted to the full extent that the ss.  $704\,(b)$  Regulations may apply (including applying the concepts of qualified income offsets and minimum gain chargebacks).

17

- 4.5 Transfer. If any Units are Transferred in accordance with this Agreement, the Capital Account of the Transferor that is attributable to the Transferred Units will carry over to the Transferee.
- 4.6 Market Value Adjustments. Appropriate Capital Account adjustments will be made upon any Transfer of any Units, including those that apply upon the constructive liquidation of the Company under ss. 708(b) of the Code, all in accordance with the ss. 704(b) Regulations. Similarly, if optional basis adjustments are made under ss. 734 or ss. 743 of the Code, appropriate Capital Account adjustments will be made as required by the ss. 704(b) Regulations.
- 4.7 No Withdrawal of Capital. Except as specifically provided in this Agreement, no Member will be entitled to withdraw all or any part of its Capital Contribution from the Company prior to the Company's Dissolution and Liquidation or, when such withdrawal of capital is permitted, to demand a Distribution of property other than money or as otherwise provided in this Agreement.
- 4.8 No Interest on Capital. No Member will be entitled to receive interest on such Member's Capital Account or Capital Contribution.
- $4.9~\mathrm{No}$  Drawing Accounts. The Company will not maintain a drawing account for any Member. All Distributions to Members will be governed by Article 6 (relating to Distributions not in Liquidation of the Company) and by Article 12 (relating to Liquidation).

## ARTICLE 5. PROFITS AND LOSSES

- 5.1 Determination. The terms "Profits" and "Losses" mean, respectively, the net profits and losses of the Company determined for each Fiscal Year in accordance with the method of accounting adopted by the Company for federal income tax purposes, except that such net profit or loss will be determined [a] by including as an item of income any income that is exempt from taxation, [b] by deducting as an expense any expenditure of the Company not deductible in computing its taxable income and not properly chargeable to any Capital Account, or deemed not deductible in computing its taxable income and not properly chargeable to any Capital Account in accordance with the ss. 704(b) Regulations and [c] by calculating the gain, loss, depreciation and amortization on property that is reflected in the Capital Accounts at a book basis different from the basis of such property for federal income tax purposes based on the book basis of such property in accordance with the ss. 704(b) Regulations. Any allocation of Profits or Losses will be considered a pro rata allocation of each item entering into the computation of Profits and Losses.
- 5.2 Allocation of Profits and Losses Generally. Except as otherwise provided in this Agreement, the Profits or Losses of the Company, including items of income, gain, loss and deduction for each Fiscal Year, will be allocated to the Members in proportion to their Units, without distinction as to class or series; provided that [a] no portion of the Profits or Losses attributable to the realization of the value of the Company as of the date that a Profits Interest Unit is issued shall be allocated to that Profits Interest Unit and [b] no portion of any item of income, gain, loss or deduction

attributable to operations of the Company after the date that the Profits Interest Unit is issued, but that a Profits Interest Unit shall participate to the same extent as a Common Unit in gain from the disposition of all or substantially all assets of the Company only to the extent that such gain reflects an increase in the fair market value of the assets of the Company from the date of the issuance of the Profits Interest Unit.

- 5.3 Nonrecourse Deductions. Losses attributable to any Company nonrecourse liability (for which no Member or related Person (within the meaning of the ss. 752 Regulations) bears the economic risk of loss) will be allocated in the same manner as Losses are allocated pursuant to Section 5.2, and Losses of the Company attributable to any Member nonrecourse liability (that is nonrecourse to the Company, but for which one or more Members or related Persons bear the economic risk of loss) will be allocated in accordance with the ss. 704(b) Regulations to those Members bearing (or who, because of their relationship to Persons who bear such economic risk of loss, are deemed to bear) the economic risk of loss for the liability. The allocation of liabilities to a property, the determination of nonrecourse deductions, the effect of property revaluations and all other issues affecting the allocation of nonrecourse deductions will be determined in accordance with the ss. 704(b) Regulations.
- 5.4 Minimum Gain Chargeback. Notwithstanding the general rule on allocation of Profits stated in Section 5.2, if there is a net decrease in Company minimum gain for any Fiscal Year, each Member will be allocated items of Profits for such year equal to such Member's share of the net decrease in Company minimum gain. If there is a net decrease in Member nonrecourse debt minimum gain for any Fiscal Year, each Member having a share of such minimum gain will be allocated items of Profits equal to such Member's share of such net decrease in Company nonrecourse debt minimum gain. The determination of net decreases in Company minimum gain and Member nonrecourse debt minimum gain, allocations of such net decreases, exceptions to minimum gain chargebacks and all other issues affecting the minimum gain chargeback requirements will be determined in accordance with the ss. 704(b) Regulations.
- 5.5 Tax Allocations.Allocation of items of income, gain, loss and deduction of the Company for federal income tax purposes for a Fiscal Year will be allocated, as nearly as is practicable, in accordance with the manner in which such items are reflected in the allocations of Profits and Losses among the Members for such Fiscal Year. To the extent possible, principles identical to those that apply to allocations for federal income tax purposes will apply for state and local income tax purposes.
- 5.6 Qualified Income Offset. Notwithstanding any other provision of this Agreement to the contrary (except Section 5.4, which will be applied first), if in any Fiscal Year or other period a Member unexpectedly receives an adjustment, allocation or Distribution described in the qualified income offset provisions of the ss. 704(b) Regulations, such Member will be specially allocated items of income in an amount and manner sufficient to eliminate, to the extent required

19

by the ss.  $704\,(b)$  Regulations, the Adjusted Capital Account Deficit of such Member as quickly as possible.

- 5.7 Limit on Loss Allocations. Notwithstanding the provisions of Section 5.2 or any other provision of this Agreement to the contrary, Losses (or items thereof) will not be allocated to a Member if such allocation would cause or increase a Member's Adjusted Capital Account Deficit and will be reallocated to the Members (other than any such Member to which the limitations of this Section 5.7 apply), subject to the limitations of this Section 5.7.
- 5.8 ss. 754 Adjustments. To the extent an adjustment to the adjusted tax basis of any Company asset under ss. 734(b) or ss. 743(b) of the Code is required to be taken into account in determining Capital Accounts under the ss. 704(b) Regulations, the amount of the adjustment to the Capital Accounts will be treated as an item of gain (if the adjustment increases the basis of the asset) or loss (if the adjustment decreases the basis), and the gain or loss will be specially allocated to the Members in a manner consistent with the manner in which their Capital Accounts are required to be adjusted under the ss. 704(b) Regulations.
- 5.9 Contributed Property. All items of gain, loss and deduction with respect to property that is reflected in the Capital Accounts of the Members at a basis different from such property's adjusted tax basis will be allocated, solely for tax purposes, among the Members to take into account the variation between the adjusted tax basis of the property and the basis reflected in the Member's Capital Account according to the principles of the ss. 704(c) Regulations. For example, if there is built-in gain with respect to certain property at the time of such property's contribution to the Company, upon the Company's sale of that property the pre-contribution taxable gain (as subsequently adjusted under the ss. 704(c) Regulations during the period such property was held by the Company) would be allocated to the contributing Member (and such pre-contribution gain would not again create a Capital Account adjustment because the property was credited to Capital Account upon contribution at its Fair Market Value). Except as limited by the following sentence, the allocation of tax items with respect to ss. 704(c) property to

Members that do not reflect a basis difference with respect to such property in their Capital Accounts will, to the extent possible, be equal to the allocation of the corresponding book items made to such Members with respect to such property. All tax allocations made under this Section 5.9 will be made in accordance with ss. 704(c) of the Code, and the method of making such allocations will be determined by the Members, acting together.

5.10 Tax Credits.To the extent that the federal income tax basis of an asset is allocated to the Members in accordance with the Regulations promulgated under ss. 46 of the Code, any tax credit attributable to such tax basis will be allocated to the Members in the same ratio as such tax basis. With respect to any other tax credit, to the extent that a Company expenditure gives rise to an allocation of loss or deduction, any tax credit attributable to such expenditure will be allocated to the Members in the same ratio as such loss or deduction. Consistent principles will apply in determining the Members' interests in tax credits that arise from taxable or non-taxable receipts of the Company. All allocations of tax credits will be made as of the time such credit arises. Any

20

recapture of a tax credit will be allocated, to the extent possible, to the Members in the same manner as the tax credit was allocated to them. Except as otherwise specifically provided in the ss. 704(b) Regulations (such as the adjustments required when there is an upward or downward adjustment in the tax basis of investment credit property), allocations of tax credits and their recapture will not be reflected by any adjustment to Capital Accounts.

- 5.11 Gross Income Allocation. In the event any Member has a deficit Capital Account at the end of any Company Fiscal Year that is in excess of the sum of [i] the amount such Member is obligated to restore to the Company pursuant to any provision of this Agreement, [ii] the amount such Member is deemed to be obligated to restore pursuant to the penultimate sentence of Treasury Regulations Section 1.704-2(g)(1) and [iii] the amount such Member is deemed to be obligated to restore pursuant to the penultimate sentence of Treasury Regulations Section 1.704-2(i)(5), each such Member shall be specially allocated items of Company income and gain in the amount of such excess as quickly as possible; provided, however, that an allocation pursuant to this Section 5.11 shall be made only if and to the extent that such Member would have a deficit Capital Account in excess of such sum after all other allocations provided for in this Article 5 have been tentatively made as if Section 5.6 and this Section 5.11 were not in the Agreement.
- 5.12 Curative Allocations. The allocations set forth in Sections 5.3, 5.4, 5.6, 5.8 and 5.11 (the "Regulatory Allocations") are intended to comply with certain requirements of the Treasury Regulations. It is the intent of the Members that, to the extent possible, all Regulatory Allocations shall be offset either with other Regulatory Allocations or with special allocations of other items of Company income, gain, loss or deduction pursuant to this Section 5.12. Therefore, notwithstanding, any other provision of this Article 5 (other than the Regulatory Allocations), the Board of Managers shall make such offsetting special allocations of Company income, gain, loss or deduction in whatever manner it determines appropriate so that, after such offsetting allocations are made, each Member's Capital Account balance is, to the extent possible, equal to the Capital Account balance such Member would have had if the Regulatory Allocations were not part of the Agreement and all Company items were allocated pursuant to Section 5.2. In exercising its discretion under this Section 5.12, the Board of Managers shall take into account any future Regulatory Allocations under Section 5.4 that, although not yet made, are likely to offset Regulatory Allocations made under Section 5.3.
- 5.13 Allocation on Transfer. If any Unit is Transferred during any Fiscal Year of the Company (whether by liquidation or Transfer of a Unit or otherwise), the books of the Company will be closed as of the effective date of Transfer. The Profits or Losses attributed to the period from the first day of such Fiscal Year through the effective date of Transfer will be allocated to the Transferor and the Profits or Losses attributed to the period commencing on the day after the effective date of Transfer will be allocated to the Transferee. In lieu of an interim closing of the books of the Company and with the agreement of the Transferor and the Transferee, the Company may allocate Profits and Losses for such Fiscal Year between the Transferor and the Transferee based on a daily proration of items for such Fiscal Year or any other reasonable method of allocation (including an allocation of extraordinary Company items, as determined by the Company, based on when such items are recognized for federal income tax purposes).

21

## ARTICLE 6. DISTRIBUTIONS

6.1 Distributions Generally. Except for liquidating Distributions under Section 12.3 and except as otherwise provided by Section 6.5, the Company will make all Distributions of Available Cash to the Members in proportion to their Units, without distinction as to class, subject to the limitation that Distributions in respect of Profits Interest Units shall relate only to Profits earned and increases in value of the Company after the date that such Profits Interest Units are issued. Except as provided by Section 6.5, the timing and amount of Distributions shall be determined by the Board of Managers.

- 6.2 Payment. All Distributions will be made to Members owning Units on the date of record, such date being the Business Day immediately preceding the date of Distribution, as reflected on the books of the Company.
- 6.3 Withholding. If required by the Code or by state or local law, the Company will withhold any required amount from Distributions to a Member for payment to the appropriate taxing authority. Any amount so withheld from a Member will be treated as a Distribution by the Company to such Member. Each Member will timely file any agreement that is required by any taxing authority in order to avoid any withholding obligation that otherwise would be imposed on the Company.
- 6.4 Distribution Limitations. Notwithstanding any other provision of this Agreement, the Company will not make any Distribution to the Members if, after the Distribution, the liabilities of the Company (other than liabilities to Members on account of their Units) would exceed the Fair Market Value of the Company's assets. With respect to any property that is subject to a liability for which the recourse of creditors is limited to the specific property, such property will be included in assets only to the extent the property's Fair Market Value exceeds its associated liability, and such liability will be excluded from the Company's liabilities. Notwithstanding any other provision of this Agreement, the Company will not make a Distribution to any Member if such Distribution would cause or increase any Member's Adjusted Capital Account Deficit.
- 6.5 Tax Distribution. For each Fiscal Year the Company will, during such Fiscal Year or the immediately subsequent Fiscal Year, but not later than 90 days following the end of each Fiscal Year, to the extent that there is Available Cash distribute to each Member, with respect to such Fiscal Year, a distribution in an amount equal to such Member's Presumed Tax Liability for such Fiscal Year (a "Tax Distribution"). Any amount distributed pursuant to this Section 6.5 will be deemed to be an advance distribution of amounts otherwise distributable to the Members pursuant to Sections 6.1 and will reduce the amounts that would subsequently otherwise be distributable to the Members pursuant to such provisions in the order they would otherwise have been distributable. The Board of Managers may distribute Tax Distributions quarterly on an estimated basis prior to the end of a Fiscal Year, but if the amounts so distributed as estimated Tax Distributions exceed the amount of Tax Distributions to which such Member is entitled to for such Fiscal Year, the Member will promptly after the end of the Fiscal Year return such excess to the Company and the excess will be treated as a distribution to such Member pursuant to Section 6.1, as applicable until it is returned.

### 2.2

## ARTICLE 7. MANAGEMENT

7.1 Management. Management of the Company will be vested exclusively in the Board of Managers. Except as otherwise provided in this Agreement: [a] the Board of Managers has complete and unrestricted power and authority to manage the business, properties and activities of the Company in its sole and exclusive discretion, [b] no Person dealing with the Company will be required to inquire into the authority of the Board of Managers (or any designee of the Board of Managers) to take any action or make any decision, [c] notwithstanding any powers granted to members of a limited liability company under the Act, no Member will take part in the operations, management or control of the Company's business, transact any business in the Company's name, or have the power to sign documents for or otherwise bind the Company except for such actions that are specifically authorized by the Board of Managers or as otherwise provided by this Agreement, and [d] the Board of Managers has the rights, authority and powers of a "manager" under the Act with respect to the Company business and assets as provided in the Act as in effect on the Effective Date. Without limiting the foregoing, the Board of Managers has all of the responsibilities and authority of the board of directors of an Alaska business corporation, subject to the express provisions of this Agreement; provided, that the reference to Alaska business corporations is not intended and will not be construed to subject the Company to any restriction or limitation or to subject the Managers to any duty or liability applicable to Alaska corporations or their directors that is not otherwise applicable to an Alaska limited liability company or its managers or agents. Concurrently with the execution of this Agreement, the Company shall enter into the Management Agreement, pursuant to which Fire Lake will perform certain management duties for the Company as described therein. Notwithstanding any other provision of this Agreement to the contrary, the Company will not take any of the following actions, and neither the Board of Managers nor Fire Lake acting under the Management Agreement shall have any authority to take any of the following actions on behalf of the Company, without the written consent of the GCI Member: [i] cause or permit the Company to engage in any business other than the business described in Section 2.1; [ii] except as provided in the current Annual Budget, cause or permit the Company to purchase or otherwise acquire additional assets having an aggregate cost of \$250,000 or more in any transaction or series of related transactions; [iii] cause or permit the Company to merge or consolidate with any Person; [iv] except as provided in the current Annual Budget, cause or permit the Company to sell, lease or otherwise dispose of assets having an aggregate value of \$250,000 or more in any transaction or series of related transactions; [v] cause or

permit the Company to engage in, enter into or amend any contract, arrangement or transaction in which any Member or any Affiliate of a Member has a direct or indirect interest; [vi] cause or permit the Company to authorize, issue or enter into any agreement providing for any issuance (contingent or otherwise) of any additional Units or other securities, except as contemplated by Article 3 above; [vii] cause or permit the Company to authorize or permit any mandatory or permissive Additional Contributions or to admit any additional Members (other than in connection with a Transfer of Units made in accordance with the provisions of Article 13, which will be governed in all respects by the provisions of Article 13); [viii] cause or permit the Company to redeem any Units or make any other extraordinary distributions not contemplated by Article 6, except as contemplated by Article 3 above; [ix] cause or permit the Company

2.3

to incur or guarantee any indebtedness (other than the incurrence of trade payables in the ordinary course of business) or incur or quarantee any Lien, except for indebtedness or Liens provided in the Annual Budget and except for indebtedness or Liens contemplated by the Reorganization Agreement; [x] cause or permit the Company to approve any deviation from the Annual Budget then in effect of 10% or greater from an approved line item or budget category or to engage in any transaction which has not been budgeted for in the Annual Budget then in effect; [xi] cause or permit the Company to liquidate, wind up, dissolve, or cease to continue as an ongoing business concern (other than in connection with any events of Dissolution specified in Section 11.1, which will be governed in all respects by the provisions of Section 11.1), effect a recapitalization or reorganization in any form of transaction, commence any bankruptcy or insolvency proceeding, acquiesce to the appointment of a receiver, trustee, custodian or liquidator or admit the material allegations of a petition filed against the Company in any bankruptcy or insolvency proceeding; [xii] cause or permit the Company to change from a limited liability company to a different organizational form; [xiii] cause or permit any direct or indirect subsidiary of the Company to do any of the foregoing in respect of the subsidiary; or [xiv] enter into an agreement or otherwise commit to do any of the foregoing.

- 7.2 Appointment of Board of Managers. The Board of Managers will initially consist of five (5) Managers. The number of Managers on the Board of Managers shall be fixed from time to time by the Board of Managers, but shall not be less than four Managers or more than eight Managers. The GCI Member shall at all times have the sole right to appoint and remove one Manager and the AKD Holdings Member shall have the sole right to appoint and remove the remaining members of the Board of Managers. The AKD Holdings Member may not appoint as a Manager, any Person who serves on the board of directors or comparable governing body of a communications company that competes with GCI. The AKD Holdings Member will elect a sufficient number of Managers deemed independent from GCI under applicable laws, regulations or stock exchange rules to allow GCI to comply with any such applicable laws, regulations or stock exchange rules. The name of the initial Manager appointed by GCI Member will be John M. Lowber and the names of the initial Managers appointed by the AKD Holdings Member will be Stephen Roberts, William M. Yandell III, James D. Lackie and John Tindall. Each Manager is entitled to appoint an alternate to serve in his or her absence at any meeting of the Board of Managers. Each Manager will serve on the Board of Managers until his or her resignation or removal by the Member that appointed such Manager. Either the GCI Member or the AKD Holdings Member may, at any time, remove a Manager appointed by such Member and appoint a substitute Manager by delivering Notice of such removal and appointment to the other Members. Any vacancy on the Board of Managers resulting from the death, disability or resignation of a Manager will be filled by the Member that appointed such Manager. No compensation will be paid to any Manager for serving in such capacity, except that Managers will be entitled to reimbursement for reasonable expenses incurred in connection with such service.
  - 7.3 Procedural Requirements -- Meetings Of Members and the Board of Managers.
  - [a] Action by Members. Except as otherwise expressly provided in this Agreement, [i] all actions requiring the approval of the Members will be deemed approved if Members owning more than 50% of the outstanding Units as of the record date for the meeting or written consent Vote in favor of approval, [ii] all Units will vote together as a single voting group, and [iii] each Unit will have one vote.

24

- [b] Action by Board of Managers. Except as otherwise expressly provided in this Agreement, the act of the majority of the Managers present at a meeting at which a quorum is present shall be the act of the Board of Managers.
- [c] Meetings of Members and the Board of Managers.
  - [i] Annual Meeting of Members. An annual meeting of the Members will be held on such date and at time as may be determined by the Board of Managers. The purpose of the annual meeting is to review the Company's operations for the preceding Fiscal Year and to transact such other business as may come before the meeting. The

failure to hold any annual meeting has no adverse effect on the continuance of the Company.

- [ii] Special Meetings. Special meetings of the Members, for any purpose or purposes, may be called by the Board of Managers or by any Member or Members owning at least 10% of the Units then outstanding.
- [iii] Meetings of the Board of Managers. The Board of Managers will meet from time to time at the request of any Manager.
- [iv] Place. The person calling a meeting of Members or of the Board of Managers may designate the place of the meeting. If the place so designated for a meeting of Members is not in the Anchorage, Alaska metropolitan area and for a meeting of the Board of Managers is not in the Anchorage, Alaska or Memphis, Tennessee metropolitan area, then such location must be agreed in the case of a Member meeting, by Members who own more than 50% of the Units then outstanding, and in the case of a Board of Managers meeting, by all of the Managers. If no designation is made by a person calling a meeting of Members or the Board of Managers, the place of meeting will be the Company's principal place of business.
- [v] Notice. Notice of any Board of Managers or Members meeting must be given not less than three Business Days nor more than 30 days before the date of the meeting. Such Notice must state the place, day and hour of the meeting and, in the case of a special Members meeting, the purpose for which the meeting is called.
- [vi] Waiver of Notice. Any Member or Manager may waive, in writing, any Notice required to be given to such Member or Manager, whether before or after the meeting or other event to which such Notice relates. Attendance by a Member or Manager at a meeting will constitute a waiver of notice of the meeting, unless the

25

Member or Manager attends the meeting for the sole and express purpose of objecting to the transaction of any business on the ground that the meeting is not lawfully called.

- [vii] Record Date. For the purpose of determining Members entitled to Notice of and to vote at any meeting of Members, or to sign any written consent, the last Business Day before the day on which such Notice or consent is first transmitted to the Members will be the record date. Any such determination of Members entitled to vote at any meeting of Members will apply to any adjournment of a meeting.
- [viii] Quorum. A quorum at any meeting of Members will consist of Members who own more than 90% of the outstanding Units on the record date for the meeting (which Members may be in attendance in person, by proxy, by telephone or by video conference). A quorum at any meeting of the Board of Managers will consist of a majority of the number of directors fixed by the Board of Managers pursuant to Section 7.2 (which Managers may be in attendance in person, by proxy, by telephone or by video conference). Any Board of Managers or Members meeting at which a quorum is not present may be adjourned to a specified place, day and hour without further Notice.
- [ix] Proxies. At any meeting of Members or the Board of Managers, a Member or a Manager may vote in person or by written proxy given to another Member or Manager. Such proxy must be signed by the Member or Manager, or by a duly authorized attorney-in-fact, and must be filed with the Company before or at the time of the meeting. No proxy will be valid after eleven months from the date of its signing unless otherwise provided in the proxy. Attendance at the meeting by the Member or Manager giving the proxy will revoke the proxy during the period of attendance.
- [x] Meetings by Telephone or Video. The Members and the Managers may participate in a meeting by means of conference telephone or video or similar communications equipment by which all Members or Managers participating in the meeting can hear each other at the same time. Such participation will constitute presence in person at the meeting and waiver of any required Notice. The Company will take all reasonable steps to ensure that Members and Managers are able to participate by telephone or video conference in meetings of Members and meetings of the Board of Managers, respectively.
- [xi] Observers. The Board of Managers will permit any individuals designated by the GCI Member or the AKD Holdings Member to be observers at any meetings of the Board of Managers, except that

the AKD Holdings Member can not designate any person to be an observer if such person serves on the board of directors or comparable governing body of a communications company that competes with GCI.

26

- [d] Action Without a Meeting. Any action required or permitted to be taken at a meeting of Members may be taken without a meeting if the action is evidenced by one or more written consents describing the action taken, signed by Members whose aggregate Units would enable them to approve the action at a meeting of the Members at which all Members were present and voted. Any action required or permitted to be taken at a meeting of Managers may be taken without a meeting if, and will be effective when, the action is evidenced by one or more written consents describing the action taken, signed by all Managers. Action so taken is effective when sufficient Members or Managers approving the action have signed the consent, unless the consent specifies a different effective date. If any action is taken by a written consent that is not signed by all Members, Notice of the action, accompanied by a copy of the written consent, will be sent to each Member who did not sign.
- 7.4 Officers. The Board of Managers may from time to time appoint executive officers of the Company and designate their authority and duties to manage the day-to-day operations of the Company. Unless otherwise determined by the Board of Managers, if the title of an officer is one commonly used for officers of a business corporation formed under the Alaska Corporation Code, the assignment of such title will constitute the delegation to such person of the authorities and duties that are normally associated with that office. Such officers will take all actions which are necessary and appropriate to conduct the day-to-day operations of the Company's business, subject to the supervision of the Board of Managers and the provisions of this Agreement. No compensation will be paid to any officers for serving in such capacity, unless such compensation is paid pursuant to an employment agreement approved by all members of the Board of Managers.
- 7.5 Annual Budget. The Board of Managers or a committee duly appointed by the Board of Managers, which committee must include the Manager appointed by the GCI Member under Section 7.2 (the "Budget Committee") will require the appropriate officers, employees and representatives of the Company to prepare and present an Annual Budget for the Company and its subsidiaries at least ninety (90) calendar days in advance of the beginning of the applicable Fiscal Year.
  - [a] Each Annual Budget shall cover a one-year period corresponding to a Fiscal Year, provided that the first Annual Budget shall cover the 12-month period commencing January 1, 2006. Each Annual Budget shall include an income statement prepared on an accrual basis which shall show in reasonable detail the revenues and expenses projected for the operations of the Company and its subsidiaries for the forthcoming Fiscal Year and a cash flow statement which shall show in reasonable detail the receipts and disbursements projected for the operations of Company and its subsidiaries for the forthcoming Fiscal Year, the amount of any corresponding cash deficiency or surplus, and contemplated borrowings under credit facilities, if any.
  - [b] Such Annual Budget shall be prepared on a basis consistent with the financial statements of the Company and its subsidiaries and GAAP. The Board of Managers or the Budget Committee shall review and discuss the proposed Annual Budget in consultation with the appropriate officers, employees and representatives of the Company. The proposed Annual Budget shall be deemed approved if all of the Managers then in office approve the

27

Annual Budget, or if approved by all of the members of the Budget Committee. If such approval is obtained, then such Annual Budget shall for all purposes of this Agreement constitute the Annual Budget and shall supersede any previously approved Annual Budget. If such approval is not obtained, then, until a new budget is approved, the Annual Budget for the Company for the immediately preceding Fiscal Year will remain in effect, adjusted (without duplication) to reflect the following increases or decreases: [i] the operation of escalation or de-escalation provisions in contracts then in effect solely as a result of the passage of time or contracts entered into pursuant to an approved Annual Budget or the occurrence of events beyond the control of the Company, to the extent such contracts are still in effect; [ii] elections made in any prior year under contracts contemplated by the Annual Budget for the prior year regardless of which party to such contracts makes such election; [iii] the effect of the existence of any multi-year contract entered into in accordance with a previous budget to the extent not fully reflected in the prior year's Annual Budget; [iv] increases or decreases in expenses attributable to the annualized effect of employee additions or reductions during the prior year contemplated by the Annual Budget for the prior year; [v] interest expense attributable to any loans; [vi] increases or decreases in overhead expenses in an amount equal to the total of overhead expenses reflected in the Annual Budget for the prior year (excluding non-recurring items) multiplied by the percentage increase or decrease in the U.S. Department of Labor Bureau of Labor Statistics Consumer Price Index for all Urban Consumers or a successor index for the prior Fiscal Year (but in no event will such change be more than 10% of the corresponding items in the prior Annual Budget); and [vii] decreases in expenses attributable to non-recurring items reflected in the prior year's Annual Budget.

- [c] The initial Annual Budget shall be attached hereto as Exhibit E.
- 7.6 Curative Provision. To the extent the GCI Member's management rights in the Company exceed any allowable control requirements for ownership of wireless communications carriers under any agreement or understanding to which GCI may be bound or under any law or regulation of any Governmental Authority to which GCI may be subject, the Members will negotiate in good faith an amendment to this Agreement that will contain curative provisions regarding voting interests in the Company.

#### ARTICLE 8. LIABILITY OF A MEMBER

8.1 Limited Liability. Except as otherwise provided in the Act, the debts, obligations and liabilities of the Company (whether arising in contract, tort or otherwise) will be solely the debts, obligations and liabilities of the Company, and no Member (or former Member) of the Company is liable or will be obligated personally for any such debt, obligation or liability of the Company solely by reason of such status. No Manager or Officer of the Company nor any officer, director, employee or agent of any Member will have any personal liability for the performance of any obligation of any Member under this Agreement.

28

- 8.2 Capital Contribution. Each Member is liable to the Company for [a] the Initial Contribution deemed to be made under Section 4.1 and [b] subject to Section 8.3, any Capital Contribution or Distribution that has been wrongfully or erroneously returned or made to such Person in violation of the Act, the Articles or this Agreement.
- 8.3 Capital Return. Any Member who has received the return of all or any part of such Member's Capital Contribution will have no liability to return such Distribution to the Company after the expiration of the applicable period of time specified by the Act or other applicable law unless Notice of an obligation to return is given to such Person within such time period; provided that if such return of capital has occurred without violation of the Act, the Articles or this Agreement, such obligation to return capital will apply only to the extent necessary to discharge the Company's liability to its creditors who reasonably relied on such obligation in extending credit prior to such return of capital.
- 8.4 Reliance. Any Member will be fully protected in relying in good faith upon the records of the Company and upon such information, opinions, reports or statements by [a] any of the Company's other Members, Board of Managers or Officers or [b] any other Person who has been selected with reasonable care as to matters such Member reasonably believes are within such other Person's professional or expert competence. Matters as to which such reliance may be made include the value and amount of assets, liabilities, Profits and Losses of the Company, as well as other facts pertinent to the existence and amount of assets from which Distributions to Members might properly be made.

### ARTICLE 9. INDEMNIFICATION

9.1 General. To the full extent permitted by law, the Company will indemnify, defend and hold harmless each Member (and each such Member's shareholders, directors, officers, partners, members, employees, Affiliates and agents), Manager and each Officer of the Company (collectively, "Indemnified Persons") from and against any and all claims, damages, causes of action, losses, expenses (including reasonable fees and expenses of attorneys and other advisors and any court costs incurred by such Indemnified Person) and liabilities (collectively, "Damages") arising from or in connection with the business or affairs of the Company, the preservation of the business and property of the Company or the defense or disposition of any claim, demand or Proceeding in which such Indemnified Person may be involved or with which such Indemnified Person may be threatened to be involved, as a party or otherwise because such Person was a Member, Manager or Officer (or was a shareholder, director, officer, partner, member, employee, Affiliate or agent of a Member) or acted or failed to act with respect to the business or affairs of the Company if [a] such Person acted in good faith, [b] such Person reasonably believed that its conduct in an official capacity was in the Company's best interests or, if the conduct was not in an official capacity, that its conduct was at least not opposed to the Company's best interests and [c] such Person, in the case of any criminal Proceeding, had no reasonable cause to believe its conduct was unlawful. The termination of any action or Proceeding by judgment, order, settlement, conviction, or upon a plea of nolo contendere or its equivalent will not of itself create a presumption that indemnification is not available under this Agreement.

- 9.2 Exception. Notwithstanding the general rule stated in Section 9.1, the Company will not indemnify any Person in connection with [a] any Proceeding by or in right of the Company in which such Person was adjudged liable to the Company, or [b] in connection with any Proceeding charging improper personal benefit to such Person (or another Person of which such Person is or was a shareholder, director, officer, partner, member, employee or agent) (whether or not involving action in an official capacity) in which such Person was adjudged liable on the basis that personal benefit was improperly received.
- 9.3 Expense Advancement. With respect to the reasonable expenses incurred by an Indemnified Person who is a party to a Proceeding, the Company may provide funds to such Person (and, in the case of a Member, to the shareholders, directors, officers, partners, members, employees, Affiliates and agents of such Person) in advance of the final disposition of the Proceeding if [a] such Person furnishes the Company with such Person's written affirmation of a good-faith belief that it has met the standard of conduct described in Section 9.1, [b] such Person agrees in writing to repay the advance if it is determined that it has not met such standard of conduct and [c] the Company determines that, based on then known facts, indemnification is permissible under this Article.
- 9.4 Insurance. The indemnification provisions of this Article do not limit any Person's right to recover under any insurance policy maintained by the Company. If, with respect to any loss, damage, expense or liability described in Section 9.1, any Person receives an insurance policy indemnification payment that, together with any indemnification payment made by the Company, exceeds the amount of such loss, damage, expense or liability, then such Person will immediately repay such excess to the Company.
- 9.5 Indemnification of Others. To the same extent that the Company will indemnify and advance expenses to a Member, the Company may indemnify and advance expenses to any employee or agent of the Company. In addition, the Company, in its discretion, may indemnify and advance expenses to any employee or agent to a greater extent than a Member.
- 9.6 Exculpation. No Indemnified Person will be liable to the Company or any other Member for any Damages incurred by reason of any act or omission performed or omitted by such Indemnified Person in good faith on behalf of the Company and in a manner reasonably believed to be within the scope of the authority conferred on the applicable Member, Manager or Officer by this Agreement, except that an Indemnified Person will be liable to the Company for any such Damages incurred by reason of such Indemnified Person's willful misconduct.
- 9.7 Rights Not Exclusive. The rights accruing to each Indemnified Person under this Article 9 will not exclude any other right to which such Indemnified Person may be lawfully entitled.

### 30

### ARTICLE 10. ACCOUNTING AND REPORTING

- 10.1 Fiscal Year. For income tax and accounting purposes, the fiscal year of the Company will be the Fiscal Year.
- $10.2~{\rm Tax}$  Accounting Method. For income tax purposes, the Company will use the accrual method of accounting, unless otherwise required by the Code. The Tax Matters Partner will have the authority to adopt all other accounting methods for tax purposes.
- 10.3 Tax Elections. Notwithstanding any other provision of this Agreement, no Member, Manager or employee of the Company may take any action (including, but not limited to, the filing of a U.S. Treasury Form 8832 Entity Classification Election) which would cause the Company to be characterized as an entity other than a partnership for U.S. federal income tax purposes without the consent of the GCI Member. The Tax Matters Partner will have the authority to make any other tax elections, and to revoke any such election, as the Tax Matters Partner may from time to time determine. Notwithstanding the preceding sentence, following any Transfer (within the meaning of ss. 754 of the Code) of a Unit, the Tax Matters Partner will make the election under ss. 754 of the Code.
- $10.4~{
  m Returns.}$  At the expense of the Company, the Tax Matters Partner will cause the preparation and timely filing of all tax returns required to be filed by the Company pursuant to the Code, as well as all other tax returns required in each jurisdiction in which the Company does business.
- 10.5 Reports; Annual Financial Statements; Regulatory Reporting Obligations. The Company will prepare or will cause the preparation of within the time frames requested by the GCI Member from time to time, such financial statements of the Company and other financial information prepared in accordance with GAAP as GCI may require (the "GCI Requested Financial Information"), as reasonably determined by GCI, to enable it to consolidate the Company's results of operations with GCI's results of operations for purposes of U.S. financial accounting reporting rules and regulations and to meet on a timely basis, GCI's reporting or other obligations under applicable law, the rules and regulations

promulgated thereunder and interpretations thereof by the applicable regulatory authority or its staff, including, without limitation, the U.S. Securities Act of 1933, as amended, and the U.S. Securities Exchange Act of 1934, as amended. The Company will provide such GCI Requested Financial Information to GCI within 10 to 13 days of each fiscal year, quarter or month end, as applicable, or such shorter time period as may be required by GCI pursuant to the preceding sentence. The Company shall within the time frame requested by the GCI Member from time to time, take such action or produce such other information, statements and reports, as may be required by applicable stock exchange or stock associations rules or by applicable law, the rules and regulations promulgated thereunder or interpretations thereof by the applicable regulatory authority or its staff, as reasonably determined by the GCI Member to timely meet its or its Affiliates' disclosure, reporting or other obligations under the rules of any stock exchange or stock association on which its shares are listed and under any applicable law and the rules and regulations promulgated thereunder or

31

interpretations thereof by the applicable regulatory authority or its staff, including, without limitation, the U.S. Securities Act of 1933, as amended, the U.S. Securities Exchange Act of 1934, as amended, and the Sarbanes-Oxley Act of 2002, as amended.

- 10.6 Books and Records.
- [a] The following books and records of the Company will be kept at its principal office: [i] a current list of the full name and last known business, residence or mailing address of each Member; [ii] originals of the Articles and of this Agreement, as amended (as well as any signed powers of attorney pursuant to which any such document was executed); [iii] a copy of the Company's federal, state and local income tax returns and reports and annual financial statements of the Company, for the ten most recent years; and [iv] minutes, or minutes of action or written consent, of every meeting of Members of the Company.
- [b] The Company will keep at the Company's principal office separate books of account for the Company which will show a true and accurate record of all costs and expenses incurred, all credits made and received, and all income derived in connection with the operation of the Company in accordance with GAAP consistently applied as to the Company's financial position and results of operations.
- [c] Each Member will have the right, at any time with reasonable Notice to the Board of Managers and at such Member's sole expense, to examine, copy and audit the Company's books and records during normal business hours. All books and records (including bills and invoices), reports and returns of the Company required by this Article will be maintained in a commercially reasonable manner as reasonably determined by the Board of Managers.
  - 10.7 Information.
- [a] Each Member has the right, from time to time and upon reasonable demand for any purpose reasonably related to such Person's interest as a Member of the Company, to obtain from the Company: [i] a current list of the full name and last known business, residence or mailing address of each Member; [ii] a copy of the Articles and of this Agreement, as amended (as well as any signed powers of attorney pursuant to which any such document was executed); [iii] a copy of the Company's federal, state and local income tax returns and reports and annual financial statements of the Company, for the six most recent years; [iv] minutes, or minutes of action or written consent, of every meeting of the Members of the Company and the Board of Managers; [v] true and full information regarding the amount of money and a description and statement of the agreed value of any other property or services contributed or to be contributed by each Member, and the date on which each became a Member; [vi] true and full information regarding the status of the business and financial condition of the Company; and [vii] other information regarding the affairs of the Company as is just and reasonable. Any demand by a Member under this 10.7 must be by Notice to the Company, and must state the purpose of the demand. Any inspection or

32

copying of the Company's books and records under this 10.7 will be during normal business hours, and at the expense of the Member making the demand.

- [b] The Board of Managers will cause the Company to provide to each Member [i] not more than ten days following the end of the fiscal quarter, an estimate of any taxable income or gain to be allocated to such Member for such fiscal quarter and [ii] not more than 75 days after each Fiscal Year end, such information for such Fiscal Year as the Member reasonably requires to prepare tax returns or reports required to be filed by it or one or more of its Affiliates, including federal and state tax information and projections and estimates.
- 10.8 Banking. The Company may establish and maintain one or more accounts or safe deposit boxes at banks or other financial institutions. The Company may authorize one or more individuals to sign checks on and withdraw

funds from such bank or financial accounts and to have access to such safe deposit boxes, and may place such limitations and restrictions on such authority as the Company deems advisable. No funds of the Company will be commingled with funds of any Member or any other Person.

10.9 Tax Matters; Tax Matters Partner. Until further action by the Company, the GCI Member (or any Transferee of a majority of the Units owned by such Member) is designated as the Tax Matters Partner under ss. 6231(a)(7) of the Code. The Tax Matters Partner will take no action that is reasonably expected to have a material adverse effect on one or more of the Members unless such action is approved by the unanimous Vote of the Members. The Tax Matters Partner will be responsible for notifying all Members of ongoing tax Proceedings, both administrative and judicial, and will represent the Company throughout any such Proceeding. The Members will furnish the Tax Matters Partner with such information as it may reasonably request to provide the Internal Revenue Service with sufficient information to allow proper notice to the Members. If an administrative Proceeding with respect to a partnership item under the Code has begun, and the Tax Matters Partner so requests, each Member will notify the Tax Matters Partner of its treatment of any partnership item on its federal income tax return, if any, which is inconsistent with the treatment of that item on the partnership return for the Company. Any settlement agreement with the Internal Revenue Service will be binding upon the Members only as provided in the Code. The Tax Matters Partner will not bind any other Member to any extension of the statute of limitations or to a settlement agreement without such Member's written consent. Any Member who enters into a settlement agreement with respect to any partnership item will notify the other Members of such settlement agreement and its terms within 30 days after the date of settlement. If the Tax Matters Partner does not file a petition for readjustment of the partnership items in the Tax Court, federal District Court or Claims Court within the 90-day period following a notice of a final partnership administrative adjustment, any notice partner or 5-percent group (as such terms are defined in the Code) may institute such action within the following 60 days. The Tax Matters Partner will timely notify the other Members in writing of its decision. Any notice partner or 5-percent group will notify the other Members of its filing of any petition for readjustment.

33

10.10 Classification of Company as Partnership for Tax Purposes, Not State Law. The Company will be classified as a partnership for federal (and, as appropriate, state and local) income tax purposes. This characterization, solely for tax purposes, does not create or imply a general partnership or limited partnership among the Members for state law or any other purpose. Instead, the Members acknowledge the status of the Company as a limited liability company formed under the Act. All duties and obligations of the Members to each other are expressly set forth in this Agreement. Without limiting the foregoing, the Members do not owe to each other or to the Company the duties that a general partner owes to a partnership and its other partners nor do the Managers owe such duties to each other, the Company or its Members, it being acknowledged that the duties owed by the Managers to each other and the Company are as set forth in Article 9. The Members do not have any express or implied fiduciary duties to the Company or each other except the fiduciary duties, if any, that shareholders in an Alaska corporation might have to each other or the corporation.

## ARTICLE 11. DISSOLUTION

- 11.1 Dissolution. Dissolution of the Company will occur upon the happening of any of the following events:
  - [a] The affirmative Vote of Members owning more than 90% of the outstanding Common Units;
  - [b] The sale, lease or other disposition of all or substantially all of the assets of the Company in any transaction or series of transactions;
  - [c] Entry of a decree of judicial dissolution under the Act;
  - [d] an event of Withdrawal (as defined in Section 11.2) of a Member and the election of the remaining Members to dissolve in accordance with Section 11.3; or
  - [e] if either the GCI Member or any Non-GCI Member has materially breached a material provision of this Agreement and such breach has not been cured within 30 days after receipt of a Notice from the non-breaching Member providing reasonable detail concerning the nature of the breach, then upon the election of the non-breaching Member.
- 11.2 Events of Withdrawal. An event of Withdrawal of a Member occurs when any of the following occurs:
  - [a] with respect to any Member, upon the Transfer of all of such Member's Units (which may only be done as otherwise permitted under this Agreement and which Transfer is treated as a resignation);
  - [b] with respect to any Member, upon the voluntary withdrawal, retirement

34

- [c] with respect to any Member that is a corporation, upon filing of articles of dissolution of the corporation;
- [d] with respect to any Member that is a partnership, a limited liability company or a similar entity, upon dissolution and liquidation of such entity (but not solely by reason of a technical termination under ss. 708(b)(1)(B) of the Code); or
- [e] with respect to any Member, the Bankruptcy of the Member.

Within 10 days following the happening of any event of Withdrawal with respect to a Member, such Member must give Notice of the date and the nature of such event to the Company.

11.3 Continuation. In the event of Withdrawal of a Member, the Company will be continued unless the remaining Members (including the Permitted Transferee of a withdrawing Member, if applicable) unanimously elect to dissolve. If the Company is so continued, any Member as to which an event of Withdrawal specified in Sections 11.2[b] through 11.2[e] has occurred, or such Member's Transferee or other successor-in-interest (as the case may be) if a Member has made a Transfer in violation of this Agreement and such Transfer is found not to be null and void, will, without further act, become a "Limited Owner" of its own Units or the Units of the withdrawn Member. A Limited Owner has no right: [a] to participate or interfere in the management or administration of the Company's business or affairs, including by virtue of appointment of one or more Managers, [b] to vote or agree on any matter affecting the Company or any Member, [c] to require any information on account of Company transactions or [iv] except as provided in the next succeeding sentence, to inspect the Company's books and records. The only rights of a Limited Owner are: [i] to obtain the information specified in Section 10.7 if it executes a confidentiality agreement (in form and substance satisfactory to the Board of Managers) concerning such information if not already bound by Section 10.7, [ii] to receive the allocations and Distributions to which the Units of the Limited Owner are entitled and [iii] to receive all necessary tax reporting information. Neither the Company, the Board of Managers nor the Members will owe any fiduciary duty of any nature to a Limited Owner. However, each Limited Owner will be subject to all of the obligations, restrictions and other terms contained in this Agreement as if it were a Member.

## ARTICLE 12. LIQUIDATION

12.1 Liquidation. Upon Dissolution of the Company, the Company immediately will proceed to wind up its affairs and liquidate pursuant to this Section 12.1. If there is only one remaining Member, that Member will act as the liquidating trustee. Otherwise, any Person appointed by the affirmative Vote of Members owning more than 50% of the outstanding Units will act as the liquidating trustee. The Liquidation of the Company will be accomplished in a businesslike manner as determined by the liquidating trustee. A reasonable time will be allowed for the orderly Liquidation of the Company and the discharge of liabilities to creditors so as to enable the Company to minimize any losses attendant upon Liquidation. Any gain or loss on disposition of any Company assets in Liquidation will be allocated to Members in accordance with the provisions of Article 5. Any liquidating trustee is entitled to reasonable

35

compensation for services actually performed, and may contract for such assistance in the liquidating process as such Person deems necessary or desirable. Until the filing of a certificate canceling the Articles under Section 12.8, and without affecting the liability of the Members and without imposing liability on the liquidating trustee, the liquidating trustee may settle and close the Company's business, prosecute and defend suits, dispose of its property, discharge or make provision for its liabilities, and make Distributions in accordance with the priorities set forth in this Article.

## 12.2 [Reserved].

- 12.3 Priority of Payment. The assets of the Company will be distributed in Liquidation in the following order:
  - [a] First, to creditors by the payment or provision for payment of the debts and liabilities of the Company and the expenses of Liquidation, including the setting up of any reserves that are reasonably necessary for any contingent, conditional or unmatured liabilities or obligations of the Company;
  - [b] Second, to the Members that own Units in proportion to the positive balances in their respective Capital Accounts, measured immediately following the Capital Account adjustments arising from the transactions contemplated by Sections 2.2, 2.3 and 2.4 of the Reorganization Agreement; and
  - [c] Third, to the Members that own Units in proportion to the positive

balances in their respective Capital Accounts for their Units after such Capital Accounts have been adjusted to account for the distributions contemplated by Section 12.3[b] and after such Capital Accounts have been adjusted for all allocations of Profits and Losses and items thereof for the Fiscal Year during which such Liquidation occurs.

12.4 Liquidating Distributions. Liquidating Distributions will be made by distributing the assets of the Company in kind to the Members in proportion to the amounts distributable to them pursuant to Section 12.3, valuing such assets at their Fair Market Value (net of liabilities secured by such property that the Member takes subject to or assumes that were not taken into account in determining the Fair Market Value of such assets) on the date of Distribution. Notwithstanding the preceding sentence, but only upon the affirmative Vote of Members owning more than 50% of the outstanding Units, liquidating Distributions may be made by selling the assets of the Company and distributing the net proceeds. Each Member receiving a liquidating Distribution in kind agrees to save and hold harmless the other Members from any and all liabilities assumed by such Member or to which assets distributed to such Member are taken subject by such Member. Appropriate and customary prorations and adjustments will be made incident to any Distribution in kind. The Members will look solely to the assets of the Company for the return of their Capital Contributions, and if the assets of the Company remaining after the payment or discharge of the debts and liabilities of the Company are insufficient to return such contributions, they will have no recourse against any other Member. The Members acknowledge that Section 12.3 may establish Distribution priorities on Liquidation different from

36

those set forth in the Act, as in effect at the time of any Distribution; and, in such event, it is the Members' intention that the provisions of Section 12.3 will control, to the extent possible.

- 12.5 No Restoration Obligation. Except as otherwise specifically provided in Sections 8.2 and 8.3, nothing contained in this Agreement imposes on any Member an obligation to make an Additional Contribution in order to restore a deficit Capital Account upon Liquidation of the Company.
- 12.6 Timing. Final Distributions in Liquidation will be made by the later of [a] the date that all necessary consents from any governmental authorities or third parties are obtained to make an in kind Distribution of any assets, or [b] the end of the Company's Fiscal Year in which such actual Liquidation occurs (or, if later, within 90 days after such event) in the manner required to comply with the ss. 704(b) Regulations. Payments or Distributions in Liquidation may be made to a liquidating trust established by the Company for the benefit of those entitled to payments under Section 12.3, in any manner consistent with this Agreement and the ss. 704(b) Regulations.
- 12.7 Liquidating Reports. A report will be submitted with each liquidating Distribution to Members made pursuant to Section 12.4, showing the collections, disbursements and Distributions during the period that is subsequent to any previous report. A final report, showing cumulative collections, disbursements and Distributions, will be submitted upon completion of the Liquidation.
- 12.8 Articles of Dissolution. Upon Dissolution of the Company and the completion of the winding up of its business, the Company will file Articles of Dissolution (to cancel the Articles) with the Alaska Department of Commerce, Community, and Economic Development pursuant to the Act. At such time, the Company will also file an application for withdrawal of its certificate of authority in any jurisdiction where it is then qualified to do business.

### ARTICLE 13. TRANSFER RESTRICTIONS

- 13.1 General Rule. Except as otherwise provided in Section 13.2, no Member may directly or indirectly Transfer (including by a direct or indirect Transfer of equity interests in or voting rights with respect to the Member, except that a Transfer of equity interests in the GCI Member or a Non-GCI Member will not constitute a Transfer of any Units held by such Member) any of its Common Units without the approval of all members of the Board of Managers, subject in all cases to satisfaction of the conditions set forth in Section 13.4 below. The Profits Interest Units shall not be transferable, except as provided in Article 14 below. Any attempt to Transfer Units in violation of this Agreement will be void ab initio, the Company will not register such attempted Transfer in its records and any purported Transferee will not be recognized as the holder of such Units.
- 13.2 Permitted Transfers. Any transfer of Common Units contemplated by Section 2.4 of the Reorganization Agreement will be permitted without approval of all members of the Board of Managers. In addition, a Transfer of Common Units

to a Person that is an Affiliate of a Member will be permitted without approval of all members of the Board of Managers, so long as any such Person that is an Affiliate of such Member continues to be an Affiliate of such Member at all times while it owns any Common Units. If at any time an Affiliate of a Member to whom a Member has Transferred Common Units under this Section 13.2 ceases to be

an Affiliate of such Member, prior to such cessation such Person will Transfer its Common Units back to such Member or to an Affiliate of such Member. Any Transfer of Common Units under this Section 13.2 shall be subject to the satisfaction of the conditions set forth in Section 13.4.

- 13.3 Permitted Pledges. Any pledge of Common Units pursuant to a bona fide loan transaction or any hedging transaction affecting all or any part of a Member's Common Units will not in itself constitute a Transfer hereunder or cause the Member to cease to be a Member provided that [a] the Member owning such Common Units gives Notice to the other Members of such pledge or hedge at least ten days prior to effecting it and [b] the pledgee or counterparty to the hedging transaction, as applicable, agrees in writing to be bound by and comply with all provisions of this Agreement applicable to the Member effecting such pledge or hedge. Any transfer of title to, or beneficial interest in, any Common Units to a pledgee upon foreclosure or to a counterparty to a hedging transaction upon physical settlement of such hedging transaction will be subject to the transfer restrictions under this Article 13. Any Transfer of Units under this Section 13.3 shall be subject to the satisfaction of the conditions set forth in Section 13.4.
- 13.4 General Conditions on Transfer. No Transfer of a Unit will be effective unless all the conditions set forth below are satisfied:
  - [a] Unless waived by each nontransferring Member, the Transferor signs and delivers to the Company an undertaking in form and substance satisfactory to the Company to pay all reasonable expenses incurred by the Company in connection with the Transfer (including reasonable fees of counsel and accountants and the costs to be incurred with any additional accounting required in connection with the Transfer, and the cost and fees attributable to preparing, filing and recording such amendments to the Articles or other organizational documents or filings as may be required by law);
  - [b] Unless waived by each nontransferring Member, the Transferor delivers to the Company [i] an opinion of counsel for the Transferor reasonably satisfactory in form and substance to the Company to the effect that, assuming the accuracy of the statement of the Transferee described in [ii] below, the Transfer of the Units as proposed does not violate requirements for registration under applicable federal and state securities laws and [ii] a statement of the Transferee in form and substance reasonably satisfactory to the Company making appropriate representations and warranties with respect to compliance with the applicable federal and state securities laws and as to any other matter reasonably required by the Company;
  - [c] The Transferor signs and delivers to the Company a copy of the assignment of the Units to the Transferee (substantially in the form of the attached Exhibit A), which assignment will provide that the Transferor will continue to be liable for the performance of its liabilities under this Agreement; and

## 38 **** CONFIDENTIAL TREATMENT

[d] Unless the Transferee is already a Member, the Transferee signs and delivers to the Company an agreement (substantially in the form of the attached Exhibit B) to be bound by this Agreement.

The Transfer of the Units will be effective as of 12:01 a.m. (Alaska Time) on the first day on which all of the above conditions have been satisfied. The Company will amend Schedule 1.3 as of the effective time of any Transfer of any Units to give effect to such Transfer. No consent of any Member will be necessary to make any such amendment to Schedule 1.3.

13.5 Rights of Transferees. Any Transferee acquiring Units in compliance with this Agreement will become a Member automatically on the effective date of the Transfer

ARTICLE 14. LIQUIDITY RIGHTS.

14.1 ****.

- [a] **** ****. At **** following the **** and **** on the **** following the **** of ****, the **** may **** the **** to **** the ****, of the **** by the ****, at **** [i] the **** accordance **** of ****, or [ii] the **** the **** following the ****. The **** must **** this **** this **** at **** the ****. Upon **** by the **** of the **** will be **** and **** or **** in ****.
- [b] **** ****. At **** the **** and **** on the **** following the **** of ****, the **** may **** the **** to **** the ****, of the **** by the ****, at **** [i] the **** accordance **** of ****, or [ii] the **** the ****. The **** must **** this **** the **** at **** the ****. Upon **** by the **** of the ****, [A] **** that the **** the ****, the **** from the **** and the **** the **** the **** for a **** the **** of such

****; and [B] **** that the **** the **** of the **** by the **** and **** in the ****.

[c] **** - ****. At **** of the ****, the **** may **** the **** the ****, of the **** the **** the **** of such ****. The **** must **** to the **** the **** of the ****. Upon the **** of such **** by the **** of the ****, the **** from the **** and the **** the **** of the **** the **** in **** for a **** the **** of such ****.

## 39 **** CONFIDENTIAL TREATMENT

- [d] **** ****. At **** at **** on the **** following the **** following the ****, the **** may **** the ****, of the **** by the ****, at a **** the **** of such ****. The **** this **** the **** the **** of the ****. As a **** the ****, concurrent with the ****, the **** the **** with **** to the **** that the **** have the **** at the time the **** or that the **** has the **** a **** the **** at the **** of such ****. If the **** the **** set forth in this Section 14.1[d], the ****, within **** of the **** any **** that it has to **** the **** pursuant to Section 14.1[a], 14.1[b] or 14.1[c] (in each case including ****, which shall **** the **** set forth in this Section 14.1[d].
- [e] **** The **** of **** and **** under this Section 14.1 **** at such **** and **** to such **** in ****.

14.2 ****.

- [a] ****. At **** the **** following the ****, the **** may **** the ****, of the **** by the ****, at a **** the **** of such ****. The **** this **** the **** this **** at **** following the ****. At the **** the **** the **** of the ****, the **** may **** the ****, of the **** the **** in **** a **** the **** of such Profits Interest Units.
- [b] ****. At **** the **** following the ****, the **** may **** the **** or the **** the ****, at a **** the **** of such ****. The **** this **** a **** whichever of the **** or the **** that **** in the **** following the **** following the **** in the **** in the **** or the **** reasonably satisfied that the **** in the **** then **** for the **** at the **** of such ****. The ****, as applicable, will cooperate in all respects **** that the **** and, if necessary, the **** or the **** to procure **** for such a ****.
- [c] ****. The **** of **** and **** under this Section 14.2 **** at such **** and **** to such **** in ****.
- 14.3 Determination of Appraised Unit Value. For purposes of this Agreement, "Appraised Unit Value" with respect to Common Units shall mean the Fair Market Value of all, but not less than all, of the Common Units owned by the Member whose Common Units are **** and if applicable, the Fair Market Value of all, but not less than all, of the Profits Interest Units owned by the Fire Lake Member, in each case with no discount or premium for the fact that such Units represent a minority or a controlling interest in the Company. When the Appraised Unit Value of Common Units and/or Profits Interest Units are to be determined **** each will be determined as agreed by the GCI Member and the AKD Holdings Member or if the GCI Member and the AKD Holdings Member fail to agree on the Appraised Unit Value and such failure to agree continues for 10 Business Days after the **** has been delivered, then as determined pursuant to the following appraisal procedure:
  - [a] Each of the GCI Member and the AKD Holdings Member will, within 10 Business Days after the deadline for the GCI Member and the AKD Holdings Member to agree on the Appraised Unit Value, appoint a Qualified Appraiser who will be required as part of its appointment to determine, using an Authorized Valuation Methodology, the Appraised Unit Value of the Common Units and/or the Profits Interest Units and deliver its written independent appraisal thereof to the GCI Member and the AKD Holdings Member within 30

40

days after its appointment. If either the GCI Member or the AKD Holdings Member fail to appoint a Qualified Appraiser within the period provided above, the one Qualified Appraiser appointed will proceed to make the appraisal alone and its appraisal will be the applicable Appraised Unit Value, which will be final and binding upon the Members. "Qualified Appraiser" shall mean a Person experienced in valuing assets owned by the Company and who has no prior business relationship with any Member of the Company within the two years prior to its engagement, and the Members will agree not to hire such Person for six-months following the engagement. "Authorized Valuation Methodology" shall mean one or more valuation methodologies customarily used in the evaluation and appraisal of wireless communications assets, except that any such valuation methodology must be based on an earnings before interest, taxes, depreciation and amortization analysis and shall not be based on a revenue or subscriber count analysis.

[b] If the higher of the two appraisals with respect to the Common Units

and/or the Profits Interest Units is less than 110% of the lower appraisal, the Appraised Unit Value with respect to such Units will be the average of the two appraisals. If the higher appraisal with respect to the Common Units and/or the Profits Interest Units is more than 110% of the lower appraisal, a third Qualified Appraiser, who will be required as part of its appointment to determine, using one or more Authorized Valuation Methodologies, the Appraised Unit Value for such Units within 30 days after its selection and deliver its written independent appraisal thereof to the GCI Member and the AKD Holdings Member, must be selected by the two initial Qualified Appraisers within five Business Days after both initial appraisals have been completed and delivered to the GCI Member and the AKD Holdings Member. The Appraised Unit Value with respect to such Units will be the average of the third appraisal and the one of the first two appraisals that is closest in amount to the third appraisal.

- [c] In the event of the inability or unwillingness of any Qualified Appraiser to act, a new Qualified Appraiser must be appointed in its place within 14 days, such appointment being made in the same manner as provided above for the appointment of the Qualified Appraiser who is being replaced.
- [d] The expense of the Qualified Appraiser appointed by the AKD Holdings Member will be borne by the AKD Holdings Member. The expense of the Qualified Appraiser appointed by the GCI Member will be borne by the GCI Member. The expense of a third Qualified Appraiser will be borne half by the AKD Holdings Member and half by the GCI Member. Notwithstanding the forgoing, the AKD Holdings Member shall be required to pay the expenses of all Qualified Appraisers if [i] the Appraised Unit Value is determined under Section 14.1[a], or [ii] if the Appraised Unit Value determined under Section 14.1[b] is finally determined to be less than the amount set forth in Section 14.1[b][ii].
- [e] In connection with any appraisals performed pursuant to and in accordance with this Agreement, all of the applicable appraisers will be subject to a duty of confidentiality, and all of the parties will cooperate with all applicable appraisers and will provide such appraisers with all necessary and appropriate information reasonably requested by such appraisers in connection with such appraisals.

## 41 **** CONFIDENTIAL TREATMENT

### 14.4 Closing Procedures.

- [a] Unless otherwise agreed by the applicable parties, the closing of the purchase and sale of Common Units and/or Profits Interest Units **** will be completed at 10:00 a.m. local time on a date designated by the Member that provides the applicable **** that is within 15 days after [i] delivering such **** or if a determination of the Appraised Unit Value is made pursuant to Section 14.3, after such determination is made, or [ii] such longer period as is reasonably required to satisfy all of the conditions set forth in Section 14.4[c]. At the closing, the applicable selling Members will deliver to the applicable purchasing Member a written instrument of assignment, substantially in the form attached hereto as Exhibit A, transferring their respective Units to the purchasing Member free and clear of Liens, and the purchasing Member will pay the applicable purchase price. The closing will be consummated at the principal executive offices of the Company unless the GCI Member and the AKD Holdings Member otherwise agree.
- [b] The Member transferring any Common Units and/or Profits Interest Units pursuant to Section 14.1 or Section 14.2 shall be deemed to have represented and warranted that: [i] the purchaser will receive good and valid title to the applicable Units free and clear of all Liens of any nature whatsoever; and [ii] all of such Units can be purchased and sold without any notice to, or consent, approval, order or authorization of, or declaration or filing with, any other Person other than those already obtained and except for any required Governmental Approvals.
- [c] The closing of any purchase and sale of Units will be subject to the satisfaction of the following conditions, it being agreed that the parties will use Commercially Reasonable Efforts to cause such conditions to be met: [i] all material consents, notices, approvals, including Governmental Approvals expressly required with respect to the transactions to be consummated at such closing will have been obtained; and [ii] there will be no preliminary or permanent injunction or other order by any court of competent jurisdiction restricting, preventing or prohibiting the consummation of the transactions to be consummated at such closing.
- [d] Unless the applicable parties agree otherwise, the purchase price on any purchase and sale of Units will be payable by wire transfer of same day funds to an account at a bank designated by the applicable party, such designation to be made no less than two Business Days prior to the applicable closing.
- [e] Except for Sections 13.4[a] and 13.4[b], each of which shall be deemed

to have been waived, all conditions set forth in Section 13.4 must also be satisfied

[f] Notwithstanding the failure of any Member to assign or deliver certificates representing Units on the applicable closing date as required by this Section 14.4, from and after the applicable closing date, the purchaser of the applicable Units shall for all purposes be deemed the record and beneficial owner of such Units, the selling Member shall have only the right to receive the applicable purchase price for such Units, without interest, and any certificates representing the applicable Units shall represent only the right to receive the applicable purchase price, without interest, upon surrender thereof to the purchaser.

42

#### ARTICLE 15. CERTAIN BUSINESS MATTERS

- 15.1 Distribution of Assets in Kind. Any distribution of assets by the Company other than cash will be subject to the receipt of any regulatory or other approvals and waivers deemed necessary by the Board of Managers. Any distribution of assets will be made in accordance with the provisions of Article 6.
- 15.2 Other Business Ventures. Except as provided for in the Management Agreement, each of the Members and their respective Affiliates may engage in or possess interests in other businesses or ventures of any nature without regard to whether such businesses or ventures are or may be deemed to be competitive in any way with the business of the Company or of any Person in which the Company holds an equity interest. Except as provided for in the Management Agreement, no Member will have any obligation to offer any business or investment opportunity to the Company.

## 15.3 Confidentiality.

- [a] Each Member covenants and agrees that so long as he is a Member and thereafter, it will not [i] disclose to any other Person any Confidential Information (hereinafter defined), except for disclosures to Members, Managers, key employees, independent accountants and attorneys of the Company as may be necessary or appropriate in the performance of a Member's duties hereunder; or [ii] use any Confidential Information for any purpose other than the Company's business. Each Member covenants and agrees to cause its respective officers, directors and other representatives, including, without limitation, each Manager appointed by such Member to the Board of Managers (collectively, "Representatives") to observe all terms of this Agreement and shall be responsible for any breach of this Section 15.3 by any of its Representatives.
- [b] The term "Confidential Information" means and includes any and all non-public and proprietary information regarding the assets, liabilities, operations, business, affairs, financing, services, products and trade secrets of the Company, any of its Affiliates or any of their respective officers, directors, shareholders, partners, members, employees or agents. The term "Confidential Information" shall include, without limitation, all financial statements, financial information, projections, forecasts, business plans, methods, ideas, concepts, materials, documents, records, computer programs, customer lists, referral sources, work, models, processes, designs, drawings, plans, inventions, devices, parts, improvements, other physical and intellectual property or other information in any form whatsoever; provided, however, the term "Confidential Information" shall not include any information which [i] at the time of disclosure or thereafter is generally available to and known by the public (other than as a result of its disclosure by a Member or its Representatives in breach of this Section 15.3), [ii] was available to the Member on a non-confidential basis prior to disclosure by the Company, [iii] becomes available to the Member on a non-confidential basis from a Third Party who is not bound by a confidentiality agreement with the Company, or is not otherwise prohibited from transmitting the information to such Member, or [iv] GCI may need to disclose under any applicable law or stock exchange rule as a consequence of it being a public reporting company under the Securities Exchange Act of 1934.

43

- [c] The Confidential Information shall remain the property of the Company; no rights, to use, license or otherwise exploit the Confidential Information are granted to any Member by implication or otherwise; and no Member shall by virtue of the disclosure of the Confidential Information and/or the Member's use of the Confidential Information acquire any rights with respect thereto, all of which rights shall remain exclusively with the Company.
- [d] Each Member acknowledges and agrees that the Company would be irreparably damaged by any unauthorized disclosure or use of any Confidential Information by a Member or its Representatives. Accordingly, without prejudice to the rights and remedies otherwise available to the Company, each Member agrees that the Company shall be entitled, without the requirement of posting a bond or other security, to equitable relief,

including an injunction or specific performance, from any court of competent jurisdiction, wherever located, in the event of any breach or threatened breach of the provisions of this Section 15.3 by a Member or its Representatives. Such remedies shall not be deemed to be exclusive remedies but shall be in addition to all other remedies available at law or equity to the Company.

- [e] In the event that a Member or any of its Representatives become legally compelled (by deposition, interrogatory, request for documents, subpoena, civil investigation, demand, order or other legal process) to disclose any Confidential Information, or is legally required or requested by any regulatory or self-regulatory organization to disclose any Confidential Information, such Member and its Representatives may do so without liability, provided Recipient [i] promptly notifies the Company prior to any such disclosure, [ii] cooperates with the Company in any attempts it may make to obtain a protective order or other appropriate assurance that confidential treatment will be afforded the Confidential Information and [iii] strictly limit any such disclosure to that which is expressly required by the terms of the legal action or regulatory or self regulatory organization compelling such disclosure.
- [f] The obligations of this Section 15.3 shall survive the termination of this Agreement and any Dissolution and Liquidation of the Company.

#### ARTICLE 16. GENERAL PROVISIONS

16.1 Amendment. Except as otherwise provided in this Agreement, this Agreement may be amended in a writing signed by Members who own more than 50% of the outstanding Units; provided, however, that [a] no amendment that would require any Member to make an additional Capital Contribution to the Company or impose personal liability on a Member for any debt, obligation or liability of the Company shall be effective unless set forth in a writing signed by such Member; [b] no amendment that would change adversely the rights and or obligations of the holders of a Profits Interest Unit in a manner that is different than its effect on the rights or obligations of the holders of Common Units shall be effective unless approved by the holders of two-thirds of the outstanding Profits Interest Units; [c] except for an amendment required in connection with the admission of an additional Member in accordance with the terms of this Agreement, modify or alter the method of determining, the order of priority or the interest of a Member in [i] allocations of Profits or Losses, [ii] allocations or Distributions of Available Cash, or [iii] allocation or Distribution of proceeds resulting from the Liquidation of the Company, unless

44

such amendment receives the affirmative vote or written consent of each Member adversely affected thereby; or [d] amend the provisions of Sections 7.1 or 7.2 unless approved by the holders of 90% of the Common Units. Any duly adopted amendment to this Agreement is binding on, and inures to the benefit of, each Person who holds a Unit at the time of such amendment, without the requirement that such Person sign the amendment or any republication or restatement of this Agreement.

- 16.2 Representations. Each Member hereby represents and warrants to each other Member that, as of the signing of this Agreement:
  - [a] Such Member is duly organized, validly existing and in good standing under the laws of the jurisdiction where it purports to be organized, and is a United States Person;
  - [b] Such Member has full power and authority as a corporation or limited liability company to enter into and perform its obligations under this Agreement;
  - [c] All actions on the part of such Member necessary to authorize the signing and delivery of this Agreement, and the performance by such Member of its obligations hereunder, have been duly taken;
  - [d] This Agreement has been duly signed and delivered by a duly authorized officer or other representative of such Member and constitutes the legal, valid and binding obligation of such Member enforceable in accordance with its terms, except as such enforceability may be affected by applicable bankruptcy, insolvency or other similar laws affecting creditors' rights generally, and except that the availability of equitable remedies is subject to judicial discretion;
  - [e] No consent or approval of any other Person is required in connection with the signing, delivery and performance of this Agreement and the Reorganization Agreement by such Member; and
  - [f] The signing, delivery and performance of this Agreement and the Reorganization Agreement do not violate the organizational documents of such Member or any material agreement to which such Member is a party or by which such Member is bound.
    - 16.3 Unregistered Interests. Each Member [a] acknowledges that the Units

are being offered and sold without registration under the Securities Act of 1933, as amended, or under similar provisions of state law, [b] represents and warrants that such Member is acquiring the Units for such Member's own account, for investment, and without a view to the distribution of the Units, [c] represents and warrants that it is an "accredited investor" as defined in Rule 501(a) of the Regulation D under the Securities Act of 1933 and [d] agrees not to Transfer, or to attempt to Transfer, all or any part of its Units without registration under the Securities Act of 1933, as amended, and any applicable state securities laws, unless the Transfer is exempt from such registration requirements.

16.4 Waiver of Alternative Withdrawal Rights. Each Member hereby waives and renounces any alternative rights that might otherwise be provided by law upon the withdrawal of such Person and accepts the provisions under this Agreement as such Person's sole entitlement upon the happening of such event.

4.5

- 16.5 Waiver of Partition Right. Each Member hereby waives and renounces any right that it might otherwise have prior to Dissolution and Liquidation to institute or maintain any action for partition with respect to any property held by the Company.
- 16.6 Waivers Generally. No course of dealing will be deemed to amend or discharge any provision of this Agreement. No delay in the exercise of any right will operate as a waiver of such right. No single or partial exercise of any right will preclude its further exercise. A waiver of any right on any one occasion will not be construed as a bar to, or waiver of, any such right on any other occasion.
- 16.7 Equitable Relief. If any Member proposes or refuses to Transfer all or any part of its Units in violation of the terms of this Agreement, the Company or any Member may apply to any court of competent jurisdiction for an injunctive order prohibiting or requiring such proposed Transfer, and the Company or any Member may institute and maintain any action or Proceeding against the Person proposing or refusing to make such Transfer to compel the specific performance of this Agreement. Any attempted Transfer in violation of this Agreement is null and void, and of no force and effect. The Person against whom such action or Proceeding is brought hereby irrevocably waives the claim or defense that an adequate remedy at law exists, and such Person will not urge in any such action or proceeding the claim or defense that such remedy at law exists. The prevailing party in any such proceeding shall be entitled to recover its costs and expenses, including reasonable attorneys' fees, of preparing for and participating in the proceeding.
- 16.8 Arbitration. The Members will attempt in good faith to resolve any controversy or claim arising out of or relating to this Agreement through discussions between the senior management of the Members. If these discussions are unsuccessful, except as provided in Section 16.7, the Members agree that any action asserting a claim by one Member against another Member hereto arising out of or relating to this Agreement shall, on the written notice by one Member to the others, be submitted to binding arbitration to be held in Seattle, Washington. The arbitration shall be conducted by and in accordance with the Commercial Arbitration Rules of the American Arbitration Association. The parties shall hold an initial meeting within thirty (30) days from receipt of notice from the requesting party of a request for arbitration. Unless otherwise agreed in writing, they will jointly appoint a mutually acceptable arbitrator not affiliated with either party. If they are unable to agree upon such appointment within thirty (30) days of the initial meeting, the parties shall obtain an odd numbered list of not less than five (5) potential arbitrators from the Superior Court for the Third Judicial District, State of Alaska. Each party shall alternatively strike a single name from the list until only one name remains, with such person to be the arbitrator. The party requesting the arbitration shall strike the first name. Each party shall pay one-half (1/2) of the costs related to the arbitration, unless the arbitrator's decision provides otherwise. Each party shall bear its own costs to prepare for and participate in the arbitration. Each party shall produce at the request of the other party, at least thirty (30) days in advance of the hearing, all documents to be submitted at the hearing and such other documents as are relevant to the issues or likely to lead to relevant information. The arbitrator shall promptly render a written decision, in accordance with Alaska law and supported by substantial evidence in

46

the record. The prevailing party shall be entitled to recover reasonable attorneys' fees, costs, charges and expended or incurred therein, if the arbitrator's decision so provides. Failure to apply Alaska law, or entry of a decision that is not based on substantial evidence in the record, shall be additional grounds for modifying or vacating an arbitration decision. Judgment on any arbitration award shall be entered in any court of competent jurisdiction. In any subsequent arbitration, the decision in any prior arbitration of this Agreement shall not be deemed conclusive of the rights among the parties hereunder.

16.9 Remedies for Breach. The rights and remedies of the Members set forth in this Agreement are neither mutually exclusive nor exclusive of any right or remedy provided by law, in equity or otherwise, and all legal remedies

(such as monetary damages) as well as all equitable remedies (such as specific performance) will be available for any breach or threatened breach of any provision of this Agreement.

- 16.10 Costs. If the Company or any Member retains counsel for the purpose of enforcing or preventing the breach or any threatened breach of any provisions of this Agreement or for any other remedy relating to it, then the prevailing party will be entitled to be reimbursed by the nonprevailing party for all costs and expenses so incurred (including reasonable attorney's fees, costs of bonds and fees and expenses for expert witnesses.
- 16.11 Counterparts. This Agreement may be signed in multiple counterparts, the signature pages of which may be detached and reattached to another identical counterpart. Each counterpart will be considered an original instrument, but all of them in the aggregate will constitute one agreement.
- 16.12 Notice. All Notices under this Agreement will be in writing and will be either delivered or sent addressed as follows: [a] if to the Company, at the Company's principal office in Anchorage, Alaska, and [b] if to any Member, at such Person's address as then appearing in the records of the Company. In computing time periods, the day Notice is given will be included.
- 16.13 Deemed Notice. Any Notices given to the Company or any Member in accordance with this Agreement will be deemed to have been duly given: [a] on the date of receipt if personally delivered, [b] five days after being sent by U.S. mail, postage prepaid, [c] on the date of receipt, if sent by registered or certified U.S. mail, postage prepaid, [d] on the date of receipt, if sent by facsimile or telecopier transmission (with telephonic confirmation by the recipient) or [e] one Business Day after having been sent by a nationally recognized overnight courier service.
- 16.14 Partial Invalidity. Wherever possible, each provision of this Agreement will be interpreted in such manner as to be effective and valid under applicable law. However, if for any reason any one or more of the provisions of this Agreement are held to be invalid, illegal or unenforceable in any respect, such action will not affect any other provision of this Agreement. In such event, this Agreement will be construed as if such invalid, illegal or unenforceable provision had never been contained in it.
- 16.15 Entire Agreement. This Agreement (including its Schedules and Exhibits) contains the entire agreement and understanding of the Members concerning its subject matter.
- 16.16 Benefit. Except as otherwise set forth in Section 9.1, this Agreement and the rights and obligations of the Members hereunder will inure

47

solely to the benefit of the Members and their Transferees and the Company, without conferring on any other Person any rights of enforcement or other rights.

- 16.17 Further Assurances. Each Member will sign and deliver, without additional consideration, such other documents of further assurance as may reasonably be necessary to give effect to the provisions of this Agreement.
- 16.18 Headings. Article and section titles have been inserted for convenience of reference only. They are not intended to affect the meaning or interpretation of this Agreement.
- 16.19 Terms. Terms used with initial capital letters will have the meanings specified, applicable to both singular and plural forms, for all purposes of this Agreement. All pronouns (and any variations) will be deemed to refer to the masculine, feminine or neuter, as the identity of the Person may require. The singular or plural includes the other, as the context requires or permits. The word "include" (and any variation) is used in an illustrative sense rather than a limiting sense. The word "day" means a calendar day, unless a Business Day is specified.
- 16.20 Governing Law. This Agreement will be governed by, and construed in accordance with, the laws of the State of Alaska. Any conflict or apparent conflict between this Agreement and the Act will be resolved in favor of this Agreement, except as otherwise required by the Act.
- 16.21 No Tax Advice. All Members acknowledge that any tax advice express or implicit in the provision of this Agreement are not intended or written to be used, and cannot be used, by any taxpayer for the purpose of avoiding penalties that may be imposed on tax payer by the Internal Revenue Service. Each Member should seek advice based on its particular circumstances from an independent tax advisor.

IN WITNESS WHEREOF, the undersigned have caused this Operating Agreement of Alaska Digitel, LLC, to be duly executed and delivered, effective from the date first above mentioned, notwithstanding the actual date of signing.

GENERAL COMMUNICATION, INC.

By: Name: Title:

AKD HOLDINGS, LLC

By: Name: Title:

PACIFICOM HOLDINGS, L.L.C.

By: Name: Title:

FIRE LAKE PARTNERS, L.L.C.

By: Name: Title:

RED RIVER WIRELESS, LLC

By: Name: Title:

GRAYSTONE HOLDINGS, LLC

By: Name: Title:

LIST OF SCHEDULES

Schedule

1.3 Names and Addresses of Members4.1 Initial Capital Contributions

LIST OF EXHIBITS

Exhibit

A Form of Assignment of Units

B Form of Transferee's Agreement

C Form of Management Agreement

D Examples of Adjustments to Common Units and Profits Interest

Units Outstanding

E Initial Budget

**** CONFIDENTIAL TREATMENT

SCHEDULE 1.3

Names, Addresses and Unit Ownership

#### of Members

Members

General Communication, Inc. 2550 Denali Street, Suite 1000

Anchorage, AK 99503 Fax: (907) 868-5676

Attn: Corporate Counsel

With a copy of any Notice to:

General Communication, Inc. 2550 Denali Street, Suite 1000 Anchorage, AK 99503

Fax: (907) 868-5676

Attn: General Manager & Executive Vice President

AKD Holdings, LLC 5350 Poplar Avenue, Suite 875 Memphis, TN 38119

Fax: (901) 763-3369 Attn: Stephen M. Roberts

With a copy of any Notice to:

Jack S. Magids, Esq. The Bogatin Law Firm, PLC 1661 International Place Drive, Suite 300 Memphis, Tennessee 38120-1431 Fax: (901) 767-2803

**** CONFIDENTIAL TREATMENT

Members

Red River Wireless, LLC 5350 Poplar Avenue, Suite 875 Memphis, TN 38119 Fax: (901) 763-3369 Attn: Stephen M. Roberts

With a copy of any Notice to:

Jack S. Magids, Esq. The Bogatin Law Firm, PLC 1661 International Place Drive, Suite 300 Memphis, Tennessee 38120-1431 Fax: (901) 767-2803

Graystone Holdings, LLC 5350 Poplar Avenue, Suite 875 Memphis, TN 38119 Fax: (901) 763-3369 Attn: J. Michael Keenan

With a copy of any Notice to:

Jack S. Magids, Esq. The Bogatin Law Firm, PLC 1661 International Place Drive, Suite 300 Memphis, Tennessee 38120-1431 Fax: (901) 767-2803

Pacificom Holdings, L.L.C. 5350 Poplar Avenue, Suite 875 Memphis, TN 38119 Fax: (901) 763-3369 Attn: Stephen M. Roberts

With a copy of any Notice to:

Jack S. Magids, Esq. The Bogatin Law Firm, PLC 1661 International Place Drive, Suite 300 Memphis, Tennessee 38120-1431 Fax: (901) 767-2803

52 **** CONFIDENTIAL TREATMENT

Members Fire Lake Partners, L.L.C. 3127 Commercial Drive

Unit Ownership **** Common Units

**** Common Units

Unit Ownership **** Common Units

**** Common Units

**** Common Units

Unit Ownership **** Profits Interest Units Anchorage, AK 99501

Fax:

Attn: Stephen M. Roberts

With a copy of any Notice to:

Jack S. Magids, Esq. The Bogatin Law Firm, PLC 1661 International Place Drive, Suite 300 Memphis, Tennessee 38120-1431 Fax: (901) 767-2803

53
**** CONFIDENTIAL TREATMENT

SCHEDULE 4.1

Initial Capital Contributions

Agreed Fair Market Value

Members of Deemed Initial Contribution

General Communication, Inc.

AKD Holdings, LLC

Red River Wireless, LLC

Graystone Holdings, LLC

Pacificom Holdings, L.L.C.

Fire Lake Partners, L.L.C.

______

(1) **** under the Reorganization Agreement in accordance with Section 4.1.

EXHIBIT A

Form of

Assignment of Units

The undersigned Transferor hereby transfers and assigns

Digitel, LLC, a Alaska limited liability company, to
, as Transferee. The Capital Account of the Transferor that is attributable to the transferred Units will carry over to the Transferee. The Units transferred is subject to all of the terms and conditions of that certain Second Amended and Restated Operating Agreement of , dated as of , as such Agreement may be amended ("Operating Agreement"). The Transferor shall remain liable for all of its liabilities under the Operating Agreement.

Transferor:

By:

Name: Title: Date:

EXHIBIT B

Form of

Transferee's Agreement

As a Transferee of Units in Alaska Digitel, LLC, a Alaska limited liability company governed by a Second Amended and Restated Operating Agreement dated as of , the undersigned agrees to be bound as a party to such Agreement (which, as it may be amended, is hereby incorporated by reference). The Transferee acknowledges and agrees that, unless admitted as a Member of the limited liability company as provided in such Agreement, the Transferee will have only the limited rights of an assignee as specified by law.

 ${\tt Transferee:}$ 

By:

Name:

Title:

Address:

Taxpayer ID Number:
Telephone Number:
Fax Number:
EXHIBIT C
Form of Management Agreement
[This exhibit does not contain information material to an investment decision that is not otherwise disclosed in the agreement and is not included in the filing of the agreement with the SEC as being not material to investor decisions.]

EXHIBIT D

Examples of Adjustments to Common Units and Profits Interest Units Outstanding See attached Schedule 2.3.3 to the Reorganization Agreement.

EXHIBIT E

Initial Budget

[To be inserted at the Closing under the Reorganization Agreement.]

SECOND AMENDED AND RESTATED OPERATING AGREEMENT

OF

ALASKA DIGITEL, LLC

[], 2006

THE UNITS IN THIS LIMITED LIABILITY COMPANY HAVE NOT BEEN REGISTERED WITH THE SECURITIES AND EXCHANGE COMMISSION OR STATE SECURITIES AUTHORITIES AND MAY NOT BE SOLD OR TRANSFERRED IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AND ANY APPLICABLE STATE SECURITIES LAWS OR AN OPINION OF COUNSEL ACCEPTABLE TO THE COMPANY THAT REGISTRATION IS NOT REQUIRED. THE SALE OR OTHER TRANSFER OF THE UNITS IS ALSO RESTRICTED BY PROVISIONS OF THIS AGREEMENT.

Allocation of Profits and Losses Generally......18

 Payment
 21

 Withholding
 21

MANAGEMENT 22

1.3

1.4 1.5

1.6

1.7

1.8

2.1

2.2

2.3

3.1

3.2

3.3 3.4

3.5

3.6

4.1

4.2

4.3

4.4 4.5

4.6

4.7

4.8

4.9

5.1

5.2

5.3

5.4

5.5

5.6

5.7

5.8

5.9 5.10

5.11

5.12

5.13

6.1

6.2

6.3 6.4

6.5

7.1

7.2 7.3

7.4

7.5

7.6

ARTICLE 8.

ARTICLE 6.

ARTICLE 7.

ARTICLE 5.

ARTICLE 4.

ARTICLE 2.

ARTICLE 3.

8.1	Limited Liability27
8.2 8.3	Capital Contribution
8.4	Reliance
ARTICLE 9.	INDEMNIFICATION
9.1	General
9.2	Exception
9.3	Expense Advancement
9.4	Insurance
9.5	Indemnification of Others
9.6	Exculpation
9.7	Rights Not Exclusive
ARTICLE 10.	ACCOUNTING AND REPORTING
10.1	Fiscal Year29
10.2	Tax Accounting Method
10.3	Tax Elections
10.4	Returns
10.5	Reports; Annual Financial Statements; Regulatory Reporting Obligations
10.6	Books and Records
10.7	Information
10.8	Banking
10.9	Tax Matters; Tax Matters Partner
10.10	Classification of Company as Partnership for Tax Purposes, Not State Law32
ARTICLE 11.	ii DISSOLUTION
ARIICLE II.	D1550H0110N
11.1	Dissolution
11.2	Events of Withdrawal
11.3	Continuation
ARTICLE 12.	LIQUIDATION
10 1	Liquidation
12.1 12.2	[Reserved]
12.3	Priority of Payment
12.3	Liquidating Distributions
12.5	No Restoration Obligation
12.6	Timing
12.7	Liquidating Reports
12.8	Articles of Dissolution
ARTICLE 13.	TRANSFER RESTRICTIONS
13.1	General Rule
13.2	Permitted Transfers
13.3	Permitted Pledges
13.4	General Conditions on Transfer
13.5	Rights of Transferees
ARTICLE 14.	LIQUIDITY RIGHTS
14.1	Call Rights
14.2	Put Rights
14.3	Determination of Appraised Unit Value
14.4	Closing Procedures40
ARTICLE 15.	CERTAIN BUSINESS MATTERS41
15.1	Distribution of Assets in Kind41
15.2	Other Business Ventures
15.3	Confidentiality
	-
ARTICLE 16.	GENERAL PROVISIONS42
16.1	Amendment
16.2	Representations
16.3	Unregistered Interests
16.4	Waiver of Alternative Withdrawal Rights
16.5	Waiver of Partition Right
16.6	Waivers Generally44
16.7	Equitable Relief
16.8	Arbitration
16.9	Remedies for Breach45
16.10	
10.10	Costs
16.10	Costs
16.11	Counterparts45
16.11 16.12	Counterparts
16.11 16.12	Counterparts

16.15	Entire Agreement	.45
16.16	Benefit	.45
16.17	Further Assurances	.46
16.18	Headings	.46
16.19	Terms. Terms.	.46
16.20	Governing Law	.46
16.21	No Tax Advice	.46

  |  |Exhibit C

[This exhibit does not contain information material to an investment decision that is not otherwise disclosed in the agreement and is not included in the filing of the agreement with the SEC as being not material to investor decisions.]

Exhibit D

[This exhibit does not contain information material to an investment decision that is not otherwise disclosed in the agreement and is not included in the filing of the agreement with the SEC as being not material to investor decisions.]

Exhibit E

## **** CONFIDENTIAL TREATMENT

## NONCOMPETITION AGREEMENT

This Noncompetition Agreement (the "Agreement") is made and entered into as of this day of , 200 by and among , an individual residing in the State of ("Seller"), Alaska Digitel, LLC, an Alaska limited liability company ("AKD"), and General Communication, Inc., an Alaska corporation ("GCI").

## RECITALS

A. GCI, AKD, AKD Holdings, LLC, a Delaware limited liability company, the members of AKD and the members of Denali PCS, LLC, an Alaska limited liability company, are parties to that certain Reorganization Agreement dated as of June , 2006 (the "Reorganization Agreement"), pursuant to which, among other things, GCI will acquire a majority interest in AKD and AKD will redeem certain membership interests of the members of AKD.

B. This Agreement is a Noncompetition Agreement referred to in Section 7.1(o) of the Reorganization Agreement and, pursuant to the Reorganization Agreement, must be entered into by the parties hereto as a condition to the consummation of the transactions contemplated by the Reorganization Agreement. The execution and performance of this Agreement is **** and **** the transactions contemplated by the Reorganization Agreement.

C. Seller is a member of , which owns a membership interest in AKD, and accordingly, Seller will benefit from the transactions contemplated by the Reorganization Agreement. Seller is also a member of Fire Lake Partners, LLC, an Alaska limited liability company that will enter into a Management Services Agreement with AKD upon closing of the transactions contemplated by the Reorganization Agreement (the "Management Services Agreement"). In return for the consideration provided under the Reorganization Agreement and the Management Services Agreement, Seller is willing to refrain from competing with GCI, AKD and their respective subsidiaries as provided herein.

## AGREEMENT

NOW, THEREFORE, in consideration of the mutual covenants and the agreements hereinafter set forth, GCI's willingness to consummate the transactions contemplated by the Reorganization Agreement and the Management Services Agreement, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto covenant and agree as follows:

1. Covenant Against Competition; Confidentiality and Nondisclosure. Seller acknowledges that (i) GCI, AKD and their respective subsidiaries are engaged in the Business (as defined below); (ii) Seller is one of a limited number of persons who has developed the Business of AKD and will continue to provide management services to AKD after the transactions contemplated by the Reorganization Agreement; (iii) Seller's work for AKD has brought Seller into close contact with many confidential matters of AKD not readily available to the public; and (iv) GCI would not have consummated the transactions contemplated by the Reorganization Agreement unless Seller entered

## **** CONFIDENTIAL TREATMENT

into this Agreement. For purposes of this Agreement, "Business" means the provision of any of the following services provided by, or of services which compete with any of the following services provided by, or which would result in a loss of business with respect to any of the following services by, GCI, AKD and each of their respective subsidiaries, affiliates, successors and assigns

(collectively, the "GCI Parties"): the sale, resale or provision of wireless telecommunications services, the ownership, licensing, sublicensing, acquisition or disposition of licenses for spectrum capacity, and any other business which constitutes a reasonable extension of the foregoing. Accordingly, Seller covenants and agrees that,  $\$ from the date of this  $\$ Agreement **** date **** (the "Restricted Period"), except as contemplated by the Management Services Agreement, Seller will not, within the Geographic Area (as defined in Section 1(d) below), without the prior written consent of GCI, (i) engage in the Business for Seller's own account; (ii) enter into the employ of, or render services to or for, any entity that competes with any GCI Party, with respect to the Business (referred to herein as a "Restricted Entity"); or (iii) become interested in any Restricted Entity in any capacity, including as an individual, partner, shareholder, officer, director, principal, agent, trustee, employee or consultant; provided, however, that Seller may own, directly or indirectly, solely as a passive investment, securities of any entity traded on any national securities exchange or automated quotation system if Seller is not a controlling person of, or a member of a group which controls, such entity and does not, directly or indirectly, "beneficially own" (as defined in Rule 13d-3 of the Securities Exchange Act of 1934, as amended) five percent (5%) or more of any class of securities of such Restricted Entity.

(b) During the Restricted Period, Seller shall keep secret and retain in strictest confidence, and shall not use for the benefit of Seller or others except in connection with the business and affairs of the GCI Parties, all confidential information relating to the Business and to any GCI Party, including, without limitation, trade secrets, "know-how," client lists, subscriber lists, details of contracts, pricing policies, operational methods, marketing plans or strategies, business acquisition, disposition or reorganization plans, requisition forms or procedures, software programs, new personnel acquisition plans, technical processes, designs and design projects, inventions and research projects, technical services, and other confidential business affairs relating to, arising from or in connection with the Business and learned by Seller heretofore or hereafter, and shall not disclose them to anyone outside of the GCI Parties, either during or after retention as an employee by any GCI Party, except as required by law or with GCI's express written consent or as reasonably necessary in the course of Seller's services to a GCI Party.

For purposes of this Agreement, confidential information shall not include any information (i) which is or becomes so generally known as to be part of the public domain, (ii) which is lawfully received from a third party, without restriction, (iii) which was known by the Seller without restriction on disclosure prior to receipt from the GCI Party, or (iv) which is disclosed pursuant to subpoena, court order or similar legal process where the party whose confidential information is to be disclosed has had a reasonable opportunity to appear and object to disclosure.

In the event that Seller is requested or required (by oral questions, interrogatories, requests for information or documents, subpoena, civil investigative demand or similar process) to disclose any confidential information as described in this Section 1(b), Seller agrees to provide GCI with prompt notice of such request(s) so that it may seek an appropriate protective order and/or waive Seller's compliance with the provisions of this Agreement. If in the absence of a protective order or the

# 2 **** CONFIDENTIAL TREATMENT

receipt of a waiver hereunder, Seller is nonetheless, in the opinion of Seller's counsel, compelled to disclose information concerning any GCI Party to any tribunal or else stand liable for contempt or suffer other censure or penalty, Seller may disclose such information to such tribunal without liability hereunder.

(c) During the Restricted Period, except as contemplated by the Management Services Agreement, Seller shall not, directly or indirectly, (other than for purposes of the Business of any GCI Party) (i) for purposes of conducting Business, solicit any customer or client of any GCI Party with whom the Seller has had contact in the Business, (ii) for purposes of conducting Business in the Geographic Area, solicit any person or entity that is or has been a customer or client of any GCI Party, (iii) solicit or encourage to leave the employment or service of any GCI Party any employee or consultant of such GCI Party, or (iv) hire or retain any employee or consultant who is employed or retained by, or who, **** the date of solicitation or encouragement, has left the employment or service of, any GCI Party to engage in the Business or any business substantially similar to the Business in the Geographic Area.

(d) The parties agree that for purposes of this Agreement, "Geographic Area" shall mean ****.

2. Rights and Remedies Upon Breach of the Restrictive

Covenants.

(a) Seller recognizes that GCI and AKD may not have an adequate remedy at law to protect their rights hereunder. Accordingly, Seller

agrees that in the event of a breach of this Agreement by Seller, GCI, AKD or any other GCI Party shall have the right to an injunction without bond in any court of competent jurisdiction permanently enjoining Seller from a violation of Section 1 of this Agreement (the "Restrictive Covenants").

(b) Seller recognizes and agrees that in the event of a violation of any of the Restrictive Covenants, the period during which Seller shall not compete shall be suspended during the period Seller is engaged in conduct constituting such violation and shall resume after such violation has been remedied.

(c) The remedies set forth in Section 2(a) above shall not limit, eliminate, prohibit or restrict any other rights that GCI, AKD or any other GCI Party may have under law for violation by Seller of this Agreement and shall not be mutually exclusive and any one or all may be pursued without the pursuit of one impairing or precluding the pursuit of another except as otherwise provided by law.

3. Severability of Restrictive and Other Covenants. The parties agree that the provisions of each of the preceding sections of this Agreement are independent of and severable from each other and the invalidity of any section or any portion thereof shall not affect the validity or hinder the enforceability of the remaining provisions of this Agreement. The parties expressly agree and declare that the time limitation and geographic scope set forth in Section 1 of this Agreement are reasonable, are properly required for the adequate protection of the Business of the GCI Parties and that in the event such time limitation and/or geographic scope is deemed to be unreasonable by the final decision of an arbitrator or a court of competent jurisdiction, each of

3

the parties agrees to submit to such revision or modification thereof as said arbitrator or court shall deem reasonable.

4. Notices. All notices and other communications required or permitted hereunder shall be in writing (including any facsimile transmission or similar writing), and may be given by any means selected by the sender. Each such notice or other communication shall be effective (i) if sent by facsimile to the recipient's fax number given below, when such facsimile is transmitted and the sender's facsimile machine confirms transmission, (ii) if sent by reputable overnight courier to the recipient's address given below, one business day after being delivered to such courier or (iii) if sent by any other means, when actually received.

(a) If to GCI:

General Communication, Inc. 2550 Denali Street, Suite 1000 Anchorage, AK 99503 Attention : General Manager & Executive Vice President Fax No. : (907) 868-5676

(b) If to AKD:

Pacificom Holdings, LLC c/o Wireless Partners 5350 Poplar Avenue, Suite 875 Memphis, TN 38117 Attention: Stephen M. Roberts Fax No.: (901) 763-3369

(b) If to Seller:

Facsimile:			

or to such other address or facsimile number as any of the persons designated above may have specified in a notice or communication duly given to the other designated person as provided herein.

- 5. Assignability. This Agreement shall be binding upon and inure to the benefit of GCI, AKD and each of their respective successors and assigns. This Agreement shall not be assignable by Seller.
- 6. Entire Agreement. This Agreement contains the entire agreement among the parties hereto with respect to the subject matter hereof.
- 7. Waivers, Amendments and Further Agreements. Neither this Agreement nor any term or condition of this Agreement, including without

4

limitation the terms and conditions of this Section 7, may be waived, modified or amended in whole or in part as against any of the parties except by written

instrument executed by each of the parties expressly stating that it is intended to operate as a waiver, modification or amendment of this Agreement or the applicable term or condition of this Agreement. Each of the parties hereto agrees to execute all such further instruments and documents and to take all such further action as the other party may reasonably require in order to effectuate the terms and purposes of this Agreement.

- 8. Counterparts. This Agreement may be executed in any number of counterparts, each of which shall be deemed an original, but all of which shall constitute one and the same instrument.
- 9. Governing Law. This Agreement shall be governed by and construed and enforced in accordance with the laws of the State of Alaska.
- 10. Attorneys' Fees. In the event litigation arising out of or related to this Agreement, the substantially prevailing party shall be entitled to reimbursement of its reasonable attorneys' fees and costs of suit in addition to such other relief as may be granted, including but not limited to such fees and costs incurred in collections and in enforcement and appeal of any judgment that may be rendered (except to the extent determined otherwise by the applicable tribunal).

IN WITNESS WHEREOF, the parties hereto have executed or caused to be executed this Agreement as of the date first above written.

GENERAL COMMUNICATION, INC.

By:	
Name:	
Title:	
ALASKA DIGITE	
Ву:	
Name:	
Title:	
SELLER	
	5

Schedule 1.1

[This exhibit does not contain information material to an investment decision that is not otherwise disclosed in the agreement and is not included in the filing of the agreement with the SEC as being not material to investor decisions.]

Schedule 2.3.2

[This exhibit does not contain information material to an investment decision that is not otherwise disclosed in the agreement and is not included in the filing of the agreement with the SEC as being not material to investor decisions.]

Schedule 2.3.3

[This exhibit does not contain information material to an investment decision that is not otherwise disclosed in the agreement and is not included in the filing of the agreement with the SEC as being not material to investor decisions.]

Schedule 4

[This exhibit does not contain information material to an investment decision that is not otherwise disclosed in the agreement and is not included in the filing of the agreement with the SEC as being not material to investor decisions.]

Schedule 5

[This exhibit does not contain information material to an investment decision that is not otherwise disclosed in the agreement and is not included in the filing of the agreement with the SEC as being not material to investor decisions.]

Schedule 7.1(n)

[This exhibit does not contain information material to an investment decision that is not otherwise disclosed in the agreement and is not included in the

filing of the agreement with the SEC as being not material to investor  $\mbox{\tt decisions.}]$ 

## SECTION 302 CERTIFICATION

- I, Ronald A. Duncan, certify that:
- I have reviewed this quarterly report on Form 10-Q of General Communication, Inc. for the period ended June 30, 2006;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are

reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

August 8, 2006

/s/ Ronald A. Duncan
Ronald A. Duncan
President and Director

## SECTION 302 CERTIFICATION

- I, John M. Lowber, certify that:
- 1. I have reviewed this quarterly report on Form 10-Q of General Communication, Inc. for the period ended June 30, 2006;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to

record, process, summarize and report financial information; and

b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

August 8, 2006

/s/ John M. Lowber

John M. Lowber Senior Vice President, Chief Financial Officer, Secretary and Treasurer CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of General Communication, Inc. (the "Company") on Form 10-Q for the period ended June 30, 2006 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Ronald A. Duncan, Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. ss. 1350, as adopted pursuant to ss. 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and result of operations of the Company.

Date: August 8, 2006 /s/ Ronald A. Duncan

-----

Ronald A. Duncan Chief Executive Officer General Communication, Inc. CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of General Communication, Inc. (the "Company") on Form 10-Q for the period ended June 30, 2006 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, John M. Lowber, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. ss. 1350, as adopted pursuant to ss. 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and result of operations of the Company.

Date: August 8, 2006 /s/ John M. Lowber

_____

John M. Lowber Chief Financial Officer General Communication, Inc.